INSIDE BAREFOOT ECONOMICS

Patrick Thomas Kletzka



Inside Barefoot Economics

Patrick Thomas Kletzka (*1988) is a German economist. He holds a doctorate in economics from the University of Wuppertal (Germany), has directed a number of grassroots development projects and worked as an independent researcher in the slums of Nairobi (Kenya). He was a friend and mentee of the late German-Chilean economist, Alternative Nobel laureate and father of barefoot economics, Manfred A. Max-Neef (1932–2019).

Inside Barefoot Economics

Patrick Thomas Kletzka

Logos Verlag Berlin



Bibliographic information published by Die Deutsche Bibliothek

Die Deutsche Bibliothek lists this publication in the Deutsche Nationalbibliografie; detailed bibliographic data is available in the Internet at http://dnb.ddb.de.



© Logos Verlag Berlin GmbH, 2021 All rights reserved.

ISBN 978-3-8325-5346-3

Logos Verlag Berlin GmbH Georg-Knorr-Str. 4 Geb. 10 D-12681 Berlin

Tel: +49 (0)30 / 42 85 10 90 Fax: +49 (0)30 / 42 85 10 92 https://www.logos-verlag.de To Manfred, in memoriam.

"One only understands the things that one tames," said the fox. ... "What must I do, to tame you?" asked the little prince.

"You must be very patient," replied the fox. "First you will sit down at a little distance from me, over there in the grass. I shall look at you out of the corner of my eye, and you will say nothing. Language is the source of misunderstandings. But you will sit a little closer to me, every day ..."

– Antoine de Saint-Exupéry

Contents

1	Introduction											
	1.1	Motiv	ation	13								
	1.2	Resear	rch Context	14								
	1.3	Resear	rch Agenda	18								
	1.4	Struct	ure of the Treatise	22								
2	The	Princip	les of Barefoot Economics									
	A Theoretical-Conceptual Groundwork											
	2.1	Introd	luction	25								
	2.2	The 'Shoes' of Economics										
		2.2.1	Milton Friedman's Methodology of									
			Positive Economics	28								
		2.2.2	Positivism in Economics	31								
	2.3	2.3 'Touching the Ground' of Phenomena										
		2.3.1	Edmund Husserl's Phenomenology	33								
		2.3.2	Manfred Max-Neef's Phenomenological									
			Thought	37								
	2.4	Conclusion										
		2.4.1	Summary	42								
		2.4.2	Contributions and Implications	43								
		2.4.3	Limitations	46								
3	Exp	eriences	s in Barefoot Economics									
	Ani	Empirica	al Case Study	49								
	3.1	Introduction										
	3.2	Field Description										
		3.2.1	The Selection of the Field	52								
		3.2.2	The Phenomenon of Slums	52								
		3.2.3	The Mathare Slums of Nairobi	53								

	3.2.4	Non-Formal Schooling in Mathare	55
3.3	Theore	etical Background	60
	3.3.1	Preliminary Remarks	60
	3.3.2	Joseph Schumpeter's Theory of Economic	
		Development	60
	3.3.3	C. K. Prahalad's Fortune at the Bottom	
		of the Pyramid	66
	3.3.4	Jason Fairbourne et al.'s Microfranchising .	70
3.4	Metho	d	74
	3.4.1	The Appropriateness of the Methodology	
		of Barefoot Economics	74
	3.4.2	The Method of Linguistic Pruning	75
	3.4.3	The Method of Real-World	
		Experimentation	76
	3.4.4	The mpito [®] Project	78
3.5	Result	8	87
	3.5.1	General Observations	87
	3.5.2	Key Performance Driver: 'Branding'	89
	3.5.3	Key Performance Driver: 'Standardisation'	94
3.6	Discus	sion	111
	3.6.1	Preliminary Remarks	111
	3.6.2	Niklas Luhmann's Concept of Inclusion/	
		Exclusion	111
	3.6.3	Karl Polanyi's Concept of Embeddedness .	114
	3.6.4	Manfred Max-Neef et al.'s Theory of	
		Human Scale Development	116
3.7	Conclu	usion	126
	3.7.1	Summary	126
	3.7.2	Contributions and Implications	128
	3.7.3	Limitations	131
Con	clusion		135
4.1	Overal	ll Summary	135
4.2	Overal	ll Contributions and Implications	137
4.3	Overal	ll Limitations	140

4.4 4.5	Outlook Final Remarks .	•	•••	•	•	•	•	•	•			•	•	141 143
Referen	ces													145

1 Introduction

1.1 Motivation

This treatise is driven by the motivation to contribute to the United Nations' (2015: 14) Sustainable Development Goal 1 (SDG 1): "End poverty in all its forms everywhere". The phenomenon of poverty can be defined in many ways. In line with the United Nations, poverty is defined by the World Bank (2005: 11) as "the condition that results from not having adequate resources to satisfy one's basic needs." In this sense, "poverty is hunger. Poverty is lack of shelter. Poverty is unclean drinking water. Poverty is being sick and not being able to see a doctor, being illiterate and not being able to go to school, ..." (ibid.). In other words, not having adequate resources to satisfy basic human needs means to lack access to economic goods, such as food, clothes, shelter, educational and health services. These goods are deemed necessary to fill "the basic needs basket" (UN-DESA 2009: 49). Moreover, the assumption is generally made that people living in poverty cannot access the necessary economic goods because they cannot afford them in monetary terms. Therefore, poor people are generally assumed to have purchasing power classified below a certain real (i.e. inflationadjusted) income threshold. Such income thresholds are referred to as 'poverty lines'. To measure to what extent progress is being made towards achieving SDG 1, the United Nations, as well as other organisations, use as a baseline the international poverty line of PPP (purchasing power parity) US\$ 1.90 a day, as set by the World Bank in 2015. According to this measurement, the United Nations (2020a) estimate that more than 700 million people live under conditions of extreme poverty. Furthermore, the United Nations (2020b) estimate that the impacts of the 2020 'COVID-19 pandemic' will, in the short term, force more than 70 million additional

people into poverty and, thus, lead to the first increase in the global poverty rate since 1998. With regards to the long term, the United Nations and the World Bank predict that the impacts of climate change will push at least 120 million more people into poverty by 2030 (United Nations 2019a). In view of these global trends, the relevance of long-term strengthening of poverty alleviation efforts seems self-evident.

1.2 Research Context

Beyond the identification and elimination of the diverse single causes of poverty (e.g. by means of health or climate research), modern poverty research has, since its inception in the 19th century, tried to approach the phenomenon of poverty, its root cause and its alleviation in more general terms (O'Connor 2001). Thereby, poverty research can be described as an interdisciplinary field extending into various scientific disciplines, such as sociology, political science, the science of history, anthropology, biology and psychology (e.g. Haveman 1997; O'Connor 2001; Brady & Burton 2016). Within the social sciences, poverty has also been addressed by economic scholars. Poverty research can, thereby, be viewed as essential to the discipline of economics, if needs satisfaction is seen as the raison d'être (final cause or purpose) of all economic activity (compare Oeconomica 1.1343a-b). As a result, many theoretical concepts relating to poverty alleviation have been developed in the history of modern economics. These concepts have largely been contradictory. For example, divergent poverty alleviation concepts can be derived from Marxian economics, neoclassical economics, Keynesian economics and Schumpeterian economics (see also section 3.3). Since the 1980s, modern economics gradually entered "the Age of Schumpeter" (Giersch 1984). This heralded the dominance of a poverty alleviation concept based on Schumpeterian economics, namely C. K. Prahalad's 1998 concept of the fortune at the bottom of the pyramid; in short, the BoP concept (see Prahalad & Lieberthal 1998; Prahalad & Hart 1999, 2002; Prahalad [2004]

2005; see also London & Hart 2011; Casado Cañeque & Hart 2015). The concept identifies a low-income population of around four billion people who are generally excluded from the markets of global capitalism. This demographic population segment is called the 'base/bottom of the pyramid' (ibid.). In order to alleviate poverty at the base of the pyramid, the BoP concept argues in favour of inclusive capitalism achieved through large-scale entrepreneurship (ibid.).¹ Recent bibliometric analyses show that "the study of [the] BoP [according to Prahalad's concept] has attracted considerable attention, and hundreds of publications, conferences, and business summits have addressed the issue, from academic, policy, and practitioner stances" (Pineda-Escobar & Merigó 2020: 5537). *Figure 1* illustrates how the number of publications on the BoP concept has increased rapidly over time, with the most significant increase in the second decade of the 21st century.

There can be little doubt that the BoP concept has come to dominate the current academic discourse on poverty alleviation within the scientific discipline of economics and, particularly, business and entrepreneurship research (Peredo et al. 2018; Pineda-Escobar & Merigó 2020). The concept has, however, faced a number of critics (e.g. Jaiswal 2007; Karnani 2007; Landrum 2007; Ilahiane & Sherry 2012; Arora & Romijn 2012; Chatterjee 2014; Peredo et al. 2018), who have found fault with the concept particularly for pragmatic and postcolonial reasons. From a pragmatic perspective, the concept has been criticised for lacking the necessary empirical evidence to demonstrate its effectiveness as a tool for alleviating poverty (e.g. Karnani 2009; Dembek et al. 2020; Landrum 2020). From a postcolonial perspective, the concept has been criticised for stigmatising non-capitalist lifestyles under the banner of poverty (Peredo et al. 2018). Its attempt to include 'the poor' in the markets of global capitalism is considered a reinforcement of "capitalist

¹ Please note that, in this treatise, the term 'base of the pyramid' will be used to refer to the population described above, while the term 'BoP concept' will be used to refer to Prahalad's concept on how to alleviate poverty in this population.



Figure 1: Publications on the BoP concept per year (Pineda-Escobar & Merigó 2020: 5539, adapted).

hegemony" (Montgomery et al. 2012: passim). Consequently, alternatives to the BoP concept are frequently called for (e.g. Jaiswal 2007; Karnani 2007; Landrum 2007; Ilahiane & Sherry 2012; Arora & Romijn 2012; Chatterjee 2014; Peredo et al. 2018).

However, poverty research in economics is not only a theoretical conceptualisation of strategies to alleviate poverty, such as the BoP concept; it also entails distinct methodological approaches to empirical research. The most prominent scientific method of poverty research in economics is the *randomised controlled trial* (*RCT*) (e.g. Cameron et al. 2016; de Souza Leão & Eyal 2019; Banerjee et al. 2019). RCTs are a type of field experiment which test hypotheses by randomly allocating participants to treatment and control groups (see Banerjee & Duflo 2011; Banerjee et al. 2019). The RCT method was developed by the statistician Ronald A. Fisher in the 1920s (Fisher 1925; compare Banerjee et al. 2019). Since the mid-1990s, RCTs have been used increasingly by economists in the context of poverty research (see *Figure 2*) and the method has been particularly promoted by the U.S. economists Esther Duflo, Abhijit Banerjee and Michael Kremer. In 2019, they were jointly awarded the Nobel Prize in Economic Sciences "for their experimental approach to alleviating global poverty" (Nobel Media AB 2020).



Figure 2: Published Randomised Controlled Trials per year (Banerjee et al. 2019: 441).

RCTs are currently generally considered as the methodological "gold standard" of poverty research (Banerjee et al. 2019: passim), capable of achieving "unbiasedness" (Ogden 2017: 40) and, consequently, of producing "hard evidence" (Banerjee 2007: 12). However, the methodological dominance of RCT-based research has also attracted criticism (e.g. Mookherjee 2005; Rodrik 2009; Deaton 2010; Labrousse 2016a; 2016b; Stevano 2020; Lisciandra 2020). Specifically, RCTs have been criticised for methodological and ethical aspects, as well as from a philosophy of science perspective (ibid). From a methodological perspective, the RCT method has been criticised for its inability to examine small populations in a statistically significant manner and for its prerequisite of large-scale investigations (e.g. Lisciandra 2020). Moreover, randomisation does not necessarily result in the treatment group and control group sharing equal characteristics (ibid.). Lastly, social diversity, complexity and progress would undermine the external validity (generalisability) of causal effects identified by the RCT method (see ibid.). From an ethical perspective, RCT-based research has been criticised for its disregard of ethical concerns, which inevitably occur in a randomised experimentation with human subjects (ibid.). Finally, economists using the RCT method in the context of poverty research – commonly called "the randomistas" (Ravallion 2009: passim) – have been criticised for their strong rhetoric of revolutionising 21st century poverty research by entering a new age of scientific objectivity, while simultaneously missing sufficient reflections on their own underlying philosophy of science, which is "imbued with positivism" (Labrousse 2016a: 298).

1.3 Research Agenda

The dominance of the BoP concept and the RCT method in 21st century poverty research in the economic sciences has largely ruled out other approaches (Peredo et al. 2018; Lisciandra 2020). The resultant theoretical and methodological monism has narrowed poverty research to such an extent that it is unable to reflect on criticism (ibid.). An attempt to counterbalance the shortcomings of the pervasive approaches in poverty research in the economic sciences must, therefore, respond to the frequently expressed desire for pluralism in economics (see e.g. King 2002; Sent 2003; Fullbrook 2003; van Bouwel 2005; Alcorn & Solarz 2006; Reardon 2009; Dobusch & Kapeller 2012; Söderbaum 2012; Heise 2017, 2018, 2019; Haucap & Erlei 2019; Petersen et al. 2019). In this sense, this treatise intends to contribute to poverty research by stimulating a greater variety of approaches within the scientific discipline of economics. To achieve this objective, this treatise revives, elaborates and applies an alternative and – until now – neglected approach to economics that was developed in the context of poverty research by the internationally renowned German-Chilean economist and

Alternative Nobel laureate Manfred Max-Neef (1932–2019) in the 1970s, namely *barefoot economics*.

In an initial step into the research subject of barefoot economics, an evaluative literature review (Efron & Ravid 2019) was conducted. Electronic searches were made on EBSCO, Elsevier, Google Scholar and Web of Science using the Boolean search string: ("barefoot economics" OR "barefoot economic" OR "barefoot economist") AND "Max-Neef". In total, the databases delivered 326 search results.² The review of these results revealed the use of the term 'barefoot economics' with explicit reference to Max-Neef in a number of diverse scientific publications over recent decades (e.g. Holden 1984; Henderson 1989; Dodds 1997; Lategan 1997; Lucena 2010: Cato 2012; Imas et al. 2012; Truong et al. 2014; Balkema & Pols 2015; Göpel 2016; Smolski 2016; Hidalgo-Capitán & Cubillo-Guevara 2017; Kellev & Kester 2017; Weston et al. 2017; Weston & Imas 2018; Madrueño & Martínez-Osés 2019; Flynn 2020; Alcoff 2020; Stahel 2020). However, despite mentioning the term, none of these publications addresses barefoot economics as its primary object of investigation.³ The low degree of engagement with barefoot economics in these publications, despite Max-Neef's extensive scientific oeuvre, reveals a fundamental research gap. By selecting barefoot economics as its research subject, this treatise tries to close this research gap posing and answering the following overall research question:

What is the scientific contribution of barefoot economics to poverty research?

² The search results included a considerable number of duplicates.

³ A probable explanation is that Max-Neef followed the grand tradition of modern economists – such as Adam Smith (1723–1790), David Ricardo (1772–1823), John Stuart Mill (1806–1873), Karl Marx (1818–1883), Thorstein Veblen (1857– 1929), John Maynard Keynes (1883–1946) and Joseph Alois Schumpeter (1883– 1950) – who were deeply engaged in philosophical issues (compare Heilbroner [1953] 1999; Kohr [1982] 1992; see also Max-Neef [1982] 1992, [1988] 1991a, [1989] 1991b, 2005a, 2007, 2009, 2016). Contemporary economists, in turn, are generally not concerned with philosophical issues (see Blackwell et al. 2016).

In order to approach this overall research question, *two consecutive scientific studies* were conducted and these are presented in this treatise.⁴ Acknowledging Max-Neef's view that "[t]heory and praxis are both indispensable" (Max-Neef 1983, as cited in Right Livelihood Foundation [1983a] 2020a: para. 14), the first study is dedicated to the *theory* of barefoot economics and the second study is dedicated to the *practice* of barefoot economics. A brief overview of both studies is given below.

The *first study* investigates barefoot economics in theory. In this sense, it is a study *about* barefoot economics. The study takes a meta-economic perspective to investigate how to define barefoot economics and how to distinguish it from other approaches to economics and, specifically, poverty research.⁵ Both research questions are approached by means of a hermeneutical investigation into Max-Neef's writings on the philosophy of science. The main result of the first study is the elaboration of barefoot economics as a well-defined and distinctive methodological approach to empirical economic research in the context of poverty-related phenomena.

The second study investigates barefoot economics in practice. In this sense, it is a study within barefoot economics. More precisely, the study draws on the barefoot economic research approach as established in the first study and applies it in empirical research practice. In order to do this, the methodological principles of barefoot economics are operationalised in an empirical case study conducting a five-year real-world experiment on bottom-up franchising in the non-formal education sector of the Mathare slums in

⁴ Please note that considering the two scientific studies as consecutive is a simplification with the aim of enhancing the intersubjective comprehensibility (Steinke 1999) of this treatise. Factually, this research process must be viewed as circular, not linear (see Flick 2009: 92). Within this circularity, the results of the second study are used to reflect on and validate the research results of the first study (see *section 4.5*).

⁵ Here, the term 'meta-economic' is used to indicate that the object of research is economics itself, and not the economy (cf. Kohr 1956; Schumacher [1973] 2011; Menger 1979).



Figure 3: Research agenda of this treatise (source: own illustration).

Nairobi. The content of the experiment addresses the research question of why bottom-up franchises are rarely observed at the base of the pyramid. In broader terms, the second study demonstrates how barefoot economics can be applied as a scientific research approach, as well as the performative impacts that the practice of barefoot economic research can have on existing poverty in the real world.⁶

Based on both scientific studies, the overall research question of this treatise is ultimately answered in a theoretically and empirically substantiated manner. *Figure 3* provides an illustrative overview summarising the research agenda of this treatise on barefoot economics.

⁶ The above-mentioned investigation into the *performativity* (Callon 1998) of barefoot economics in terms of poverty here refers to the capability of barefoot economics to perform poverty alleviation in the course of its empirical research practice.

1.4 Structure of the Treatise

This treatise consists of four chapters. In addition to a general introduction (*chapter 1*) and an overall conclusion (*chapter 4*), the main body of this treatise includes two chapters presenting the two scientific studies described above. *Chapter 2* presents the first study, dedicated to barefoot economics in theory, while *chapter 3* presents the second study, dedicated to barefoot economics in practice.

Chapter 1 includes four sections comprising the general introduction to this treatise. The first section (1.1) outlines the motivation behind this treatise, together with the overall research aim and its practical relevance. The second section (1.2) provides a brief overview of the research context and, by doing so, identifies the current state of poverty research in the economic sciences. The existing theoretical and methodological problems associated with poverty research in the economic sciences are also highlighted. The third section (1.3), describes the research agenda of this treatise by defining the research objective, the research gap addressed and the overall research question. It also provides a brief outline of the two scientific studies conducted to address the overall research question and sets out the specific research questions and approaches of both studies. Finally, the underlying research rationale of the treatise is presented. The fourth section (1.4) describes the structure of this treatise.

Chapter 2 contains four sections, presenting the first scientific study of this treatise. The first section (2.1) is the introduction to the theoretical-conceptual study conducted. This introduction refers to the general background, the specific research questions and the chosen research approach of the study, as well as providing a brief overview of the subsections in the sections of *chapter 2*. The second and third sections (2.2 and 2.3) together shape the main argument of the study. The fourth section (2.4) draws conclusions, summarising the line of argument and presenting the contributions and implications of the study, as well as its limitations.

Chapter 3 includes seven sections presenting the second scientific study of this treatise. The first section (3.1) is the introduction to the empirical case study conducted. This introduction deals with the general background, the specific research question and the chosen research approach of the study. A brief overview of the subsections in the sections of *chapter 3* is given at the end of the introduction. The second section (3.2) describes the case study's empirical field and the third section (3.3) clarifies the theoretical background to the case study. The fourth section (3.4) sets out the study's method and the fifth section (3.5) presents the empirical research findings, which are discussed in the sixth section (3.6). Lastly, a conclusion is drawn in the seventh section (3.7), summarising the course of the investigation and presenting the contributions and implications of the study, as well as its limitations.

Chapter 4 contains five sections, which make up the overall conclusion to this treatise. The first section (4.1) summarises the contents of *chapter 1* to *chapter 3*. Subsequently, the second section (4.2) presents the overall contributions and implications of this treatise, thereby giving an answer to the overall research question. The third section (4.3) deals with the overall limitations. The fourth section (4.4) provides an outlook on avenues for future research, followed by some final remarks in the fifth and last section (4.5).

2 The Principles of Barefoot Economics A Theoretical-Conceptual Groundwork

2.1 Introduction

The notion of 'barefoot economics' was first introduced by Manfred Max-Neef in his 1982 classic, From the Outside Looking In: Experiences in 'Barefoot Economics'. The acclaimed book deals with two development projects conducted by Max-Neef during the 1970s and tells their stories: "[t]he first is about the miseries of Indian and black peasants in the Sierra and coastal jungle of Ecuador[,] [and] [t]he second is about the miseries of craftsmen and artisans in a small region of Brazil" (Max-Neef [1982] 1992: 22). A year after the book's publication, Max-Neef was awarded the Alternative Nobel Prize for his poverty alleviation "through 'Barefoot Economics'" (Right Livelihood Foundation [1983b] 2020b: para. 1). From that point onwards, Max-Neef used the notion of barefoot economics on many occasions throughout his life to refer to his way of practicing economics (see e.g. Max-Neef [1988] 1991a: 102; Smith & Max-Neef 2011: 10). The question is: what characterises that approach to practicing economics? Or, more succinctly, what is barefoot economics? To make this question more applicable to scientific investigation, the following two sequential research questions are posed and answered in the course of this study:

- (1) How can barefoot economics be defined?
- (2) What renders barefoot economics distinct from other approaches to economics and, specifically, poverty research?

Regarding the selection of an appropriate research approach to address both research questions, it should be noted that Max-Neef was strongly committed to the 'linguistic turn' philosophies of the mid-20th century. This made him consider all definitions as essentially tautological and, thus, "perfectly meaningless" (Max-Neef [1988] 1991a: 100).¹ As a consequence, Max-Neef did not attach great importance to a sophisticated definition and theorisation of barefoot economics. Instead, his writings are enriched with metaphors, allegories, parables and anecdotes ripe for interpretation (Drekonja-Kornat 2010: 159). Although the interpretative imperative of Max-Neef's writings may be atypical in the field of contemporary economic research, this study attempts to examine the meaning of barefoot economics by means of a hermeneutical investigation (Paterson & Higgs 2005; McCaffrey et al. 2012) into Max-Neef's writings on the philosophy of science. This investigation, thereby, takes as its starting point the interpretation of (1) the metaphor of barefoot economics itself, and (2) the anecdote describing how Max-Neef became a barefoot economist.

Most publications mentioning the Max-Neefian term 'barefoot economics' interpret the metaphor of barefoot economics as a figurative description of participatory, economic field research in poor regions of the world (see e.g. Holden 1984; Lucena 2010; Imas et al. 2012; Balkema & Pols 2015; Göpel 2016; Smolski 2016; Kelley & Kester 2017; Weston et al. 2017; Madrueño & Martínez-Osés 2019; Flynn 2020; Alcoff 2020). Such an interpretation may stem from the fact that Max-Neef conducted fieldwork of this nature for more than a decade. It could also be based on cultural presuppositions equating 'bare feet' with poverty. While this interpretation of the metaphor of barefoot economics is undeniably correct to some extent, it does not distinguish barefoot economics substantially from other approaches to poverty research that exist within the

¹ For example: If 'A' is *defined* as the first letter of the alphabet, then what is a letter? If a letter is *defined* as a written character used in a society, then what is a written character and what is society? And so on (Derrida 1972; Habermann 2008). Hence, the final result of definitions seems always to be either an infinite regress or more likely – since every ordinary language has its limits – a tautological circle.

plethora of empirical social research – such as ethnography, action research, postcolonial studies or grounded theory. A more promising, strictly analytic interpretation was put forward by Cruz, Stahel & Max-Neef in 2009. According to them, the metaphor expresses the idea that "economics is in need of taking off [...] [its] shoes and touch the ground, [in a word:] becoming 'barefooted'" (Cruz et al. 2009: 2030). This gives rise to two sub-questions: (1) *What is meant by the 'shoes' of economics*? and (2) *What is meant by 'touching the* ground'?

In addition to using the term 'barefoot economics' as a metaphor, Max-Neef described barefoot economics in practical terms as "the economics that an economist who dares to step into the mud must practice" (Max-Neef 2010a, as cited in Goodman 2010: 40). To answer the obvious question of the actual nature of such a practice of economics, Max-Neef often used an anecdotal narrative in which he tells the story of how he became a barefoot economist. The anecdote – in the version as it appears in Smith & Max-Neef's 2011 book *Economics Unmasked* – reads as follows:

[O]ne day, in a village of the Andean Sierra, I [Manfred Max-Neef] was standing in the mud and in front of me. also in the mud, was standing a thin man, hungry and jobless, with five children, a wife and a grandmother. While we were looking at each other I was overwhelmed by a sudden consciousness that I lacked a language that could make sense in such a situation. My whole discourse as a conventional academic economist was absolutely inadequate for me to say anything significant. I was used to diagnosis and analysis, but I was not used to understanding. I knew all about poverty and had all the statistics, yet there I was, speechless, when looking poverty in the face. It became clear to me that I had to invent a new language. That was the origin of my 'principles of barefoot economics', and my rebirth as an absolute dissident of mainstream economics (Smith & Max-Neef 2011: 9-10).

In summary, the narrative outlines how Max-Neef's principle of barefoot economics arose from the fact that he was 'not used to understanding' and, therefore, had to 'invent a new language'. This leads to two further sub-questions: (1) *What does Max-Neef mean by the term 'understanding' in this context?* and (2) *What does he mean by 'the invention of a new language'*?

To explain the principles of barefoot economics, the following sections first put forward answers to the sub-questions arising from the metaphor and narrative of barefoot economics, as mentioned above. Finally, the two research questions of how to define and distinguish barefoot economics in a substantiated manner are answered. The line of argument begins with a brief review of the current prevailing economic methodology as established by Milton Friedman (section 2.2.1). Following that, positivism is identified as the underpinning for that economic methodology in terms of the philosophy of science (section 2.2.2). This positivist foundation of economics is contrasted with the antagonistic philosophical school of phenomenology, as laid down by Edmund Husserl (section 2.3.1). Max-Neef's barefoot economic perspective is analysed by putting it into the context of phenomenological philosophy (section 2.3.2). Finally, conclusions are drawn (section 2.4). These conclusions include a summary of the line of argument (section 2.4.1), a presentation of the main contributions and implications (section 2.4.2), and a brief note on the limitations of this study (section 2.4.3).

2.2 The 'Shoes' of Economics

2.2.1 Milton Friedman's Methodology of Positive Economics

Barefoot economics as an approach to economic research appears very different to the dominant methodology of economics. The latter was largely developed from the work of Nobel Prize winner and leader of the renowned Chicago School of Economics Milton Friedman (1912–2006) in his 1953 seminal essay, *The Methodology of Positive Economics*. Today, there is general consensus that Friedman's essay was "the most influential work on economic methodology of [the 20th] century" (Hausman 2008: 145).

In his essay, Friedman ([1953] 1966) argues in favour of hypotheses that can be tested empirically to enable the prediction of measurable economic phenomena not yet observed. A hypothesis of that kind "abstracts the common and crucial elements from the mass of complex and detailed circumstances surrounding the phenomena [...] and permits valid predictions on the basis of them alone" (ibid.: 14). Moreover, such hypotheses should ideally be tested "by experiments explicitly designed to eliminate what are judged to be the most important disturbing influences" (ibid.: 10).² This is why today's leading economists – such as the 2019 Nobel laureates Esther Duflo, Abhijit Banerjee and Michael Kremer – consider large-scale *randomised controlled trials* (*RCTs*) and related types of *lab-in-the-field experiments* as the methodological "gold standard" for economic research (Banerjee et al. 2019: passim).

Despite its dominance, Friedman's methodology inevitably attracted – and still attracts – criticism. For more than half a century, critics have found fault with the methodologically inherent discrepancies between reality and the assumptions on which hypotheses and, ultimately, economic theories³ are based (see Friedman [1953] 1966: 31). Friedman ([1953] 1966: 41) himself anticipated such criticism, noting the "perennial criticism of economic theory as unrealistic", and pre-empted it with an argumentative defence of his methodology. He points out that "[s]uch criticism is largely irrelevant" (ibid.: 41), because it stems from "confusion about the role of 'assumptions' in economic analysis" (ibid.: 40). Clarifying the role of assumptions, Friedman shows that unrealistic assumptions are an *inevitable result* of the necessary process of *abstraction* involved in hypothesising. Hypotheses that can be empirically tested can only be formed if common and crucial features are abstracted from the

² Experimentation can be defined as a research method involving an *empirical intervention stimulus* (see section 3.4.3).

³ Friedman ([1953] 1966: 8) defines a theory as "a body of substantive hypotheses".

complex reality surrounding economic phenomena. These features must be *assumed* to provide a complete description of reality, even though they do not, in actual fact, fully represent reality (see also Rappaport 1996). As Friedman ([1953] 1966: 40) states: "It is frequently convenient to [assume][...] that the phenomena [...] desired to predict behave in the world of observation *as if* they occurred in a hypothetical and highly simplified world containing only the forces that the hypothesis asserts to be important." For that reason, the assumptions on which hypotheses are based are often termed '*as if*' assumptions. In the course of his argument, Friedman also explains why efforts to draw more accurate descriptions of reality tend to be counterproductive to the formation of hypotheses and, ultimately, economic theories:

A theory or its 'assumptions' cannot possibly be thoroughly 'realistic' in the immediate descriptive sense so often assigned, to this term. A completely 'realistic' theory of the wheat market would have to include not only the conditions directly underlying the supply and demand for wheat but also the kind of coins or credit instruments used to make exchanges; the personal characteristics of wheat-traders such as the color of each trader's hair and eves, his antecedents and education, the number of members of his family, their characteristics, antecedents, and education, etc.; the kind of soil on which the wheat was grown, its physical and chemical characteristics, the weather prevailing during the growing season; the personal characteristics of the farmers growing the wheat and of the consumers who will ultimately use it; and so on indefinitely. Any attempt to move very far in achieving this kind of 'realism' is certain to render a theory utterly useless (Friedman [1953] 1966: 32).

Furthermore, Friedman argues that:

the relevant question to ask about the 'assumptions' of a theory is not whether they are descriptively 'realistic,' for they never are, but whether they are sufficiently good approximations for the purpose in hand. And this question can be answered only by seeing whether the theory works, which means whether it yields sufficiently accurate predictions (ibid.: 15).

As the excerpts above indicate, Friedman believes that an increasing level of abstraction necessarily causes the assumptions of a hypothesis (or theory) to become an inaccurate descriptive representation of reality; however, at the same time, the hypothesis (or theory) is capable of making predictions based on a decreasing number of explanatory variables. Ultimately, hypotheses that can be empirically tested need to be able to predict economic phenomena based on as few factors as possible (ibid.). In the final analysis, related criticism is irrelevant, because hypotheses can be exclusively judged by the conformity of their predictions with reality (ibid.).

In summary, Friedman's methodology of economics demonstrates a very high logical consistency on the basis of which – despite its critics – it has been the dominant approach to economic research for more than half a century.

2.2.2 Positivism in Economics

Having outlined the logical consistency of Friedman's economic methodology, it should be recognised that barefoot economics does not intend to challenge the current dominant methodology of economics itself, but rather its underlying *presuppositions from a philosophy of science perspective* (which constitute the 'shoes' of economics). As the title of Friedman's essay, *The Methodology of Positive Economics*, indicates, Friedman's perspective was one of *positive economics*. Thereby, "Friedman associated with the name of positive economics [...] [an] economics which follows the ideals of positive science as it was understood by *positivists*" (Nekrašas 2016: 295, own emphases).

Positivism is a philosophical tradition which can be regarded as the culmination of 18th century Enlightenment thought (Horkheimer & Adorno [1944] 2003). Since the 18th century. positivism has been advocated by a wide range of different philosophers, including David Hume (1711-1776), Auguste Comte (1798-1857), John Stuart Mill (1806-1873), Ernst Mach (1838-1916), Moritz Schlick (1882-1936), Rudolf Carnap (1891-1970) and to some extent – Karl Popper $(1902-1994)^{\hat{4}}$. Although diverse variants of positivism have been developed over time, all positivists agree on the fundamental tenet that science should solely be based on knowledge which is 'positive', i.e. knowledge grounded in "what is given" in the sense of empirical facts (Schlick [1932] 1948: 480).⁵ This means metaphysical statements are rejected because they are not, by definition, grounded in empirical facts (ibid.). Moreover, positive knowledge is supposed to be value-free because the observation of empirical facts does not involve "any particular ethical position or normative judgments" (Friedman [1953] 1966: 4). To ensure that knowledge is value-free, only logical and/or mathematical analyses of empirical facts are considered as appropriate (Nekrašas 2016: 4). Based on the supposed value-free

⁴ Please note that there was, and still is, great controversy about whether or not Karl Popper was a positivist (Adorno et al. 1972). Popper himself denied being a positivist, since he tried to demark his philosophy of *critical rationalism* from the *logical positivism* of the 'Vienna Circle', which was the dominant philosophy of science during Popper's lifetime during the 1940s and 1950s. The major difference between the two schools of philosophy is that logical positivists define testability as *verifiability*, while Popper defined it as *falsifiability* (ibid.). However, in terms of the broader criteria of positivism presented in this section, Popper can be attributed to be a positivist thinker insofar as the criteria can be said to be in accordance with his critical rationalism.

⁵ Two examples illustrate what positivists mean by empirical facts. Positivists would, for example, consider the following statements as empirical facts: "Coal is black" (McDonald 2012: 112). "Elephants exist, unicorns do not" (ibid.: 107).

nature of positive knowledge, positivists further argue in favour of *scientific objectivity*, which Karl Popper ([1935] 1972: 44) described as follows: "the objectivity of scientific statements lies in the fact that they can be inter-subjectively tested". Referring to this concept of objectivity, Friedman ([1953] 1966: 4) postulated that "positive economics is, or can be, an 'objective' science".

Ultimately, positivism is regularly identified as the philosophy of science upon which modern economics rests (e.g. Seligman 1969; Schumacher [1973] 2011; Caldwell 1980; Katouzian 1980; Boland 1991; Lawson 1997). The following section highlights how barefoot economics can be considered to contradict positive economics in as much as it is based on an antagonistic philosophical foundation – namely *phenomenology*.

2.3 'Touching the Ground' of Phenomena

2.3.1 Edmund Husserl's Phenomenology

Phenomenology is "the study of the phenomena themselves" (Husserl [1910] 2002: 276). Its prominent founding father was the German philosopher Edmund Husserl (1859–1938), who introduced phenomenology as a philosophy *sui generis* in his *Logical Investigations* from 1900/1901 (e.g. Bello 2009; Schnell 2019).⁶ Thereby, Husserl ([1950] 1998: 5) constituted phenomenology as a radical antithesis to what he called the "naive objectivism" of the positive sciences, which underestimated the role of the human

⁶ Please note that phenomenology "as a manner or style of thinking" existed long before it was elaborated as a philosophy *sui generis* by means of Husserl's work (Merleau-Ponty [1945] 2005: viii). Notable earlier phenomenological thinkers include, for example, Johann Wolfgang von Goethe (1749–1832), Georg Wilhelm Friedrich Hegel (1770–1831), Søren Kierkegaard (1813–1855), Karl Marx (1818–1883) and Friedrich Nietzsche (1844–1900) (Merleau-Ponty [1945] 2005: viii; on Goethe, see Heinemann 1934; Seamon & Zajonc 1998; Simms 2005). Since Husserl's collected works alone amount to around 40,000 pages, it is only possible to provide a partial overview of phenomenological philosophy in the following sections.

person as the researching *subject*. To Husserl, 'what is given' are not empirical facts, but rather phenomena as they appear to people in their 'lived experience' (original German wording: *Erleben*). In other words, phenomena as they appear in lived experiences are what constitutes our given *reality*. Therefore, the ultimate goal of science should not be the production of knowledge based on empirical facts, but rather the attainment of an *understanding of phenomena*.⁷ As Husserl stated: "we must go back to the 'things themselves'" (Husserl [1900/01] 2001: 168).

The basic goal of any phenomenological philosophy is to come into immediate contact with the world, 'to get at the things themselves,' regardless of whether these things are physical or mental, numbers or deities, feelings or values. ...; in its attempt to get at the things themselves it [phenomenology] refuses to take for granted the validity of any presupposed conceptual schema and even of the findings of any positive science whatever (Schutz 1957: 306).

More precisely, by 'getting at the things themselves' a person can grasp the *meaning* or *essence* of phenomena, i.e. grasp what makes a phenomenon what it is (Husserl [1910] 2002: 273).⁸ Therefore,

⁷ The notion of 'understanding' is also quintessential for another philosophical tradition, namely *hermeneutics* (Schnell 2019). Within hermeneutics – having its main representatives in Friedrich Schleiermacher (1768–1834), Wilhelm Dilthey (1833–1911) and Hans-Georg Gadamer (1900–2002) – the notion of understanding is closely related to the notion of *interpretation*. This chapter, however, solely refers to the notion of understanding according to "the genuine *phenomenological* meaning of the term 'understanding'" (ibid.: 68, own trans.).

⁸ With regard to the relationship between the notions of 'meaning' and 'essence' in phenomenology, please note the following explanations from Husserl student Hedwig Conrad-Martius (1951: 10, as cited in Schnell 2019: 48, own trans.): "To the phenomenologist [...] the world is full of a priori meaningfulness. 'Meaning' is here not used in a teleological way, in which the real world or the course of the real world possesses a final historical or ahistorical meaning and purpose [Greek: *telos*]. 'Meaning' is here equal to 'essence' [Greek: *eidos*]".
phenomenology has also been defined as "the study of essences" (Merleau-Ponty [1945] 2005: vii). However, such an 'understanding of phenomena' can only arise intuitively from lived experience if obscuring preconceptions do not interfere. Preconceptions of this kind must be identified and eliminated by means of the genuine phenomenological method of epoché (Greek for 'bracketing'). The "epoché [...] frees our gaze" for the 'things themselves' (Husserl [1936] 1970: 241, own italics). Husserl introduced the epoché which he also calls "phenomenological reduction" (Husserl [1913] 1983: 66) - in his second major work, Ideas, in 1913. He defines it as "the method of parenthesizing" in which "[w]e put out of action the general positing which belongs to the essence of the natural attitude" (ibid.: 60-61, italics in original removed). This natural attitude comprises that which is taken for granted within the experiences of everyday life. In his later 1936 work, The Crisis of European Sciences and Transcendental Phenomenology, Husserl refined his notion of the 'natural attitude' by introducing his concept of the *life-world*, which is defined as the 'world of the natural attitude' and is described as follows:

[T]he life-world – the 'world for us all' – is identical with the world that can be commonly talked about. Every new apperception leads essentially, through apperceptive transference, to a new typification of the surrounding world and in social intercourse to a naming which immediately flows into the common language. Thus the world is always such that it can be empirically, generally (intersubjectively) explicated and, at the same time, linguistically explicated (Husserl [1936] 1970: 209–210).

In this way, Husserl's late work takes up the insights of contemporary philosophy's *linguistic turn* that came about in the wake of Ludwig Wittgenstein's 1921 *Tractatus Logico-Philosophicus*, in which Wittgenstein ([1921] 2002: 68) concluded that "*[t]he limits of* *my language* mean the limits of my world".⁹ As a result, Husserl elaborates the phenomenological method as a "life-world epoché" (Husserl [1936] 1970: 137) that places in abeyance what is posited by "the naïve [natural] attitude of world-life, [where] everything is [...] bound to what can be named, asserted, described in common language" (ibid.: 209). In this respect, Husserl suggests that we cannot 'go back to the things themselves' while our lived experience is 'led astray' by language.

It is easy to see that even in (ordinary) human life, and first of all in every individual life from childhood up to maturity, the originally intuitive life which creates its originally self-evident structures through activities on the basis of sense-experience very quickly and in increasing measure falls victim to the *seduction of language* (Husserl [1939] 1989: 165).

Accordingly, Husserl ([1939] 1975) argued that the primordial intuitive understanding of phenomena is pre-predicative and prelinguistic, i.e. comparable to the way in which an infant experiences reality before it has learned to think in words. As the famous French phenomenologist Maurice Merleau-Ponty ([1945] 2005: xvii) said, it rests "[i]n the silence of primary consciousness" that transcends the realm of language. In contrast, knowledge is always related to predicative sentences (statements) and, consequently, is locked in language. Therefore, the 'empirical facts' referred to by positivists can be considered to be nothing more than linguistic expressions of phenomena having presuppositions in common language.¹⁰ In

⁹ Please note in this context that when we think in words, language constitutes the *conditions of possibility* of what we can think and how we can disclose the world by means of our thinking.

¹⁰ As an analogy to the linguistic presuppositions of 'empirical facts', we could say that positivists believe in what they have seen with their own eyes without taking account of the presuppositions that arise from the constitution of their eyes (compare Schumacher [1973] 2011). Consequently, positivism can be deemed to encourage "word fetishism" (Horkheimer & Adorno [1944] 2003: 9, own trans.),

light of this, Husserl ([1931] 1982: 157) had already previously concluded in his 1931 *Cartesian Meditations*: "Positive science is a science lost in the world. I must lose the world by epoché, in order to regain it by a universal self-examination." With his concept of *self-examination*, Husserl expresses the shift from the natural attitude towards a "*phenomenological attitude*" (Husserl [1936] 1970: passim, own italics) where "the knower[][is] reflecting upon himself and his knowing life in which all the scientific structures that are valid for him occur purposefully, are stored up as acquisitions, and have become and continue to become freely available" (ibid.: 97–98). After the epoché, the knower returns to a natural attitude of a self-examined world-life, which is no longer naïve but 'phenomenologically enlightened' (Overgaard 2004: 47).¹¹

2.3.2 Manfred Max-Neef's Phenomenological Thought

The influence of phenomenology on Manfred Max-Neef is evident in a number of his writings on the philosophy of science (Max-Neef [1982] 1992, [1988] 1991a, [1989] 1991b, 2005a, 2007, 2009).¹² The following section attempts to explain the principles of barefoot economics by putting Max-Neef's perspective on the philosophy of science in the context of Husserl's phenomenological philosophy.

Firstly, barefoot economics' claim of 'touching the ground' can be regarded as the Max-Neefian equivalent to Husserlian phenomenology's dictum of 'getting at the things themselves'. This

which – to use a Wittgensteinian formulation – no longer "*battle[s]* [...] the bewitchment of our intelligence by means of language" (Wittgenstein [1953] 1986: 47, own italics) but rather *worships* it. From a phenomenological perspective, this is problematic: it is not possible to grasp the essence of something through words alone, as the essential quality remains invisible to the eye.

¹¹ In this sense, phenomenology is also intended to pave the way for a new Enlightenment that counters the culmination of 18th century Enlightenment thought in positivism (see section 2.2.2; see also Horkheimer & Adorno [1944] 2003).

¹² The influence of phenomenological philosophy on Max-Neef is particularly apparent in his 1982 "theory of space-time disruptions", which investigates human beings' lived experience of space and time (Max-Neef [1982] 1992: 139).

similarity is reinforced by the fact that Husserl's dictum has also been described as the appeal "to come into immediate contact with the world" (Schutz 1957: 306). Using Merleau-Ponty's ([1945] 2005: 449) formulation, it means to get a "grip upon the world". All these phrases intend to express phenomenology's ultimate goal: to attain an understanding of phenomena.

Like others, Max-Neef also distinguished the effort to understand phenomena from the acquisition of knowledge. He often gave the following example:

Suppose that you have studied everything there is – from the anthropological, cultural, psychological, biological and biochemical points of view – about the phenomenon of love. You are an erudite. You know everything that can be known about love, but you will never understand love unless you fall in love. This principle is valid for all human systems, although it is almost always overlooked (Max-Neef [1988] 1991a: 102).

The example demonstrates a principle incorporating at least two aspects. Firstly, "knowledge [about phenomena] is not the road that leads to [an] understanding [of phenomena]" (Max-Neef 2009: 18). Secondly, an understanding of phenomena can only be attained through lived experience, i.e. if the human being experiences phenomena in the sense of 'living through them'. The principle is, in this respect, fully in line with the fundamental perspective of phenomenology.

With respect to barefoot economics and how it addresses the phenomenon of poverty,¹³ Max-Neef ([1988] 1991a: 102) claimed that "if we have so far been unable to eradicate poverty, it is because we know too much about it, without understanding the essence of its existence". The statement indicates Max-Neef's Husserlian

¹³ Please note in this context that Max-Neef considered poverty-related phenomena as the essential phenomena of economics (compare section 1.2 and section 3.6.4).

interpretation of understanding as grasping the essence. Moreover, the statement can be regarded as a critique of the dominant positive economics approach insofar as its underlying philosophy of science leads to a methodology in which "economists study and analyze poverty in their nice offices, have all the statistics, make all the models, and are convinced that they know everything that you can know about poverty[,][b]ut [...] don't understand poverty" (Max-Neef 2010a, as cited in Goodman 2010: 40). To truly understand the phenomenon of poverty, it must be experienced at first hand. Max-Neef (2019, own transcript) argued that, instead of shutting themselves away in academic ivory towers, nascent economists "should go before they graduate and live six months with an extreme poor family and contribute to their work".

However, Max-Neef considered lived experiences as necessary but not wholly sufficient to attain an understanding of phenomena, because "something happens to render us immune to experience" (Max-Neef [1989] 1991b: 107). With reference to Wittgenstein, Max-Neef identifies this 'something' as language (ibid.). He writes: "the point is that we are [...] trapped by language. Language is a form of imprisonment. The way in which we use words or concepts influences and sometimes even determines not only our behavior but our perceptions as well." (ibid.: 108). As such, Max-Neef concurs with Husserl's view that our "sense-experience [...] falls victim to the seduction of language" (Husserl [1939] 1989: 165). To overcome the cognitive distortion of our experience by language and, finally, to achieve an understanding of phenomena, Max-Neef ([1982] 1992: 114–115) was convinced that what was required was precisely not "a richer vocabulary" or a "'progressive' terminology", but rather the opposite: "an adequate pruning of key words" (Max-Neef [1988] 1991a: 99). He proposed "the pruning of language" (ibid.: 101) as a method to 'prune' those words which distort the perception of the phenomenon to be understood by making someone think based on obscuring preconceptions. The result would be "a new language that opens the door of understanding" (Max-Neef 2009: 21), i.e. a pruned language that opens the door

to the 'things themselves'. Max-Neef ([1989] 1991b, 2009) also described this as a language that is coherent with reality.¹⁴ To illustrate the principle behind the method of linguistic pruning, Max-Neef often used the allegory of an orchard:

The principle behind the act of pruning should be clear to anyone who has ever been interested in orchards. Through pruning we will achieve more and better from less. Fewer branches and leaves will allow more light to be absorbed and thus produce better fruits. In the case of a language, the pruning of chosen words will force us inevitably into higher degrees of clarity (Max-Neef [1988] 1991a: 99).

The methodological pruning of language to understanding phenomena – let us call it the 'orchard method' – can easily be identified as a version of the phenomenological method of the lifeworld epoché, by equating Max-Neef's notion of 'pruning' with Husserl's notion of 'bracketing' or 'parenthesising'. To be more precise, Max-Neef's method operationalises the philosophical concept of phenomenological reduction as a specific linguistic reduction.

¹⁴ The following Wittgensteinian example clarifies the intended meaning of a language that is coherent with reality: "Let us imagine a language [...]. The language is meant to serve for communication between a builder A and an assistant B. A is building with building-stones: there are blocks, pillars, slabs and beams. B has to pass the stones, and that in the order in which A needs them. For this purpose they use a language consisting of the words 'block', 'pillar', 'slab', 'beam'. A calls them out; - B brings the stone which he has learnt to bring at such-and-such a call. - Conceive this as a complete primitive language." (Wittgenstein [1953] 1986: 3). The described primitive language solely consisting of 'shape-words' can be interpreted as a language that is coherent with the phenomenon of building as it appears to A and B in their lived experience. Analogously, we can for example imagine a language solely consisting of 'colourwords' as coherent with the phenomenon of painting; or a language solely consisting of 'number words' as coherent with the phenomenon of counting. For the latter two languages, we can also imagine an incoherence with the phenomenon of building and, thus, an incoherence with the given reality of A and B.

Thereby, Max-Neef also acknowledges the self-examination which arises from the shift towards the phenomenological attitude by means of epoché.

Having carried out the exercise of pruning and becoming aware of the limits of knowledge on the one hand, and of the differences between knowledge and understanding on the other, there is no harm in going back to my old words, even to my old language. If I do so now (and it would be foolish if I did not), both the words as well as the language to which they conform will no longer be masks behind which ignorance remains hidden but will become fertile spaces for the permanent progress toward intellectual wholeness (Max-Neef [1988] 1991a: 103).

Having demonstrated the similarities between Husserl's and Max-Neef's philosophy of science perspectives in order to substantiate barefoot economics with a phenomenological foundation, the following analysis highlights a major difference. While Husserl's perspective was radical in its attempt to replace positivism with phenomenology as a monistic paradigm within the philosophy of science, Max-Neef advocated a paradigm pluralism based on Niels Bohr's principle of complementarity, that is contraria sunt complementa (Max-Neef 2005a).¹⁵ Thereby, Max-Neef was convinced that the incommensurable opposites of positive economics and barefoot economics are not substitutive but are, in fact, complementary. While positive economics aims to create positive knowledge, barefoot economics aims to create phenomenological understanding. Both are supposed to be desirable objectives. However, the objective of understanding has not yet been given equal weight in modern economics. Max-Neef wrote about this issue in his 2009

¹⁵ Following Thomas Kuhn (1974: 460), the term 'paradigm' denotes "all the shared commitments of a scientific group" about how scientific research should be executed (see also Kuhn [1962] 1996).

paper, *From Knowledge to Understanding*, in more general terms, stating:

[Due to the dominance of positive science,] [w]e have arrived at a point in our human evolution where we *know* a lot, but we *understand* very little. ... We have never in all of our existence, accumulated more knowledge than during the last 100 years ..., but ... we suddenly have the feeling that something is missing. ... At least we have reached a point at which, [some of us] (many conventional academics notwithstanding) ... are finally becoming aware that knowledge is not enough, and that we have to learn how to attain understanding in order to achieve ... the completeness of our science. We are, perhaps, beginning to realise that knowledge without understanding is hollow, and understanding without knowledge is incomplete (Max-Neef 2009: 18).

2.4 Conclusion

2.4.1 Summary

The above investigation used a hermeneutical approach with the aim of answering the research questions of how to define barefoot economics and what renders barefoot economics distinct from other approaches to economics and, specifically, poverty research.

This scientific study has shown that the current prevailing methodology of economics was introduced by Milton Friedman in the 1950s. Friedman's methodology proposes that economics should aim to provide accurate predictions about economic phenomena. To achieve this ultimate goal, economic research should: (1) work with hypotheses built by means of abstraction and 'as if' assumptions; and (2) test these hypotheses empirically by means of experimentation. The philosophy of science from which Friedman's view on economics is derived is deemed to be positivism. Positivism, in turn, has in its three centuries-long tradition established an ideal of scientific objectivity according to which science should solely be based on value-free knowledge that can be tested intersubjectively, and is gained from empirical facts and their logical and/or mathematical analysis. Positivism's ideal of scientific objectivity has, in particular, been criticised as a 'naïve objectivism' by the antagonistic philosophical school of phenomenology, as developed by Edmund Husserl at the beginning of the 20th century. In contrast to positivism, phenomenology proposes that science should 'go back to the things themselves', i.e. attempt to understand phenomena as they appear in lived experience. This understanding of phenomena is defined as the grasp of the essence or meaning of phenomena and is thought to be achievable only through the genuine phenomenological method of epoché. In its variant of the 'life-world epoché', the method aims to identify and eliminate all obscuring preconceptions which occur from the common language that constitutes our life-world.

Against this background, barefoot economics has been identified as an economics that takes off the 'shoes' of positive economics and engages itself in the philosophical foundation of phenomenology. Thereby, barefoot economics' effort to 'touch the ground' can be regarded as the Max-Neefian equivalent to Husserl's dictum of 'getting at the things themselves'. In this sense, barefoot economics can be described as a phenomenological study of essences – whereby its main interest lies de facto in the understanding of poverty-related phenomena as they appear in lived experience. Barefoot economics addresses this ultimate goal by its very own method of linguistic pruning. This methodological pruning of language can be viewed as an operationalisation of the phenomenological method of the life-world epoché, aiming to invent a new language that opens the door to understanding – a language that is coherent with reality.

2.4.2 Contributions and Implications

The above investigation into the meaning of barefoot economics and its principles has sought to explain that barefoot economics is not simply a figurative term for any kind of participatory economic field research in poor regions of the world, but rather a clearly distinguishable phenomenological approach to economics based on its own foundation in the philosophy of science.

Regarding the research questions of how to define barefoot economics and what renders it distinct from other approaches to economics and, specifically, poverty research, the conducted investigation has provided profound insights. A primary result of this scientific study is the suggested definition of barefoot economics as follows: barefoot economics is an approach to economics which seeks to understand the essence or meaning of poverty-related phenomena by means of lived experience and a methodological pruning of language. Defined in this way, barefoot economics is starkly differentiated from positive economics and its ultimate goal of accurate predictions by means of hypotheses testing. Moreover, this definition of barefoot economics enables it to be distinguished from other non-positivist methodological approaches in the realm of empirical social research and, specifically, poverty research. Barefoot economics (as a phenomenological economics) is, for example, distinct from (1) ethnography which, based on social constructivism (Dutta 2014), tries to "grasp the native's point of view, his relations to life, to realize his vision of his world" by means of 'going native' (Malinowski 1922: 25); (2) postcolonial studies which, based on poststructuralism (Moore-Gilbert et al. 1997), try to deconstruct the hegemonic performativity of contemporary discourses on subaltern groups by means of critical discourse analyses (e.g. Said 1978; Spivak 1988; Hall 1997); (3) action research which, based on pragmatism (Greenwood & Levin 1998), tries to improve living conditions by means of problem-solving actions (Lewin 1946); and (4) grounded theory which, based on symbolic interactionism (Aldiabat & Navenec 2011), tries to construct data-grounded theories by means of comparative data analyses (Glaser & Strauss 1967). Although barefoot economics may share similarities with all these non-positivist research approaches, it is rendered distinct by its unique phenomenological foundation and methodology.

In more general terms, this study has highlighted the potential of barefoot economics to be the blueprint for *complementing positive economics* in line with the following model (see *Figure 4*):



Figure 4: Dialectic of Positive Economics and Barefoot Economics (source: own illustration).

Although positive economics and barefoot economics approach reality from incommensurable paradigmatic perspectives, they complement each other in a dialectical manner. To overstate the case aphoristically, it could be said that positive economics is about providing answers that fit reality, while barefoot economics is about posing questions that fit reality. In reaching phenomenological understanding, the barefoot economist ensures a language that is coherent with his or her lived experience of economic phenomena. Within the limits of that language, the methodology of positive economics can be used to produce scientific knowledge and make predictions.¹⁶ On the downside, it should be acknowledged that

¹⁶ The proposition made here corresponds to the analysis of the notion of 'truth' made by Husserl's most famous student, Martin Heidegger ([1927] 1996, [1942] 1947, [1943] 1988). Heidegger asserted that phenomenology regards truth as

the barefoot economist will always enter the field of research with positive prior knowledge that needs to be pruned.

Since contemporary positive economics is seldom interested in questions of language, and barefoot economics – as mentioned at the outset of this study – has not yet entered the academic discourse in the scientific discipline of economics, the present research implies a reason rooted in the philosophy of science to challenge this status quo.

2.4.3 Limitations

The limitations of this scientific research result from its hermeneutical approach to the meaning of barefoot economics. Considering the concept of the *hermeneutic circle*, interpretations of texts can never be 'objective' (in a positivist sense) because they will always be based on pre-interpretations ('fore-meanings'), which arise from facticity of the interpreter (Gadamer [1960] 2004).¹⁷ Hence, hermeneutical investigations cannot be tested intersubjectively but can only be *intersubjectively comprehensible* (see Steinke 1999). To validate hermeneutical investigations, the *argumentative validation of interpretations* (Mayring 2016) has been proposed and used as a

the unconcealedness (Greek: aletheia) of essences, while positivism associates truth with the correctness (Greek: orthotes) of statements. With regard to these two differing concepts of truth, Heidegger [[1943] 1988: 34, own trans.) claimed that "[t]ruth as correctness of statement is quite impossible without truth as unconcealedness [...] [because] that to which the statement must direct itself, in order to be correct, must already be unconcealed". In other words, "correctness presupposes unconcealedness" (ibid.: 118, own trans., italics in original removed), since the "condition of the possibility that statements can be true or false [respectively correct or incorrect]" is a language that coheres with reality (Heidegger [1927] 1996: 208; compare also section 2.3.1).

¹⁷ A major manifestation of this facticity is the specific language used by the interpreter (Gadamer [1960] 2004). The limits inevitably set by these circumstances become apparent at the end of this treatise (see section 4.5). To transcend the limits of language, *chapter 3* moves from 'saying' to 'showing' (compare Wittgenstein [1921] 2002); i.e. *chapter 3* no longer only *talks about* barefoot economics, but rather is an image of barefoot economic research itself.

validation strategy in this research study. In doing so, the meaning of barefoot economics has been theoretically derived from a consistent line of argument (see *section 2.1*). Finally, this investigation should be considered as a *proposal* on how to interpret the meaning of barefoot economics substantiated by philosophy of science. The applicability of the proposed principles of barefoot economics in research practice and related performative impacts on poverty are examined in the following scientific case study (*chapter 3*).

3 Experiences in Barefoot Economics An Empirical Case Study

3.1 Introduction

The first scientific study of this treatise elaborated barefoot economics as a well-defined and distinctive empirical research approach for poverty-related phenomena. In this respect, the first study serves as the groundwork for the following empirical case study, which applies the established barefoot economic approach in scientific research practice. The study was conducted in the research setting of an own independent development project: the 'MPITO[®] project'. The project was carried out in the Mathare slums of Nairobi (Kenya) from January 2015 to March 2020, and performed a multi-year real-world experiment on bottom-up franchising with microentrepreneurs from the slums' non-formal education sector.

From a theoretical perspective, the conducted experiment was initially derived from Prahalad's *BoP concept*. As outlined in *section 1.2*, the BoP concept has become the prevailing approach in the economic sciences on how to alleviate poverty. Having conceptualised the 'base of the pyramid' as a low-income population of around four billion people generally excluded from global capitalism, the BoP concept promotes an "inclusive business agenda" (Casado Cañeque 2015: 5), in which the base of the pyramid is economically served through capitalist entrepreneurship on a large scale. To implement the suggested inclusive business agenda in practice, economists have developed numerous theoretical models in recent years (see e.g. Kolk et al. 2014; Casado Cañeque & Hart 2015). One of the most acclaimed is *microfranchising* (e.g. Fairbourne 2006, 2007; Gibson 2007; Kistruck et al. 2011). Microfranchising is defined as the systematisation and replication of

microenterprises at the base of the pyramid with the intention of alleviating poverty (Fairbourne 2006). As empirical research has indicated that the performance drivers of microfranchising are significantly affected by the existing institutional framework at the base of the pyramid (e.g. Kistruck et al. 2011), microfranchising has been increasingly called for as a variant of bottom-up franchising (see Fairbourne 2007; Henriques & Herr 2007; Munoz et al. 2010; Kistruck et al. 2011: Lawson-Lartego 2016). Within this bottomup approach, the franchised microentrepreneurs are supposed to co-create a replicable business model based on collective decision making (ibid.). While bottom-up franchising has been championed as a model for scaling poverty alleviation efforts (ibid.), few empirical cases of bottom-up franchises at the base of the pyramid have been observed in practice (Henriques & Herr 2007). This situation may have led economists to question "why [...] bottom-up franchises are hardly observed. [...] One problem of bottom-up franchises will be collective decision making. However, this cannot be the entire story. [...] Further research is desirable." (Hendrikse & Windsperger 2012: 9). However, economists have been unable to provide adequate answers by means of positive economics. Consequently, by means of barefoot economics, this scientific case study examines the following research question:

Why are bottom-up franchises rarely observed at the base of the pyramid?

The intention is not to give causal explanations but to *understand* why the phenomenon of the bottom-up franchise is rarely observed at the base of the pyramid (compare *section 2.3*). The barefoot economic method of linguistic pruning is applied to identify and eliminate those preconceptions that may bias the perception of relevant phenomena. Lived experiences are gained from the MPITO[®] project and its bottom-up franchise experiment in the slums. Approaching the posed research question by means of a barefoot economic research design, therefore, makes a scientific contribution through the substantive answer provided and also contributes to the more general objective of demonstrating the

applicability of barefoot economics in the practice of empirical research.

The scientific case study is presented as follows. Firstly, a description of the empirical field, in which the case study was conducted, is given (section 3.2). After a brief note on the conducted field selection (section 3.2.1), an account of the phenomenon of slums in general (section 3.2.2), the specific characteristics of the Mathare slums (section 3.2.3) and the latter's non-formal education sector (section 3.2.4) is provided. Secondly, the theoretical background of the case study is presented (section 3.3). As explained in a brief preliminary remark (section 3.3.1), Joseph Schumpeter's Theory of Economic Development (section 3.3.2), C. K. Prahalad's BoP concept (section 3.3.3), and Jason Fairbourne et al.'s concept of microfranchising and its specific manifestation as bottom-up franchising (section 3.3.4) are outlined. This shows how the concept of bottom-up franchising can be derived from the BoP concept, and how, in turn, the BoP concept can be derived from Schumpeter's development theory. Thirdly, the method of the case study is clarified (section 3.4). After an indication of the appropriateness of the barefoot economic methodology for answering the posed research question (section 3.4.1), the applied methods of linguistic pruning (section 3.4.2) and real-world experimentation (section 3.4.3) are set out. Subsequently, the design of the MPITO® project (section 3.4.4.1), its sampling method (section 3.4.4.2) and data collection method (section 3.4.4.3), as well as the issue of data verbalisation (section 3.4.4.4), are described. The research results of the real-world experiment are then presented using methodologically pruned language (section 3.5). After some general observations (section 3.5.1), the findings are structured according to the microfranchise performance drivers of 'branding' (section 3.5.2) and 'standardisation' (section 3.5.3). The subsequent section discusses the results by returning to the previously-used scientific terminology and reflecting on the results in the context of the related theoretical background (section 3.6). After some preliminary remarks (section 3.6.1), the results are discussed in light of Niklas Luhmann's concept of inclusion/exclusion (*section 3.6.2*), Karl Polanyi's concept of embeddedness (*section 3.6.3*) and Manfred Max-Neef et al.'s Theory of Human Scale Development (*section 3.6.4*). Lastly, key conclusions are drawn (*section 3.7*), including a brief summary of the case study (*section 3.7.1*), its main contributions and implications (*section 3.7.2*), and the methodological limitations of this scientific research (*section 3.7.3*).

3.2 Field Description

3.2.1 The Selection of the Field

The fieldwork for this scientific case study was conducted in the non-formal education sector of an area in Kenya's capital city Nairobi, the Mathare slums.

From a theoretical perspective, the field selection responds to poverty researchers' frequent call for broader empirical data on the economic reality within poor regions of the African continent (see e.g. Egri & Ralston 2008; Kolk & van Tulder 2010; Kolk & Lenfant 2012; Kolk et al. 2014). From a more practical perspective, Nairobi can be seen as a particularly interesting location to implement the BoP concept's inclusive business agenda, as the city incorporates the contrasting ends of the global income pyramid (see *Figure 6* in *section 3.3.3*) in geographical proximity.

3.2.2 The Phenomenon of Slums

Today, around 60 % of Sub-Saharan Africa's urban population and around one billion people worldwide are forced to live in slum areas (UN-Habitat 2016).

"Slums are the most deprived and excluded form of informal settlements characterized by poverty" (UN-Habitat 2015: 1). Informal settlements can be defined as "[residential] areas where housing is not in compliance with current planning and building regulations" (United Nations 1997: 43). In slums, which are 'the most deprived and excluded form' of those residential areas, households lack at least one of the following: (1) access to improved drinking water; (2) access to improved sanitation facilities; (3) sufficient living area; (4) durable housing; and (5) secure tenure (UN-Habitat 2015). Accordingly, slum dwellers suffer from related resource deprivations which characterise the phenomenon of poverty (see *section 1.1*).

In more practical terms, slums are usually spatially segregated, large agglomerations of shacks or huts with a high population density in urban areas. Slum dwellers usually face living environments lacking in both basic infrastructure, such as workable roads, sewerage, power grids and water supply networks, and public services, such as legal security and protection, waste management, healthcare and educational services. Slum dwellings are often dilapidated and built in a makeshift manner using construction materials such as corrugated iron, wood and clay. Families with multiple children share single-room dwellings with an average estimated size of around ten square meters (Andvig & Barasa 2014). Public toilets are shared by hundreds of people (Corburn & Karanja 2014). As a result of these conditions, "slum dwellers [...] are constantly exposed to eviction, disease and violence." (UN-Habitat 2015: 1).

3.2.3 The Mathare Slums of Nairobi

While Nairobi is celebrated as Africa's "Silicon Savannah" (an allusion to San Francisco's Silicon Valley) and is a hotspot for technology start-ups worldwide (e.g. Davies 2014), it is also a city of slums. Despite the fact that the slums cover only around 5 % of Nairobi's residential area, Kenya's capital is home to over two million slum dwellers representing around 60 % of Nairobi's total population (Amnesty International 2009; UN-Habitat 2016).

There are approximately 135 named slums in Nairobi (Wanjiru & Matsubara 2017). One of these slums is Mathare.¹ The slum covers an area of approximately 0.88 square kilometres and is located

¹ The word 'Mathare' means the *Dracaena* plant in Kikuyu, the language of Kenya's largest Bantu ethnic group (or tribe).

around three kilometres to the north-east of Nairobi's Central Business District. Mathare extends from west to east along two rivers, the Mathare River and the Gitathuru River, and is bounded by two main highways, Thika Road in the north and Juja Road in the south. A major area of the slum is in a valley – the Mathare Valley – which originated from stone quarrying during the British colonial period. The Mathare slum is currently divided into 13 named subsettlements, usually called 'villages' (Corburn et al. 2012; Wanjiru & Matsubara 2017). These are: Mashimoni, Mabatini, No. 10 (Kwa Nyangau), Village 2 (Kiandururu), Kosovo (New Millenium), 3A (Bondeni), 3B (Kwa Josphat), 3C, 4A (Mandera), 4B (Kwa Gitunguru), Gitathuru, Kiamutisya and Kwa Kariuki.² Each village is further divided into sub-villages. *Figure 5* shows a cartographic map of the Mathare slums and its villages.



Figure 5: Cartographic map of the Mathare slums (Wanjiru & Matsubara 2017: 36).

² Common alternative names of Mathare's villages are in brackets.

The history of the slum dates from the 1920s, when the first residents settled in Mathare (Karanja & Makau 2009; Wanjiru & Matsubara 2017). Rapid growth of the settlement took place with the independence of Kenya in 1963 (Amnesty International 2009). Today, the total population of Mathare is unknown (ibid.). Kenya's official population and housing census from 2009 counted 80,309 inhabitants (Corburn et al. 2012; see *Table 1*), but UN-Habitat (2017) and non-governmental organisations (NGOs) regularly estimate that more than half a million people live in Mathare. Scientific field research, however, renders both figures unrealistic and suggests a total population of approximately 200,000 people (Corburn et al. 2012; Andvig & Barasa 2014; Mkoji 2014; Wanjiru & Matsubara 2017).

3.2.4 Non-Formal Schooling in Mathare

It is estimated that more than 40 % of the total population of Mathare are infants and children of primary school age (6–13 years) (Cheng & Kariithi 2008; Karanja & Makau 2009).³ However, only three of Kenya's 18,000 public primary schools are located in Mathare (Cheng & Kariithi 2008; Amnesty International 2009). These three governmental schools provide free primary education for around 3,100 children from Mathare and have an average pupilteacher ratio of more than 60:1 (Cheng & Kariithi 2008). Classroom overcrowding and insufficient numbers of teaching staff affect the quality of the educational provision and are the result of chronic underfunding of public schools in slum areas (Tooley & Dixon 2005; Dixon 2012; IFC 2014).

To meet the demand for education, slum dwellers have established and operate their own primary schools. These schools are referred to by different names. In academic discourse, terms such as

³ The UN demographic data on the age structure of Kenya from the year 2015 shows that 41.4 % of the national population is under the age of 15 years (United Nations 2019b). The population in slum areas can be expected to be even younger.



The Mathare River (Photo: João Victor Novelletto Bolan)

A path in Mathare (Photo: João Victor Novelletto Bolan)



Village	No. of	No. of	Area in
	Residents	Households	Sq. km
3A	4,059	1,530	0.0536
3B	7,433	2,681	0.0497
3C	5,316	1,925	0.0761
4A	18,776	5,627	0.2151
4B	5,681	1,810	0.0610
Gitathuru	3,737	1,241	0.0464
Kiamutisya	5,825	2,351	0.0540
Kosovo	8,085	2,846	0.0835
Kwa Kariuki	5,290	1,878	0.0545
Mabatini	1,160	383	0.0380
Mashimoni	4,478	1,692	0.0526
No. 10	2,594	994	0.0272
Village 2	7,875	2,854	0.0720
Totals	80,309	27,812	0.8837

Table 1: Kenya Population and Housing Census 2009 (Corburn et al. 2012: 16, adapted).

'low-cost private schools' or 'low-fee private schools' are frequently used to describe these schools (e.g. Dixon 2012). Schools that are unregistered and, therefore, operate "below the radar" of the state authorities are often referred to as 'unregistered schools' or 'unrecognised schools' (Dixon 2012: 188). As they are small in scale, the schools are sometimes also called 'micro schools' (Knüppel & Groß 2011). In Kenya, the national government usually referred to them as 'complementary schools' or 'non-formal schools' (MoEST 2015). Since the introduction of the APBET policy (Alternative Provision of Basic Education and Training) by Kenva's Ministry of Education, Science and Technology (MoEST) in 2009, the term 'APBET' has replaced the term 'non-formal education' and the term 'APBET school' has replaced the term 'non-formal school' (ibid.). While the term 'non-formal education' describes systematic and intentional education activities which do not comply with, or are not tied to, the current educational regulations (Coombs & Ahmed 1974)⁴, the APBET policy aims to set separate regulations for schools in the slums. Nonetheless, the former term, 'non-formal school', is still widely used. Although the non-formal schools in Mathare do not comply with all the educational regulations - such as the requirements for schools to have their own toilets and spacious playgrounds - they usually follow the national basic educational curriculum of the KICD (Kenya Institute for Curriculum Development) and administer KCPE (Kenva Certificate of Primary Education) exams set by the KNEC (Kenya National Examination Council) at the end of Class Eight.⁵ As well as their primary education services, most non-formal schools also provide early childhood education for pre-primary school aged children (Cheng & Kariithi 2008; Dignitas 2012).

The total number of non-formal schools in Mathare is unknown. Probably one of Mathare's only comprehensive school censuses was conducted by the Dignitas project – by Harvard education scientist Tiffany Cheng Nyaggah – in 2012. Based on their census data, Cheng et al. list 85 non-formal schools (Dignitas 2012). These schools are located in the following villages of Mathare: Village 2 (5

⁴ Non-formal education can be distinguished from formal and informal education (Coombs & Ahmed 1974). In contrast to the definition of non-formal education above, formal education describes systematic and intentional education activities which comply with the current educational regulations (ibid.). Informal education, in turn, describes unsystematic and non-intentional education activities which occur spontaneously from everyday experiences (ibid.).

⁵ However, fundamental changes can be expected by 2027 at the latest, when Kenya's 8-4-4 curriculum framework is due to be replaced by a new 2-6-6-3 curriculum framework (KICD 2018).



Teacher at the blackboard in Mathare (Photo: João Victor Novelletto Bolan)

Group of students learning in Mathare (Photo: João Victor Novelletto Bolan)



schools), 3B (7), 3A (4), 3C (4), No. 10 (9), 4A (27), 4B (14), Kosovo (6) and Mabatini (9). The schools enrolled 189 students each on average, making a total of around 16,000 students (ibid.). The pupil roll ranged from 15 to more than 800 students (ibid.). Seven teachers per school were employed on average, while the numbers of teaching staff ranged from one (usually the school founder) to twenty-five teachers (ibid.). The mean average age of the schools was around seven years, with the oldest school founded as far back as 1981 (ibid.). While the Dignitas census may still provide the best available data on Mathare's non-formal schools, it must be considered as incomplete. This is because the census data was not collected from the entire slum but only from a sample of its villages. Furthermore, regular school censuses would be necessary to record given changes over time.

3.3 Theoretical Background

3.3.1 Preliminary Remarks

In order to examine the non-formal schools in Mathare based on a well-founded theoretical approach, the following section outlines Jason Fairbourne et al.'s concept of microfranchising (including its specific manifestation as bottom-up franchising) and the entire 'iceberg' of its underlying theoretical foundation. This approach demonstrates how the concept of bottom-up franchising can be derived from C. K. Prahalad's BoP concept, and how the latter concept, in turn, can be derived from Joseph Schumpeter's Theory of Economic Development.

3.3.2 Joseph Schumpeter's Theory of Economic Development

In the history of modern economics, different schools of economic thought have created different theories of economic development. Since the late 19th century, one of the most influential schools of economic thought has been and still is *neoclassical economics*. In neoclassical economics, *economic growth* is assumed to be the

conceptual equivalent of economic development; theories of economic growth are considered, by the same token, to be theories of economic development (Brinkman 1995). Thereby, neoclassical theories define economic growth as an increase in the amount of production outputs. This production increase is perceived as desirable because it leads - at least in the long run - to the creation of greater wealth, which is defined as the amount of accumulated goods. Economic growth itself is explained by increases in: (1) the amount; and/or (2) the efficiency of the input factors of production, i.e. labour and capital. Growth caused by an increase in the amount of input factors used is, thereby, termed extensive growth. Growth caused by the more efficient use of input factors is called intensive growth. Since neoclassical economics argues that the allocation pattern of the market leads to production efficiency, free trade is considered as the most effective driver for intensive growth (Meier 1994). However, neoclassical theories fail to explain aspects of economic growth that cannot be traced back to an increase in the production factors or improved implementation of the market pattern. The unexplained residual is considered to result from the exogenous 'black box' of technological progress (Scott 1989). Technological progress itself cannot be explained because of the static, macroeconomic equilibrium ideal on which neoclassical theories are based.

The Austrian economist Joseph Alois Schumpeter (1883–1950) countered this equilibrium ideal by his idea of dynamic, *innovation-based* growth, which became known as Schumpeterian growth. The theoretical starting point for Schumpeter's economic thought was his early 1911 work, *The Theory of Economic Development*. In this work, Schumpeter [1911] 1949: 215) set out to describe "the form economic development takes in the era of capitalism". At the core of this work lies Schumpeter's "concept of combinations" (ibid.: 15), which he introduced as follows:

... to produce means to combine the things and forces within our reach. Every method of production signifies some such definite combination. Different methods of production can only be distinguished by the manner of the combination, that is either by the objects combined or by the relation between their quantities. Every concrete act of production embodies for us, is for us, such a combination. This concept may be extended even to transportation and so forth, in short that is production in the widest sense. An enterprise as such and even the productive conditions of the whole economic system we shall regard as 'combinations.' This concept plays a considerable part in our analysis (Schumpeter [1911] 1949: 14).

For Schumpeter, therefore, production processes of any kind, their conditions and their enterprises are *combinations of objects*. On that basis, Schumpeter introduces the term "innovation" to describe "the carrying out of *new* combinations", whereby new combinations differ qualitatively from the old ones (ibid.: passim, own italics). Furthermore, Schumpeter also emphasises the distinction between an *innovation* and an "economically irrelevant" *invention*, i.e. a new combination that is "not [yet] carried into practice" (ibid.: 88). In this sense, an innovation can be said to be the economic application – the commercialisation – of an invention. Concerning the question *how* a new combination can be carried out in practice, Schumpeter distinguishes five different ways:

- The introduction of a new good that is one with which consumers are not yet familiar – or a new quality of a good.
- (2) The introduction of a new method of production, that is one not yet tested by experience in the branch of manufacture concerned, which need by no means be founded upon a discovery scientifically new, and can also exist in a new way of handling a commodity commercially.
- (3) The opening of a new market, that is a market into which the particular branch of manufacture

of the country in question has not previously entered, whether or not this market has existed before.

- (4) The conquest of a new source of supply of raw materials or half-manufactured goods, again irrespective of whether this source already exists or whether it has first to be created.
- (5) The carrying out of the new organization of any industry, like the creation of a monopoly position (for example through trustification) or the breaking up of a monopoly position.

(Schumpeter [1911] 1949: 66).

In contemporary innovation research, the five Schumpeterian innovation types are frequently referred to as: (1) product innovation; (2) process innovation; (3) market innovation; (4) supply chain innovation; and (5) organisational innovation (see e.g. Lazzarotti et al. 2011; Baunsgaard & Clegg 2015).

Crucial for Schumpeter's Theory of Economic Development is also the question of who actually carries out new combinations. To name the "promoter" of new combinations, Schumpeter introduced the role of the entrepreneur (Schumpeter [1911] 1949: 78, 137-138). His or her function is entrepreneurship, i.e. a 'combinatory' activity involving innovation. The fulfilment of this entrepreneurial function is what distinguishes the entrepreneur from an administrative trustee (manager) or owner of a business per se (ibid.: 45-46). Moreover, Schumpeter would later argue that only a small group of elites - a minority of extraordinarily talented people – can become entrepreneurs (Heilbroner [1953] 1999). Against this backdrop, Schumpeter's functional definition of the entrepreneur should also be distinguished from other behavioural definitions of the entrepreneur which, in their simplest form, "call individuals who start their own businesses entrepreneurs" (Bhidé 2003: 25, italics in original removed; see also Stevenson 1983).

For Schumpeter, the prime motivator for the entrepreneur is profit; the entrepreneur innovates to earn a monetary return. The entrepreneur's monetary surplus over and above costs is only temporary and results from a monopoly-like market position that he or she occupies due to his or her innovation. The surplus decreases over time as competitors imitate the new combination. Consequently, Schumpeter asserts that profit can only be generated in an imperfect competition, which – in contrast to neoclassical economics – he believes is the rule rather than the exception (Schumpeter [1942] 2003: 78).

Since Schumpeter (1923: 105) describes profit as "the premium put upon successful innovation", the question can be asked what characterises the success of an innovation from his point of view. The answer to this question lies in a process that Schumpeter would famously call, in his late 1942 work Capitalism, Socialism and Democracy, "creative destruction" (Schumpeter [1942] 2003: 81 ff.). The process of creative destruction describes how successful innovations render old combinations obsolete. In other words, an innovation is successful if it leads to technological progress by forcing all competitors to adopt the new combination and, in so doing, scales across the entire economic system. Hence, the success of an innovation can be measured by the scale it reaches. Schumpeter's concept of creative destruction continues to gain prominence in contemporary innovation research under the mantel of Christensen's (1997) concept of disruptive innovation (Priddat 2017).

As well as the concept of creative destruction, Schumpeter made a second substantial addition to his economic development theory in *Capitalism, Socialism and Democracy*: his hypothesis that large firm size is more advantageous to innovation (McCraw 2007: 640).⁶ The hypothesis became widely known as the "Schumpeter

⁶ The hypothesis is attributed to a number of statements made by Schumpeter in *Capitalism, Socialism and Democracy*. Schumpeter wrote, for example: "[I]t is not sufficient to argue that [...] the large-scale establishment or unit of control must

hypothesis" (ibid.). In this hypothesis, Schumpeter turns away from his focus on the entrepreneur as an individual person in favour of an ideal of "depersonalized" large-scale corporations (Schumpeter [1942] 2003: 133).

Based on these perspectives. Schumpeter ultimately describes the interplay between economic development and economic growth as follows. Firstly, he equates economic development with innovation, claiming: "Development in our sense is then defined by the carrying out of new combinations." (Schumpeter [1911] 1949: 66). Secondly, he makes it clear that he believes development is not the same as growth (ibid.: 63). For Schumpeter, economic growth can be defined as in neoclassical economics, but its theoretical explanation must also endogenise "entrepreneurship as a factor of economic growth" (Schumpeter 1947: 8). Thereby, Schumpeter argues that successful innovation promoted by entrepreneurs is the undetected driver of technological progress, which increases the growth-determining total factor productivity (Aghion & Howitt 1992). In the final analysis, growth does not necessarily require innovation, but innovation tends to induce growth. As a logical conclusion, it follows that growth without development is possible, but development is impossible without growth. In other words, only innovation-based growth involves economic development; production processes should, therefore, focus on innovation-based growth (Schumpeterian growth).

Since the 1980s, the heritage of Schumpeter's Theory of Economic Development has gained growing prominence in modern economics (Saßmannshausen 2012). In "the Age of Schumpeter" (Giersch 1984), Schumpeterian growth models became dominant in

be accepted as a necessary evil inseparable from the economic progress which it is prevented from sabotaging by the forces inherent in its productive apparatus. What we have got to accept is that it has come to be the most powerful engine of that progress and in particular of the long-run expansion of total output not only in spite of, but to a considerable extent through, this strategy which looks so restrictive when viewed in the individual case and from the individual point of time." (Schumpeter [1942] 2003: 106).

macroeconomics, entrepreneurship research emerged as a distinct field of scientific study, and modern innovation economics – which was largely built on the legacy of Schumpeter's thinking – set out to become one of the 21st century's most notable schools of economic thought (Terzic 2018). The following section examines more closely a recent offshoot of Schumpeter's economic thought, which devotes itself to poverty alleviation by means of entrepreneurship, innovation and Schumpeterian growth: C. K. Prahalad's *concept of the fortune at the bottom of the pyramid*.

3.3.3 C. K. Prahalad's Fortune at the Bottom of the Pyramid

In the history of modern economics, theories of economic development have regularly been connected with development policy agendas attempting to put theory into practice. Poverty alleviation has often been a major concern of such development policies. In the post-World War II era, international development agendas have been largely influenced by the so-called 'Keynesian consensus' (O'Connor 2001), which was based on the economic thought of John Maynard Keynes (1883-1946). Poverty alleviation efforts in that era focused on full employment, wage and price controls, as well as income redistribution by means of a generous welfare state (Toye 2006). The vilification of the Keynesian consensus in the 1970s as a "dirigiste dogma" that paternalises and patronises the poor, gave rise to the so-called 'Washington consensus' (Lal [1983] 2002: 39). The Washington consensus - a political reform development agenda established by the World Bank, the International Monetary Fund (IMF) and the U.S. Department of the Treasury (USDT) in the 1980s - was broadly based on neoclassical theories. In consequence, it promoted poverty alleviation by means of economic growth or, more precisely, by means of growth-enhancing global free trade (Williamson 1993). Thereby, economic growth was - and still is - assumed to alleviate poverty because an increase in production outputs inevitably leads to an increase in households' average purchasing power. However, the empirical data on global poverty at the end of the 1990s provoked increasing scepticism

about whether the Washington consensus in its original form could produce the desired results in terms of poverty alleviation. Finally, calls to adjust the consensus suggested taking Schumpeter's Theory of Economic Development into consideration (see Rodrik 2006; Williamson 2008). In that context, in 1998 U.S. economist C. K. Prahalad (1941–2010) proposed his influential concept *The Fortune at the Bottom of the Pyramid*, known as the *BoP* concept (Prahalad & Lieberthal 1998; Prahalad & Hart 1999) and stressed "the idea of large-scale entrepreneurship as a possible solution to poverty" (Prahalad [2004] 2005: xi; see also Hart et al. 2016: 403).

Prahalad draws on the demographic description of the world's population as a pyramid with a very small high-income population at the top and a much larger poor population at the bottom. As illustrated in *Figure 6*, when Prahalad introduced his concept, there were around 75 to 100 million people with an annual income above PPP US\$ 20,000 at the top of the pyramid (Tier 1), around 1,500 to 1,750 million people with an annual income of between PPP US\$ 1,500 and 20,000 in the middle of the pyramid (Tier 2–3), and around four billion poor people with an annual income of less than PPP US\$ 1,500 at the base/bottom of the pyramid (Tier 4).⁷ Prahalad points out that the aggregated purchasing power at the base of the pyramid (Tier 4) is likely to be as high as, or even higher than, the aggregated purchasing power in the middle of the pyramid (Tier 2–3).

Consequently, "Tier 4 [the base of the pyramid] represents a multitrillion-dollar market" that is largely untapped and continuously growing (Prahalad & Hart 2002: 2). Given the massive market potential at the base of the pyramid, Prahalad suggests that businesses can profit not only from serving a small number of rich people at high margins, but also from serving a much greater

⁷ Please note that the income threshold used by Prahalad to identify the base of the pyramid differs from the international poverty line used by the United Nations and the World Bank (see *section 1.1*).

⁸ In their figure, Prahalad & Hart refer to a purchasing power parity (PPP) per capita income in US\$ *per annum*.



Figure 6: The Global Income Pyramid (Prahalad & Hart 1999: 4).8

number of poor people at lower margins (Prahalad & Lieberthal 1998; Prahalad & Hart 1999, 2002; Prahalad [2004] 2005). In other words, Prahalad maintains that base of the pyramid markets are profitable markets if penetrated on a large scale.

However, Prahalad's concept goes beyond demonstrating the possibility for profit generation at the base of the pyramid; it aims for a win-win scenario in which profitability and poverty alleviation go hand in hand (Prahalad [2004] 2005). For this purpose, Prahalad created "the idea of alleviating poverty through capitalismfor-the-poor" (Woodworth 2007: 88). The base of the pyramid is here identified as "the population of the world that is generally excluded from the current system of global capitalism" (London & Hart 2011: 8, italics in original removed). Businesses targeting the base of the pyramid would include the poor populations in the global supply and value chains of capitalist markets (e.g. UNDP 2008; London & Hart 2011; Casado Cañeque & Hart 2015). Thereby, they would alleviate poverty by serving the 'underserved' poor with affordable economic goods to satisfy basic human needs (ibid.). In doing so, they would promote "inclusive capitalism" (Prahald & Hart 2002: 2) and "inclusive globalization" (Prahalad [2004] 2005: 5). For that reason, businesses targeting the base of the pyramid became commonly referred to as *inclusive businesses*,

i.e. businesses that alleviate poverty by including the base of the pyramid in the markets of global capitalism (e.g. UNDP 2008; London & Hart 2011; Casado Cañeque & Hart 2015). Concerning the question of how inclusive businesses should operate in order to achieve their objective, the BoP concept emphasises that "[i]nnovation across the board is an imperative to serve the bottom of the pyramid" (Prahalad & Hart 1999: 8). Such innovation would be obliged to consider issues of affordability in order to reach the low-income customers at the base of the pyramid (Prahalad [2004] 2005). In the end, "[t]he [market] potential at the bottom of the pyramid cannot be realized without [such] an entrepreneurial orientation" (Prahalad & Hart 1999: 8). Taking into account the fact that Schumpeter identified entrepreneurship as a factor of economic growth, the BoP concept finally merged with the concept of in*clusive growth* – offering the goal of marrying economic growth with poverty alleviation (UNDP 2008; George et al. 2012; Hall et al. 2012; Hart et al. 2016).

In terms of possible challenges that could arise in the practical application of the BoP concept, Prahalad ([2004] 2005) stresses the need for *strong institutions* to support the efficiency of market interactions at the base of the pyramid. In general terms, "[i]nstitutions are the humanly devised constraints that structure political, economic and social interaction" (North 1991: 97).

Institutions have an essential role in a market economy to support the effective functioning of the market mechanism, such that firms and individuals can engage in market transactions without incurring undue costs or risks (North, 1990; Peng, 2008). These institutions include, for example, the legal framework and its enforcement, property rights, information systems, and regulatory regimes. We consider institutional arrangements to be 'strong' if they support ... an effective market mechanism. Conversely, we refer to institutions as 'weak' if they fail to ensure effective markets or even undermine markets (Meyer et al. 2009: 63).

Considering this differentiation between 'strong' and 'weak' institutions, the "institutions within BOP [base of the pyramid] markets are generally considered [...] as 'weak'" (Kistruck et al. 2015: 438; see also Kistruck et al. 2011, 2013). Consequently, Prahalad ([2004] 2005) emphasised the need to build institutional capacity in base of the pyramid markets. This should include the capacity to create market transparency, to protect property rights and to enforce commercial contracts. Such institutional capacity would eliminate uncertainties and risks in market transactions (ibid.). In doing so, market transaction costs would be reduced, leading to an increase in the efficiency of market interactions (ibid.). To build such institutional capacity, an institutional reconfiguration of base of the pyramid markets would be necessary. This reconfiguration could not be induced from the top of the pyramid - it would have to rise from the base of the pyramid itself (see Prahalad & Hart 1999). Consequently, poverty research has increasingly stressed the necessity of bottom-up approaches, which - in contrast to top-down approaches - emphasise the active participation of the inclusive businesses' stakeholders at the base of the pyramid and assign them higher degrees of responsibility in terms of the reconfiguration of the local institutional order. Ultimately, the poor are no longer merely addressed by inclusive businesses as recipients but are also considered as co-creators (Simanis & Hart 2008; London & Hart 2011; Kolk et al. 2014; Casado Cañegue & Hart 2015).

3.3.4 Jason Fairbourne et al.'s Microfranchising

One of the most notable manifestations at the base of the pyramid is the myriad of *microenterprises* operating in the empty spaces left by the absence of markets of global capitalism (Max-Neef et al. [1986] 1989: 62). In response to the question why so many entrepreneurs at the base of the pyramid keep their enterprises on a 'micro' scale, i.e. at a very small operational scale, the argument is regularly made that they lack the opportunities to reinvest profits to expand their businesses (Burand & Koch 2010: 24). Furthermore, it is assumed that "microentrepreneur[s] [at the base of the
pyramid] may fear that a larger enterprise [...] will demand more entrepreneurial expertise and skills than the microentrepreneur currently commands" (ibid.).

Having identified the microentrepreneurs at the base of the pyramid as an attractive customer segment to be served by inclusive businesses, a number of business-to-business (B2B) innovations have been introduced to base of the pyramid markets on a large scale. Probably one of the best-known is still the *microcredit*. Nobel laureate Muhammad Yunus and his Grameen Bank developed this approach of providing small loans to people at the base of the pyramid in the early 1970s. Since then, billions of microcredits have enabled nascent and operating microentrepreneurs at the base of the pyramid to start or expand their own microbusinesses. Unfortunately, most of these small-scale businesses did not manage to scale up, or even failed, in the early years of their existence. This caused poverty researchers to pose the question: "What good is a loan if a person can't use it effectively?" (Woodworth 2007: 93-4). Consequently, poverty research has called more recently for innovations to enable microentrepreneurs at the base of the pyramid to start microbusinesses and run them effectively (ibid.). This is where *microfranchising* comes in.

The concept of microfranchising was established in poverty research by the U.S. economists Jason Fairbourne, Stephen W. Gibson and W. Gibb Dyer, who have discussed it in a number of publications since 2005 (see e.g. Gibson & Fairbourne 2005; Fairbourne 2006; Fairbourne et al. 2007). In his 2006 paper, *Microfranchising*, Fairbourne explains how microfranchising draws on the traditional concept of franchising and how the latter needs to be adapted for base of the pyramid markets. He defines microfranchising as follows:

Let's define microfranchising by dissecting the term into two sections: micro and franchising. The *micro* in microfranchising refers to the social aspect of assisting the poor at the base of the economic pyramid. The franchising in microfranchising refers to the systematization and replication of enterprises. ... Simply put, microfranchising is the systematization and replication of microenterprises with the intent to alleviate poverty (Fairbourne 2006: 19).

Fairbourne suggests that microfranchising is the systematisation and replication of microbusinesses at the base of the pyramid with the aim of poverty alleviation. To discuss this systematisation and replication in greater detail, a closer examination of the traditional concept of franchising is necessary. Here, systematisation means "paying close attention to each and every aspect of a business until it is a turn-key operation" (Fairbourne 2007: 9). In other words, systematisation involves the creation of a standardised operational system that serves as a turn-key business model - a 'business-in-abox' solution (Kistruck et al. 2011). Replication means a systematic way of upscaling in which the privilege to use this turn-key business model under a common brand is granted to other businesses. This privilege usually covers the licensing of intellectual property rights, such as trademarks. Businesses franchised in this manner, i.e. businesses operating autonomously but under the branded and standardised business model, are called the *franchisees*. The enterprise that conducts the systematisation and replication and subsequently monitors the franchisees' compliance with set operational standards is called the *franchisor*.

In the context of microfranchising, the franchisees are microenterprises at the base of the pyramid, while the franchisor can be viewed as an inclusive business in the sense of Prahalad's BoP concept (Sunanda 2016). In accordance with the BoP concept, microfranchising should benefit both parties. First and foremost, the microfranchisees are empowered to run their businesses more successfully due to the microfranchises' key performance drivers of *standardisation* and *branding* (Kistruck et al. 2011). The standardisation of business operations supports the microentrepreneurs with economies of scale within the franchise network, and also reduces their individual entrepreneurial risks (Fairbourne et al. 2007). The common branding allows customers at the base of the pyramid to distinguish the microfranchisees from other market participants, decreases customers' uncertainty about what to expect and, finally, generates trustful customer ties (see Kistruck et al. 2011). Overall, "it is the intention, [...] to move these [microfranchised] enterprises toward the mainstream economy [of global capitalism]" (Henriques & Herr 2007: 63). The benefit for the inclusive business is generally a franchise fee from the microfranchisees, which generates profit. As the franchise fee per microfranchisee will be low, for reasons of affordability, a large-scale microfranchise network is deemed necessary in order to make a profitable case (see also *section 3.3.3*).

Although "microfranchising has been championed as a model for scaling poverty alleviation efforts in BOP [base of the pyramid] markets", empirical field research has, however, demonstrated that the anticipated benefits of microfranchising are regularly not realised (Kistruck et al. 2011: 525). This is supposedly because microfranchises' performance drivers are significantly affected by the weak institutions of base of the pyramid markets (ibid.: passim). More precisely, the lack of market transparency, property rights protection and contract enforceability at the base of the pyramid undermines the ability of microfranchises to standardise operations, capitalise on a common brand, and establish monitoring mechanisms (ibid.). To deal with such institutional challenges, it was proposed that the concept of microfranchising should undergo adaptations in line with Prahalad's emphasis on the need of bottomup approaches to institutional capacity building (see Kistruck et al. 2011). These considerations gave rise to the concept of *bottom-up* franchising (Henriques & Herr 2007; Munoz et al. 2010; see also Kistruck et al. 2011; Lawson-Lartego 2016). While microfranchising was originally conceptualised by Fairbourne et al. in line with the top-down approach of traditional franchising, in which the franchisor is supposed to create and provide a turn-key business model, bottom-up franchising lets the microfranchisees themselves co-create their common operational system based on collective decision making (see ibid.; see also Hendrikse & Windsperger 2012). In the latter process, the microfranchisees establish the operational

standards and capitalise on the common brand of the microfranchise system more self-reliantly, and monitor each other in a mutual manner ('peer monitoring'). In short, the microfranchisees become co-franchisors. This may, however, give rise to further constraints and unintended consequences of bottom-up franchising. As analysed by Henriques & Herr (2007), greater autonomy of the microfranchisees can lead to a slower evolution of a branded and standardised operational system compared to the traditional top-down franchising model and may require one or more lead entrepreneurs among the microfranchisees who can convince others to join forces.

3.4 Method

3.4.1 The Appropriateness of the Methodology of Barefoot Economics

The *appropriateness of methods* is regularly regarded as a major quality criterion of empirical social research (Flick 2009). The research method of this scientific study has been derived from the principles of barefoot economics, as described in *chapter 2*. Considering the research question of why bottom-up franchises are rarely observed at the base of the pyramid, a barefoot economic approach seems particularly appropriate in order to answer the posed 'why question' not by a positivist testing of hypotheses to identify possible causal links between economic phenomena, but rather by a phenomenological understanding of the phenomena which appear to the slum dwellers in their lived experience. This approach identifies the reasons why the slum dwellers behave in the ways they do.

The phenomenological method of epoché generally intends to 'go to the things themselves', by identifying and eliminating preconceptions of the scientist. Consequently, barefoot economics enables the cognitive biases of the researcher induced by the usage of a certain scientific language to be overcome. The elimination of bias is particularly significant in "studies involving strongly normative and sensitive issues such as poverty alleviation" (Kistruck et al. 2011: 507). From a barefoot economic perspective, ex-ante hypotheses as proposed by the methodology of positive economics inevitably fail to eliminate researcher bias since they are necessarily well-formulated in a preconceived language commonly accepted in the researcher's scientific community. For the same reason, barefoot economic research does not aim to "categorize and codify others' experience in terms of [one's] own already existing frameworks and concepts" (Imas et al. 2012: 570). Instead, barefoot economics proposes to prune one's own language and gain own lived experiences.

3.4.2 The Method of Linguistic Pruning

With his barefoot economic 'orchard' method, Max-Neef proposes the methodological pruning of language focusing on those key words which may distort the perception of relevant phenomena by creating bias in terms of preconceptions. The method has already been described in *section 2.3.2*.

With respect to the practical application of the orchard method in scientific research, it should be noted that the pruned words must relate to the jargon used within the scientific discourse prevailing in the linguistic community of the researcher (Max-Neef [1982] 1992, [1988] 1991a, [1989] 1991b, 2009; Smith & Max-Neef 2011). In this context, the concept of discourse (Foucault [1969] 1972; Lyotard [1979] 1984; Habermas 1981) becomes relevant. Contemporary conceptualisations of the notion of 'discourse' originate from 20th century's linguistic turn in philosophy (Rhees 1998; compare section 2.1 and section 2.3). Against this background, a variety of definitions has emerged (Potter et al. 1990). In this study, discourse is defined in general terms as a particular way of thinking or worldview expressed by common language (see also Potter et al. 1990; Laclau & Mouffe 2001). In his barefoot economic research during the 1970s, Max-Neef identified the dominant discourse of his academic community as the "development discourse" (Max-Neef et

al. [1986] 1989: 45; see also Max-Neef [1982] 1992, 2009), which aimed at a modernisation of underdeveloped countries following the example of developed industrial nations within an institutional framework of good governance regimes (Moore & Schmitz 1995). Consequently, "[Max-Neef] chose to prune from [his] language the following words: *development, economic growth, efficiency* and *productivity*. In addition to these words, such conventional economic indicators as *Gross National Product* and its offspring were also pruned" (Max-Neef [1988] 1991a: 100, own italics).

In general, the vocabulary of scientific discourses allows the researcher to perceive relevant phenomena according to commonly accepted ideas. This, however, conflicts with the aim of gaining phenomenological understanding as described in *section 2.3.1* and *section 2.3.2*. Barefoot economics proposes pruning the commonly used scientific jargon in the process of data collection, preparation and analysis, and returning to the pre-pruned language afterwards. It should be noted that the pruning of key terminology within the practice of barefoot economic research not only involves the effort of not using related terms as communication tools, but also implies the attempt to free one's own thought from scientific preconceptions by not thinking *in* these terms. Whenever the pruned words are referred to in the course of the research process, this is done to think *about* these terms in the sense of a meta-cognitive reflection (self-examination, introspection).

3.4.3 The Method of Real-World Experimentation

Barefoot economics by its nature involves lived experience (see *chapter 2*). This lived experience of the phenomena that are intended to be understood can take place within project-based, empirical fieldwork (see Max-Neef [1982] 1992). Methodologically, this kind of field research can be described by the scientific method of real-world experimentation (e.g. Groß et al. 2005).

Real-world experiments as a promising method within barefoot economics are diametrically opposed to *lab-in-the-field experiments* and, especially, to *randomised controlled trials* (*RCTs*) which constitute the methodological gold standard of positive economics (Banerjee et al. 2019; see also Labrousse 2016a). While real-world experiments and lab-in-the-field experiments can both be viewed as methods of *field experimentation*, there are significant differences between them.

Commonly, experimentation can be described as a research method involving an empirical intervention stimulus, usually induced by the researcher. If the intervention takes place within an artificially engineered research setting, the experiment is referred to as a *laboratory experiment*. If the intervention takes place within a research setting that is basically a pre-existing natural environment, the experiment is referred to as a *field experiment*. Since the research setting of an experiment constitutes the boundary conditions of its intervention (Groß et al. 2005), field experiments are generally distinguished from laboratory experiments by their lesser control of the intervention-related boundary conditions. However, the degree of control of the boundary conditions still varies significantly between different types of field experiments. Thereby, the extent of the effort to control the boundary conditions in the field regularly depends on the objective of the field experiment (Kleining 1986; Kleining & Witt 2000, 2001). Generally, inspective field experiments, aiming to test hypotheses, require a higher degree of control than explorative field experiments, which aim to reveal unexpected discoveries (Kleining & Witt 2001).

RCTs can be characterised as inspective field experiments in line with positive economics' objective of hypothesis testing to achieve accurate predictions (Labrousse 2016a; see also Banerjee & Duflo 2011; Banerjee et al. 2019). Based on the positivist ideal of scientific value-freedom and objectivity, large quantitative data sets are collected which allow for the falsification of hypotheses by means of mathematical-statistical analyses (ibid.). Furthermore, the boundary conditions of interventions attempt to control as many variables as possible in order to eliminate confounding factors that may interfere with the hypothesis testing. Thereby, the boundary conditions are typically controlled by randomising the experiment participants into treatment and control groups (ibid.) and, in some cases, by additional pre-post-measurements (see Diekmann 2014). RCTs are referred to as lab-in-the-field experiments because their intention is to create laboratory-like conditions in the field.

Real-world experiments, on the other hand, work with situationspecific boundary conditions rather than controlled ones, and have explorative objectives rather than inspective ones (see Groß et al. 2005). In barefoot economics, the objective is an exploration of the essence or meaning of poverty-related phenomena. Furthermore, the interventions of real-world experiments are not predetermined treatments of experiment participants as in the case of RCTs. The intervention of a real-world experiment can, in contrast, be described as the performance of heuristic 'trial-and-error' actions regarding a certain task, conducted by a project team of scientists and practitioners over an extended period of vears (Groß et al. 2005; Wanner et al. 2018). During that period, researchers are obliged to 'live through' the relevant phenomena of their research. The research settings for real-world experiments are provided by what are called 'real-world lab projects' (Schneidewind et al. 2016a, 2018; Wanner et al. 2018; Rose et al. 2019). Within these projects, a project team of scientists and practitioners under the joint leadership of both parties is formed and institutionalised, funds are procured, and intervention activities are planned, conducted, documented and evaluated (Schneidewind et al. 2018; Wanner et al. 2018; Rose et al. 2019).

3.4.4 The mpito[®] Project

3.4.4.1 Setting and Design

The MPITO[®] project was designed as a real-world lab project in order to conduct a real-world experiment on bottom-up franchising in the non-formal education sector of the Mathare slums over



Figure 7: The MPITO[®] logo (designed by Alexandra Konrad; source: MPITO[®] group)

several years.⁹ The project was based on a three-month preparatory field research study carried out in Mathare in 2012–2013, which analysed the business models of the local non-formal schools (see *section 3.2.4*). The MPITO[®] project began its work by planning the intended bottom-up franchise experiment in early 2014. An interdisciplinary German-Kenyan project team was formed, comprising of around twenty academics and practitioners from the fields of economics, sociology, pedagogy, law and politics, finance and accounting, business administration, community development, social work, informatics, art and design, etc., who volunteered between 2014 and 2020. Initially, the project team worked without any financial budget. To procure funds, a non-profit legal entity, named the MPITO[®] group, was founded by the project team at the end of 2015. Subsequently, the project operated with a small annual budget, amounting to around US\$ 3,000 in 2016 and 2017

⁹ The word 'mpito' means 'transition' in Kiswahili, the Bantu language which serves as Kenya's national language.

and around US\$ 7,000 in 2018 and 2019.¹⁰ In 2016, the project appointed one salaried employee in Mathare and in mid-2018 this increased to three Kenyan employees.

In 2014, an initial project plan was written setting out the basic features of the intended bottom-up franchise experiment and 'mpito' was designed to become the common brand under which the microfranchised schools would operate. To ensure the participating schools shared a standardised visual appearance, a professional corporate design and related guidelines were created. There was also the intention to equip the schools with branded materials, such as writing pads, bags, shirts and pin-back button badges. Legally, the word 'mpito' and the MPITO[®] logo (see Figure 7) became registered trademarks of the MPITO[®] group to prevent the brand being used by third parties - especially other non-formal schools in Mathare that did not participate in the experiment. In addition to these 'top-down' branding activities, the intention was to create a platform where the participating schools could jointly exercise their branding and standardisation responsibilities at grassroots level. For this purpose, the MPITO® school network was formed. The aim was for the MPITO[®] school network to have its headquarters at one of the participating schools and host regular consultation meetings of all the head teachers. To convince the head teachers of Mathare's non-formal schools to participate in the experiment, there was no franchise fee and material incentives were provided to the schools. These included around 2,000 textbooks and 40 laptops, which were donated to the MPITO[®] group by charitable partner organisations.

The bottom-up franchise experiment of the MPITO[®] project was finally put into practice on the ground in Mathare in January 2015 without a fixed project term/end date.

¹⁰ The administrative and fundraising costs (overheads) of the MPITO[®] group were regularly around 10 % of the total budget. The legal structure of the MPITO[®] group was changed in 2018; the overheads for that particular year were consequently around 20 %.

3.4.4.2 Sampling and Panel

The sampling of the schools participating in the real-world experiment was conducted by the method of *targeted sampling* (Watters & Biernacki 1989). Targeted sampling is a method developed to reach hidden populations (ibid.). A population is described as hidden if a sampling frame is lacking and the population is hard to reach (e.g. Salganik & Heckathorn 2004; Magnani et al. 2005). In such cases, targeted sampling can be used to construct a target population as a sample frame and apply modified chain-referral sampling (Watters & Biernacki 1989).

In the case of the non-formal schools in Mathare, the population can be considered to be hidden insofar as its total size is unknown and the schools are broadly inaccessible to those outside the slum community. Using the method of targeted sampling, a sampling frame was constructed by drawing upon the 2012 Dignitas school census, as described in section 3.2. To create a representative sample for that target population, the dispersion parameters and mean values of the sample data had to be aligned with the census data. Based on this objective, the recruitment of schools took place via chain-referral sampling in the first stage. Those schools that were willing to participate in the experiment recruited further schools. In the second stage, the selection of schools was refined in order to meet the target conditions. To match the dispersion parameters of the Dignitas school census, a maximum variation sampling (Patton 1990) in terms of the schools' size, age and location was applied. The definite sample was to serve as a panel from which to collect longitudinal data over several years. The initial 2015 'MPITO® panel' of seven schools is depicted in Table 2.

The schools in the 2015 MPITO[®] panel had a combined student population of 1,543 and employed 59 teachers. On average, the MPITO[®] schools had 8 teachers and 220 students, had been established for five years and became registered a year and a quarter after their foundation. The data from the MPITO[®] panel broadly reflected the mean averages from the 2012 Dignitas school census; however, the MPITO[®] schools were slightly bigger in size and

	Success Care Centre	Star Educational Centre	Mumo Education and Orphanage Centre	Mathare Wisdom Care Centre	Destiny Community Education Centre	Daystar Junior Educational Centre	Bright Education Centre	School Name
	Otieno Kennedy Odero	Esther Ndunge Katundu	Christopher Malusi Ngombalu	Godfrey Mutanda Owino	Dixon Odhiambo Owaga	Patrick Ojiambo Juma	Caroline Muendi Kamuya	Founder/ Director
	4A	Village 2	Mabatini	Gitathuru	No. 10	Kwa Kariuki	3C	Location (Village)
	2008	2004	2009	2012	2013	2014	2009	Year of Foundation
	2010	2007	2009	2014	2013	2014	2011	Year of Registration
1,543	242	190	60	241	512	240	58	No. of Students
59	10	8	4	9	17	∞	υ	No. of Teachers

Table 2: 2015 MPITO® panel.

82

slightly newer (compare section 3.2). Regarding its dispersion parameters, the MPITO[®] panel represented nearly the entire value range in terms of pupil enrolment (58–512) and teaching staff (3–17) (compare section 3.2). Moreover, the MPITO[®] schools were spread geographically across the slum and every school was located in a different village (compare section 3.2). Figure 8 provides a map sketched by the head teachers showing the locations of the MPITO[®] schools in Mathare.



Figure 8: Sketch map of the Mathare slums and the MPITO[®] schools in 2015 (source: MPITO[®] group).

3.4.4.3 Data Collection

The tangible results of the MPITO[®] project's real-world experiment interventions were identified by the data collection method of *participant observation*. "Participant observation will be defined as a field strategy that simultaneously combines document analysis, interviewing of respondents and informants, direct participation and

observation, and introspection." (Denzin 1989: 157–158, as cited in Flick 2009: 226).¹¹ Data gathered through participant observation involves a validation of results by means of *data triangulation* (Boeri 2007; Flick 2009).¹²

The MPITO® project team has actively participated in and observed the experimental interventions in Mathare for more than five years (since January 2015). The observation data has been documented by means of research diaries, field notes, photographs and videography (see section 3.2 and section 3.5). The validity of the observation data has, in this way, been ensured by an *investigator* triangulation (Flick 2009) among the MPITO[®] team members. In January/February 2015, written quantitative surveys were conducted: these collected concrete numerical data about the characteristics of the participating schools, such as the number of students and employees, and financial figures (see Table 2 and section 3.5). In September 2015, a second survey round containing the same items was carried out to identify any significant data changes. In addition, all the schools participating in the real-world experiment provided a number of documents, including registration certificates and constitutional documents, from which school and personal data was drawn (see Table 2 and section 3.5). Meeting minutes were taken at the regular consultation meetings of the MPITO[®] schools' head teachers (see section 3.4.4.1); these were consolidated by the MPITO[®] project team into eight written interim reports dating from early 2018 (see MPITO[®] Interim Report 2018a, 2018b, 2018c, 2019a, 2019b, 2019c, 2019d, 2020). Finally, narrative face-to-face interviews were also conducted with eight of the head teachers in Mathare's non-formal schools in May 2018. All the interviews were audio recorded and fully transcribed. An overview of the interview data is given in Table 3.

¹¹ With regards to 'introspection', please see the explanations on the method of linguistic pruning in *section 3.4.2* and *section 3.4.4.4*.

¹² "Triangulation means that researchers take different perspectives on an issue under study or – more generally speaking – in answering research questions." (Flick 2009: 445).

Interview Pa	Interview	Interview	
Name	Professional Status	Date	Duration (in h)
Patrick Ojiambo Juma	Head of Daystar Junior Educational Centre, Nairobi	09 May 2018	00:17:45
Caroline Muendi Kamuya	Head of Bright Education Centre, Nairobi	08 May 2018	00:33:30
Esther Ndunge Katundu	Head of Star Educational Centre, Nairobi	10 May 2018	00:21:55
Christopher Malusi Ngombalu	Head of Mumo Education and Orphanage Centre, Nairobi	14 May 2018	00:48:40
Otieno Kennedy Odero	Head of Success Care Centre, Nairobi	10 May 2018	00:27:37
Josephat Andula Okama	Head of Ngota's Upendo Nursery School and Youth Centre, Nairobi	09 May 2018	00:19:02
George Manyasa Olusamu	Head of Ngei P.A.G. Education Centre, Nairobi	10 May 2018	00:11:11
Dixon Odhiambo Owaga	Head of Destiny Community Education Centre, Nairobi	08 May 2018	00:36:08

Table 3: Interview Data.¹³

3.4.4.4 Data Verbalisation

By applying the barefoot economic method of linguistic pruning, the following scientific vocabulary originating from the theoretical background of this scientific study (see *section 3.3*) was pruned in the process of data collection, preparation and analysis: *economic development*, *economic growth*, *efficiency*, *production*, *consumption*,

¹³All interviews were conducted at the interviewees' schools.

wealth, market, technological progress, entrepreneurship, innovation, scalability, profit, poverty, base of the pyramid, inclusion, global capitalism, economic institutions, capacity building, bottom-up, microfranchising, business model, systematisation, replication, standardisation and branding. Stemmed variations of these words were also pruned, such as *efficient*, *entrepreneur* and *brand*. The pruned terminology is commonly used scientific jargon in the prevailing discourse in the discipline of economic science (see section 3.4.2). That discourse is regularly referred to as the "neoliberal discourse" (Max-Neef 2009: 20; see also Davies & Petersen 2005; Springer 2012; Phelan 2014; Marissa 2020). The neoliberal discourse can be regarded as an offshoot of the previous development discourse (Moore & Schmitz 1995; see also section 3.4.2); it emerged in the 1980s due to revived interest in the economic work of Joseph Schumpeter (see Giersch 1984; Plehwe 2020; see also section 3.3.2).¹⁴ In this study, the neoliberal discourse is defined as a discourse that aims at "liberating individual entrepreneurial freedoms and skills within an institutional framework characterized by strong private property rights, free markets, and free trade" (Harvey 2007: 2).

Beyond this pruning of scientific vocabulary to prevent bias, the verbalisation of data was carried out using the ordinary language of the slum dwellers as identified in the course of the fieldwork. The latter allowed for 'member-checking' as a strategy for *communicative validation* (Flick 2009; Mayring 2016), in addition to the triangulation method described above. Consequently, the early drafts of this treatise were reviewed by the entire MPITO[®] project team, as well as by the MPITO[®] head teachers, in early 2020 and their feedback has been incorporated into this final version. This was done to ensure the accurate linguistic representation of the social and economic reality of all participants (Torrance 2012).

The findings presented in the following section are verbalised by means of pruned language without the abovementioned scien-

¹⁴ Please note that Schumpeter himself should not be considered as a neoliberal thinker (Plehwe 2020).

tific terminology. With regard to the research question of why bottom-up franchises are rarely observed at the base of the pyramid, the findings demonstrate how the microfranchise concept was *translated* by the participants of the experiment in their lived experience at the grassroots. Since the concept of bottom-up franchising transfers the responsibility of creating a branded and standardised operational system to the microfranchisees, the research results primarily focus on the common actions and perceptions of the MPITO[®] schools' head teachers and let them "have their say" (Max-Neef [1982] 1992: 55). In doing so, a number of illustrative, extensive direct quotations selected from the transcribed interview data are provided. This approach also creates a high degree of close association between the presentation of the results and the collected data (see Glaser & Strauss 1967).

For reasons of meta-cognitive reflection and structure, the research findings are presented along the two (parenthesised) key performance drivers of microfranchises, namely 'branding' and 'standardisation'.

3.5 Results

3.5.1 General Observations

The bottom-up franchise experiment of the MPITO[®] project took place in the Mathare slums from January 2015 to March 2020. It ended abruptly with the government-imposed closure of all Kenyan schools due to the global 'COVID-19 pandemic'.¹⁵

As described in *section 3.4.4.1*, the MPITO[®] school network was formed to let the head teachers of the participating schools collectively "do their thing" (Max-Neef [1982] 1992: 55) in terms of their branding and standardisation responsibilities as microfranchisees. In the first head teachers' meeting of the MPITO[®] school network,

¹⁵ School reopening was planned by the Kenyan government for January 2021 (as of September 2020).



Local shop in Mathare (Photo: João Victor Novelletto Bolan)

Students preparing lunch in Mathare (Photo: João Victor Novelletto Bolan)



the decision was taken to locate the network's headquarters at the Mumo Education and Orphanage Centre, and Christopher Ngombalu, director of the aforementioned school, was elected as the chairman of the network. He remained in that position throughout the entire experiment, meaning he was in charge of convening and chairing the head teachers' meetings, as well coordinating the more extensive inter-school communication.

3.5.2 Key Performance Driver: 'Branding'

The head teachers referred to the intended MPITO® 'branding' as an effort to raise community awareness and increase mutual trust within the community (Dixon Owaga, personal interview, 08 May 2018). They used at least two different channels of personal word-of-mouth communication to spread information. Firstly, they circulated information about MPITO[®] within their own schools and introduced the project to the teaching staff, students and parents. Secondly, the head teachers arranged joint events with all the MPITO[®] schools with the aim of reaching a wider audience. The events took place outdoors or in some of the most spacious buildings in Mathare, such as churches. At these events, the school directors gave speeches about MPITO[®] to the community. Groups of students from the different schools performed songs and dances at the events. Each event was attended by hundreds of people from the slum community, meaning that before long there was common awareness of MPITO® in Mathare. Figure 9 shows an exemplary photograph of a MPITO[®] event.

The feedback from the community about MPITO[®] was widely positive. "They [the members of the slum community] perceived MPITO[®] as a good organisation" (Caroline Kamuya, personal interview, 08 May 2018). Due to increasing demand for educational services at the MPITO[®] schools, the total number of students enrolled increased by 14.1 % from the first to the third school term following the implementation of the project on the ground. However, the positive perception of MPITO[®] was not based on quality improvements or assurances related to the operational standard-



Figure 9: A MPITO[®] event in Mathare in March 2016 (Photo: Faith Ariho).

isation of the schools. In fact, the implementation of standards to ensure a certain level of quality took place much later. The positive image was based on the personalities of the head teachers and their trusted social ties within the community. These ties were not primarily economic, they were personal: the kind of relationships which develop due to the complex social interactions that exist between human beings. These relationships rendered the trust-creating function of the MPITO[®] brand almost irrelevant from the outset. As the project progressed, two further aspects became apparent. On the one hand, transference of trust occurred. People who knew one of the head teachers and trusted him or her started to trust the other MPITO[®] school directors, because of positive associations with the MPITO[®] schools' umbrella. If someone from the community was asked about MPITO[®], a typical answer followed: About the MPITO[®]? Yes, I heard of it from Mr. Christopher and I can testify that Mr. Chris is a good man. He has got that heart of helping. So, I heard it from him and if he joined it, we shall be open and very much willing to work with any [MPITO[®]] school. ... I'm sure things will be okay (George Manyasa Olusamu, personal interview, 10 May 2018)

On the other hand, the project's dependence on the existence of close social bonds in the community meant that the scalability of the MPITO[®] school network was limited. It goes without saying that a person can only nurture a limited number of close social relationships and to increase the scale of the MPITO[®] project would have meant decreasing the level of personal trust and intimacy. Hence, the project and its network had to remain on a small scale. For that reason, the head teachers, who were convinced that there was an optimum scale that should not be exceeded, did not try to grow the number of MPITO® schools. Instead, they tried to safeguard and nurture close interactions between all the MPITO[®] schools and maintain the personal atmosphere. However, they received frequent requests from non-formal schools from all over Mathare to join the MPITO® network. The MPITO® school directors described the situation as follows: "One thing that we have been able to see [is that] even the schools that are around which do not belong to MPITO[®] community, they really admire. They really want to join us." (Dixon Owaga, personal interview, 08 May 2018). Pressure from the community to include new schools grew over time and when one of the original schools, the Mathare Wisdom Care Centre, had to close because it was evicted from its building in June 2016, the head teachers decided to allow a modest expansion of the school network. Step-by-step, they admitted four additional schools: Ngota's Upendo Nursery School and Youth Centre (joined in March 2017); Ngei P.A.G. Education Centre (joined in May 2018); and Upendo Family Support and Resource Centre and Lea Mathare (also called Lea Learning Centre) (both



Serving school lunch in Mathare (Photo: João Victor Novelletto Bolan)

A child in Mathare (Photo: João Victor Novelletto Bolan)



joined in July 2019). As a result, there were ten MPITO[®] schools with over 2,000 students in total by mid-2019.

In the same period, the MPITO[®] 'brand' underwent an important shift. MPITO[®] became less a brand for distinguishing one market actor from another within a competitive environment, and more a symbol of unity and solidarity that strengthened the *esprit de corps* within the slum community. More precisely, the MPITO[®] schools became a pars pro toto for all the non-formal schools in the Mathare slums. The symbolic value of MPITO[®] may have evolved in part from the fact that most community stakeholders shared intimate social relationships, a strong sense of social identity and a number of common goals, beliefs and values. The related feeling of belonging and togetherness within the slum community was also strengthened by MPITO[®]'s shared visual appearance. This shared visual appearance was supported by the supply of branded materials, including uniform shirts for the head teachers, one hundred identical school bags and one thousand pin-back button badges. All materials were branded with the official MPITO® logo (see Figure 7). The symbol of a root (grassroot) as part of the MPITO[®] logo was perceived as a symbol for the common sociocultural roots of the slum inhabitants. Dixon Owaga, head of the Destiny Community Education Centre, described how MPITO[®] strengthened the feeling of belonging and togetherness within the community and how the sense of a common bond was supported by the shared visual appearance:

When the children from Destiny come out of the school wearing these [MPITO[®]] badges, when they meet within the vicinity where they stay, they get other children who came from other schools within the MPITO[®] network, with the same badges and the same bags. This really showed them 'Oh, so we're in the right place'. This really makes them feel that for sure there is one goal that we need to achieve, all of us. ... For me, I'm even looking forward to a day where all the teachers from MPITO[®] schools would have a

dust coat that is written 'mpito' everywhere. So, when vou go to a MPITO[®] school, vou see a teacher with that dust coat and you say, 'Oh yes, this is a MPITO[®] school and this is a MPITO[®] teacher'. That would be just great. Maybe sometime in the future, as we look forward to greatness, we are very sure, sometime we will be able and as we unite together, we are now one team and even be able to secure some bank balance. it will be possible and then we can even go ahead and look for a bus. Then write on this bus: 'mpito'. So, it's like whenever we go anywhere, it is a MPITO® bus; not Destiny Community bus, not Mumo bus but MPITO[®]. I think it will just be great. It's a lifechanging thing and everybody is feeling it. You saw the children, they are very happy and they want to say 'Oh yes, MPITO[®]!' and everything. For us, I think that when we come up with this thing that makes us look similar, it is nice. It even creates more trust between us and the community. The community comes into our schools and helps in different ways. Now, they see that this is not just a school, but this is a school the same like the other schools. A MPITO[®] network. I think, this is great (Dixon Owaga, personal interview, 08 May 2018).

3.5.3 Key Performance Driver: 'Standardisation'

In terms of the MPITO[®] schools' operational system, the head teachers understood the intended 'standardisation' as an effort to create common rules and joint programmes. Before those common rules could be set and joint programmes implemented, the head teachers tried to identify the challenges facing their schools:

We meet with all the headmaster together. We have a meeting together, we discuss our challenges, what we go through. After discussing our changes, then we also discuss what are solutions towards the challenges. I have seen it's helping us so much because you come with your challenge, maybe thinking that is the major challenge you have and you find there is someone else with a bigger challenge then yours. Maybe the person has an idea towards the small challenge you have. So, we get ideas from other headmasters and we sought out some issues which are not very serious (Christopher Ngombalu, personal interview, 14 May 2018).

Christopher Ngombalu, head of the Mumo Education and Orphanage Centre and chairman of the MPITO[®] school network, outlined how 'not very serious' challenges affecting single schools were solved by the help of others. Caroline Kamuya, head of the Bright Education Centre, gave examples of sharing competencies:

If we meet as directors, each one can say what he can teach the others to do. For example, myself, I can make a sack like this [She holds a green sack made by cotton], I can draw, I can get a thread and make. So, if we can come across other teachers, I can teach them how we can make natural shirts, being given the sacks we can make natural dress, we can make natural shirts. I can demonstrate to them and then they can go and make for their schools (Caroline Kamuya, personal interview, 08 May 2018).

Unfortunately, the schools all faced a number of similar serious challenges that could not be solved through competency and knowledge sharing. As Patrick Juma, head of the Daystar Junior Educational Centre, pointed out: "All our schools that have the same problems, they have the same challenges. We are sailing the same boat." (personal interview, 09 May 2018). The head teachers identified the following common challenges: (1) *high teacher turnover*; (2) *hunger of students / food shortages*; (3) *shortage of learning materials*; (4) *high rents*; and (5) *government requirements*. The



Children at the window of a dwelling in Mathare (Photo: João Victor Novelletto Bolan)

Students at break time in Mathare (Photo: João Victor Novelletto Bolan)



head teachers agreed that these challenges could largely be overcome through better funding. Moreover, since the head teachers were open to the suspicion of corruption and other opportunistic behaviour, transparency and accountability would be necessary for the successful procurement of finances – whether through donations or school fees – and all other school resources. Transparency and accountability could only result from nurturing an appropriate moral attitude:

One thing I learnt myself or I put in place in my heart since the beginning: I realised that whenever you are working in any organisation or in any institution or whatever, first of all, you need to respect yourself. You have also to be accountable for your life, first of all, even before accounting to anybody. I'm always pushed by the way I do my work because I always talk about transparency and accountability. The way someone can be transparent to whatever you are doing. By this you need everybody to know what you are doing. Every time everybody needs to know how much you have spent for this. Also, the people who are donating to your project must know what you're doing with the money, what you're doing with the materials you have. ... Avoid selfishness because whenever you have a project and then you are selfish you will always need everything to be yours. So, this is all what I avoid to have in my life because we are just here for a while in this world and one day maybe I will not be there. Even if I will be given the whole world and someone is suffering that is like not supporting that person and I will die and that person will also die because of having problems. So, it's better for me, whatever I have and whatever I'm getting and whatever is meant for someone let it go to that person. I don't need it because that belongs to that person. What belongs to me, yeah. When I have enough, I should also use it

for somebody else. Also, to develop the other person to be more than me or like myself. Mostly when we are supporting the needy children, we want them to be more than us because I didn't have someone to take me to school, I took myself. I don't want that child to also have that same situation as I had. I want to develop that. That's my point which caused me to implement this kind of program where we support the needy of the neediest so that they may not suffer the same way I suffered, because in Mathare there are so many children who suffer a lot. They only have like a meal in a day in their families and sometimes they sleep without eating. They wait for tomorrow if the food will be there because their parents go to Eastleigh [a neighbouring district] to go and try to find some small jobs where they can wash clothes and then they are given like 50.00 shillings or 100.00 [shillings; KSh 100.00 equate to around US\$ 1.00]. They go home, buy some food and they give to their children. So, we are trying to develop them so that they may not also be like the way their mothers suffer and they may not be dependent again (Christopher Ngombalu, personal interview, 14 May 2018).

In line with Christopher Ngombalu's view, all the head teachers emphasised the need for a moral attitude encompassing the ethical principles of sufficiency, modesty and humility in one way or another. Caroline Kamuya described her commitment as follows:

What makes me stay here is that love of children. I have very good academic certificates and if I look for another job, I can even get a better job which can pay me more. That love of children is what makes me stay at Bright. I am also not intending to close Bright Education Centre whatever the circumstances. I feel encouraged, I like the young children and I stay here hoping that we shall improve as Bright Education Centre. I as the director, opened the school not because of money, not looking for money but looking for a conducive environment for children. That's why I'm still in Bright and I'm not expecting to look for other jobs (Caroline Kamuya, personal interview, 08 May 2018).

In a similar manner, Esther Katundu, head of Star Educational Centre, described how her school aimed to support the children of Mathare, as well as the entire slum community:

[There are] street boys, they don't go to school, they are just there. If they become like five years, they start snatching our phones, our money. They start standing with knives. ... When they go to school and they excel, now they exam, they get better jobs and now they come back to community. You know, if they come back to community now, they can help others and they build the others. They build themselves and also the community, because for example if someone has gone out, has done well and has got a good job, now he comes back, he can have a business and also, he can also make another school, so by so doing, you find that we are eliminating the poverty. We are eliminating the poverty in the slums. Ja, that is how we are eliminating (Esther Katundu, personal interview, 10 May 2018).

The aspiration of giving back to the community, which Esther Katundu expects from her students, also applies to the head teachers themselves as they are community members who were born, or at least grew up, in the slums and now support its local development. Kennedy Odero, head of Success Care Centre, said about himself:

Being a child who grew up in the community ... [He breaks up and starts again.] I grew up in the community, I schooled in the community, so I had that passion



Class Five student from Mathare wearing Kanga blanket (Photo: João Victor Novelletto Bolan)

Boys running in the rain along Mau Mau Road in Mathare (Photo: João Victor Novelletto Bolan)



of coming back and give back to the community where I came from (Otieno Kennedy Odero, personal interview, 10 May 2018).

Dixon Owaga described his professional career, his motives and his social ties with the community as follows:

When I came out of the school I came back to Mathare where I used to stay ... and I thought it's wise to give back to the community. Giving back to the community in the sense that ... I went into a school and I began teaching. I taught in that school for around two years, just equipping myself and seeing to it that I gain enough experience. My main aim was to also start a school and then try to offer education to children who also went through challenges like I did. This is why I was only gaining experience ..., I was just working voluntarily. I was not being paid because I wanted to give back to the community. ... Then after two years, ... I had a number of friends. ... This was good friends. ... They sat me down and we discussed. I was telling them 'Look here guys, I'm now equipped. I'm a trained teacher. How best can I help the community? I want to support. In this case, I want us to come together and form an organisation.' (Dixon Owaga, personal interview, 08 May 2018).

Deeply embedded in the community and its social interactions, all the school directors had a strong personal interest and motivation to be transparent and accountable to the community. Furthermore, Dixon Owaga stressed that having 'good friends' in the community was necessary to run a school in Mathare. Josephat Andula Okama, head of the Ngota's Upendo Nursery School and Youth Centre, reinforced Dixon Owaga's statement by stating that "this work needs many friends; one person cannot do it" (Josephat Andula Okama, personal interview, 09 May 2018). Indeed, many non-formal schools in the slums are registered as *community-based* organisations (CBOs), which legally require at least twelve members.¹⁶ The cultivation of good friendships within the schools was also considered as a way of overcoming high teacher turnover.

The biggest challenge that I have as the founder is the teacher turnover rate. High rate of teacher turnover. Teachers come and they go when ... they feel there is greener pasture (Dixon Owaga, personal interview, 08 May 2018).

It was assumed that high teacher turnover was fundamentally caused by the lack of funds and resultant low teacher salaries:

They [teachers] keep on coming, going, coming, going, because what we charge is very little, what we give them is not what we call a salary. We give them like an appreciation. Like in my case, I pay like 4,000.00 - 5,000.00 [KSh; around US\$ 40.00 - US\$ 50.00; per month]. You see that is not really what can be called a salary (Otieno Kennedy Odero, personal interview, 10 May 2018).

On the other hand, the point was made that the low salaries could be compensated for by the creation of group cohesiveness and non-hierarchical team structures:

All my teachers, we are working as a team in our school. I have seen the advantage of working as a team because whenever you work as a team you grow stronger. You develop trust ... and that's why we remain intact, we remain together. The other thing is also the issue of including the teachers to understand

¹⁶ For that reason, the schools are sometimes also referred to as 'community-based schools' or 'community schools' (Cheng & Kariithi 2008).

exactly what you are doing and also make them partisans of the project. It has also grown Mumo to be a strong school because, since then, all the teachers I employed at Mumo, I never sacked a teacher. I understand their weaknesses. I sit down with them. I talk to them. We share our information. It's not that they are so good because they are just human beings like others. I have my weaknesses; they have their weaknesses but we understand each other. I try to understand them. I don't want to stay there as a boss. I'm not a boss for them. I want to be a servant for them. So, I try to let them know that that is their school. They belong there. They have a word and I take their decisions, sometimes they have better ideas than mine. So, we sit down, we come up with these ideas, we bring them together. What is perfect, what is good and what we decide that this is the best way we follow because we want to develop. We want to move from step to step and I cannot move the school alone. I can only move the school with the team which I have, the team which we are working together (Christopher Ngombalu, personal interview, 14 May 2018).

The need for such close social bonds between school directors and the teaching staff also limited the overall size of the non-formal schools.

As a result of their discussions about transparency and accountability, the head teachers decided to set the following common rules for all MPITO^(R) schools:

- Teachers should not carry MPITO[®] property to their homes.
- If any school is not transparent and accountable ..., it will be removed from MPITO[®].
- Every school director should have a copy of all his/her teachers National ID card.



Classroom at Ngota's Upendo Nursery School and Youth Centre (Photo: João Victor Novelletto Bolan)

Patrick Ojiambo Juma in front of Daystar Junior Educational Centre (Photo: Christopher Malusi Ngombalu)



- All MPITO[®] property should be used within the schools.
- A team of four (4), 2 MPITO[®] officials and 2 appointed MPITO[®] directors should be in charge of monitoring MPITO[®] property once per term.
- The members should not miss more than two consecutive meetings.
- MPITO[®] schools should participate in all MPITO[®] activities.
 - Member's whose schools will not participate in exams should refund the expense incurred per pupil during the exam period.
 - (2) Schools that fail to participate in other MPITO[®] activities should pay a fine of 1,000.00 [KSh; around US\$ 10.00].
- Members should keep time whenever we have meetings.
- Members should attend meetings in MPITO[®] uniform.

(MPITO[®] Interim Report 2019c: 3-4, own italics).

MPITO[®] property refers to the common property that was collectively acquired by the schools for carrying out MPITO[®] activities. These MPITO[®] activities took the form of joint programmes, in which all MPITO[®] schools were obliged to participate. Six joint programmes were created by the school directors: (1) MPITO[®] teachers' union, (2) MPITO[®] joint exams, (3) MPITO[®] sports and games, (4) MPITO[®] computer classes, (5) MPITO[®] awards, and (6) MPITO[®] chama.¹⁷ While the MPITO[®] teachers' union programme

¹⁷ A seventh programme, MPITO[®] *library*, aimed to establish and operate a joint library across the MPITO[®] schools. This programme had not yet been implemented by March 2020.

was intended to strengthen the involvement of teachers and their position in the schools with the aim of reducing the high teacher turnover, the other programmes were intended to harness synergies between the schools to "use little to make the work broader" (Christopher Ngombalu, in Caroline Kamuya, personal Interview, 08 May 2018) and, in so doing, to optimise the use of their limited local resources. A brief overview of the programmes follows.

(1) MPITO[®] Teachers' Union

The central idea of the MPITO[®] teachers' union programme was to bring the teachers (numbering over 70) from the MPITO[®] schools together. The programme was not fully implemented since a consensus was not reached on its specific design. It was "implement[ed] the idea of bringing subject teachers together, so that they can discuss and find solutions to the common challenges they face in the various subjects that they teach" (MPITO[®] Interim Report 2019a: 3). At this stage, there were six MPITO[®] teachers' sub-unions for the subject areas of Mathematics, English, Kiswahili, Social Studies, Science and C.R.E. (Christian Religious Education), which met regularly.

(2) MPITO[®] Joint Exams

The MPITO[®] joint exams programme brought together the final year students (Class Eight candidates, numbering around 80) from all the MPITO[®] schools to prepare them for their KCPE exams. The students sat tests and mock exams together. The venue rotated among the biggest MPITO[®] school halls and the schools provided teachers to coordinate the MPITO[®] exams. Furthermore, the programme enabled the final year students from the smaller MPITO[®] schools to sit the official KCPE exams in the bigger school halls of other MPITO[®] schools. This was necessary for small MPITO[®] schools that did not fit the government requirement of "spacious rooms where you can have at least five teen candidates sitting in one room with a space of one-meter space from a pupil to
the other" (Christopher Ngombalu, personal interview, 14 May 2018). These small MPITO[®] schools were not allowed, therefore, to examine candidates in their own schools. Before the introduction of the joint exams programme, candidates from the small MPITO[®] schools often had to sit their KCPE exams in the unfamiliar environment of a school outside the community.

(3) MPITO[®] Sports and Games

The MPITO[®] sports and games programme focused on holding joint event days where students from all the MPITO[®] schools could participate in extracurricular activities, such as sports tournaments. The event that kicked off the programme took place in July 2019. However, hosting the first event was a logistical challenge for the schools because of its size, so plans were made to reorganise the range of extracurricular activities into a number of smaller inter-school clubs (in sports, music, drama, etc.). The intention was for these MPITO[®] *clubs* to be run collectively by all the schools.

(4) MPITO[®] Computer Classes

The MPITO[®] computer classes programme offered joint computing lessons for MPITO[®] school students. During the lessons, the pupils – who largely lack elementary computer skills – learnt the basic operation of a computer and, later, how to use word-processing and calculation software. The forty laptops provided by the MPITO[®] project in cooperation with the non-profit organisation Labdoo were used in the lessons. The programme was piloted at the Mumo Education and Orphanage Centre before it was introduced across the MPITO[®] school network. Since June 2018, the lessons had been given by a professional computer science teacher volunteering for the MPITO[®] project.



Josephat Andula Okama teaching new primary school children (Photo: João Victor Novelletto Bolan)

School building of Success Care Centre (Photo: Patrick Thomas Kletzka)



(5) MPITO[®] Awards

The MPITO[®] awards programme honoured "the best performed students and best subject teachers, for last year's KCPE results" (MPITO[®] Interim Report 2019a: 3). The award winners were selected from all the MPITO[®] students and teachers, regardless of their school. The MPITO[®] awards provided prize money of 2,000 Kenyan shillings (around US\$ 20) for the best performing students and 1,000 Kenyan shillings (around US\$ 10) for the best subject teachers. Furthermore, the best MPITO[®] schools – judged on their overall student performance in the KCPE exams – received a trophy. The awards were presented to the winners at a joint ceremony at the beginning of each school year.

(6) MPITO[®] Chama

In Kenva, micro-savings and micro-investment groups are commonly known as 'chamas'.¹⁸ Traditionally, chamas are organised as rotating savings and credit associations (ROSCAs) colloquially called 'Merry Go Round chamas' - where a group of people pool a fixed amount of money and give the total to one of its members so the member can make an investment; the beneficiary member rotates with each round of pooling. The idea of establishing a MPITO[®] chama was discussed by the head teachers at the beginning of the experiment in 2015 but was only implemented in 2018. The head teachers decided to reallocate US\$ 800 from the planned branding budget to an initial MPITO[®] chama pool. For the first round, "the headmasters agreed that the investment money should be divided equally among the schools" (MPITO® Interim Report 2018b: 4), so that each school could make a micro-investment of US\$ 100. In terms of further investment rounds, "the members [the head teachers] agreed to start savings for MPITO®

¹⁸ The Kiswahili word 'chama' can be translated as 'group', 'association', 'party', or similar.

and agreed that every member should contribute a 1,000.00 [KSh; around US\$ 10.00] monthly" (MPITO[®] Interim Report 2019c: 4). The MPITO[®] Interim Report No. 5 (2019b) gives two examples of micro-investments made by the MPITO[®] schools:

The fourth school visited was Daystar school, managed by Mr. Juma. ... As for the investments, the school used the money to establish a small uniform distribution centre that is run and managed by two women, they make uniforms at cheaper prices for the students as well as making uniforms for other schools around. Mr. Juma reported that they intend to get another sewing machine in the future since sometimes the demand is higher than the supply especially when schools reopen. So far, the business is only limited to making school uniforms but there are plans to diversify in the future and make casual clothes

The fifth school to visit was Destiny school, managed by Mr. Dixon. ... As for the investment, the school established a photo studio as well as a photocopying business. The business is used to make school documents, as well as make school exams. The photo studio also produces photos of pupils that might have been taken during a school trip. The money that would have been used to produce this school documents is then used to pay for other school needs.

(MPITO[®] Interim Report 2019b: 2-3).

The example of Destiny Community Education Centre's micro-investment may be characteristic of many of the joint activities in the MPITO[®] school network. Among other items, the school invested in a photocopier for its own use and to establish a business. This investment not only ben-

efited Destiny as the investing school, but also the entire MPITO[®] school network since the other schools could duplicate learning materials using Destiny's new photocopying business more cheaply than using the services of a copy shop outside the slum community. In this way, the schools further strengthened their relationships with each other by using their economic activities to support their internal network, instead of promoting inclusion in the bigger, external markets outside the slum. In doing so, they also prevented an outflow of their limited resources.

3.6 Discussion

3.6.1 Preliminary Remarks

The previous section presented the findings of the real-world experiment within the scope of methodologically pruned language. The following section discusses these results by returning to the previously-used scientific terminology and reflecting on the results against the theoretical background of Schumpeter's Theory of Economic Development, Prahalad's BoP concept and Fairbourne et al.'s concept of microfranchising. At the same time, the reflections are theoretically underpinned by referring to an additional body of social scientific literature that has been identified as coherent with the findings. This reference literature involves Niklas Luhmann's concept of inclusion/exclusion, Karl Polanyi's concept of embeddedness and Manfred Max-Neef et al.'s Theory of Human Scale Development. Finally, the discussion offers a theoretically substantiated answer to the posed research question of why bottom-up franchises are rarely observed at the base of the pyramid.

3.6.2 Niklas Luhmann's Concept of Inclusion/Exclusion

First and foremost, the results of the real-world experiment show how its empirical intervention stimulated the slum dwellers to implement self-reliant poverty alleviation activities, which did not identifiably propel them into markets outside their slum community. In this respect, the findings indicate a clear break with Prahalad's BoP concept, particularly its normative proposition of alleviating poverty through the inclusion of the poor in the markets of global capitalism and its "representation of the poor as eager participants in globalized markets" (Peredo et al. 2018: 414). In order to explain why the inhabitants of the slum did not drive their own inclusion, it is appropriate to "look beyond the 'feelgood mantra' of inclusion" that prevails in the context of poverty research (Meagher 2015: 836). A basic investigation begins with the concept of inclusion/exclusion itself. Previous research has already demonstrated that "the inclusion/exclusion debate leaves much to be desired with regard to conceptual clarity", and that "this lack of conceptual clarity might be less innocuous than it looks" (Braekman 2006: 66). In an effort to establish greater conceptual clarity, researchers regularly draw on the concept of inclusion/exclusion as it was first introduced by the famous German sociologist Niklas Luhmann (1927-1998) in his 1975 Systems Theory of Society (original German title: Systemtheorie der Gesellschaft) (Luhmann 2017).

Based on the mid-20th century communication theory (e.g. Watzlawick et al. 1967), Luhmann (2017) defines 'the social' as human communication. Hence, to Luhmann, society can be viewed as the totality of human communication. Society, in turn, is divided into social systems (communication systems). According to Luhmann (2006: 37), human communication becomes a social system if it is "self-referential", i.e. if it creates its own closed communication loops. In his theoretical framework, Luhmann introduced the terms inclusion and exclusion to describe the extent of participation of human individuals or groups in the communication process of a certain social system. Thereby, Luhmann points out that inclusion always generates opportunity costs, which arise from the fact that someone who takes part in the communication process of one social system cannot *at the same time* take part in the parallel communication process of another social system. In other words, someone who is *included* in the communication of one social system is ipso facto inevitably excluded from the parallel communication of another social system. For that reason, there can never be 'full' inclusion, but only partial multi-inclusion, i.e. partial inclusion in the communication of a certain number of different social systems (see also Nassehi & Nollmann 1997: Braeckman 2006; Schirmer & Michailakis 2015). Consequently, "Luhmann [...] raise[s] fundamental questions with respect to the implicit norm of full inclusion which still dominates the debate on inclusion and exclusion" (Braekman 2006: 65, own emphasis). By the same token, Luhmann's conceptualisation makes 'full' exclusion almost impossible (Nassehi & Nollmann 1997; Stichweh 1997; Braeckman 2006). To achieve full exclusion, humans would have to fall outside society, i.e. outside communication of all kinds (ibid.). Moreover, Luhmann's concept of inclusion and exclusion dispels "the normative misunderstanding that the occurrence of social exclusion is *per se* a problem, with the consequence being that inclusion is seen as the solution" (Schirmer & Michailakis 2015: 46, own italics). Whether or not inclusion is considered desirable actually depends on the opportunity costs and, hence, on the characteristics of the affected social systems. To give an example: through compulsory schooling, children are largely included in the education system. They are, however, simultaneously largely excluded from the economic system by means of the prohibition of child labour (Kronauer 2009).

After several expeditions to the Brazilian favelas in the 1990s, Luhmann set out his hypothesis that the base of the pyramid is generally excluded from the social systems of capitalist society (Luhmann 1995a: 250, 260, 1997: 632; see also Luhmann [1995b] 2008). This hypothesis can, among others, be considered as fundamental for Prahalad's BoP concept. However, as demonstrated above, Luhmann's concept of inclusion and exclusion also suggests that – since people at the base of the pyramid naturally communicate with each other – the advantages of inclusion of the base of the pyramid in capitalist society can only be evaluated if the characteristics of the affected social systems are taken into account. Having actively participated in the communication at the base of the pyramid by means of language pruned in accordance with the ordinary language of the inhabitants of the slum, this scientific study is now in a position to provide a description of relevant social system characteristics. To do this, reference is made to Karl Polanyi's 1944 *concept of embeddedness*¹⁹ and Manfred Max-Neef et al.'s 1986 *Theory of Human Scale Development*.

3.6.3 Karl Polanyi's Concept of Embeddedness

The Austro-Hungarian economist, anthropologist, sociologist and historian Karl Polanyi (1886–1964) introduced his concept of embeddedness in his 1944 classic, *The Great Transformation*, in which he describes the historical transformation of 20th century modern society into what he calls a "market society" (Polanyi [1944] 2001: 60). He identifies four traditional allocation patterns according to which economic processes have been organised in human history. These are: (1) autarchy (householding); (2) symmetry (reciprocity); (3) centricity (redistribution); and finally (4) market (trade) (ibid.: 59–60). Moreover, Polanyi (1977: 51, own italics) points out that all economic activities – regardless of their pattern – were originally "*embedded* in social relations of a noneconomic kind". These non-economic social relationships created mutual consideration,

¹⁹ Polanyi's 1944 concept of embeddedness should not be confused with Granovetter's 1985 concept of the same name. "[The concept] of embeddedness advocated by Mark Granovetter (1985), which led to the widespread use of the term in the new economic sociology, differs fundamentally from the meaning of the term in the work of Karl Polanyi. [...] According to Granovetter, [all] economic action is 'embedded in concrete, ongoing systems of social relations' (1985: 487), in other words, in actors' social networks. [...] Small wonder that institutional economists and rational choice sociologists eagerly took up this notion of embeddedness, since they could readily incorporate it into a rational choice framework" (Beckert 2007: 8–9, own emphases). "In *The Great Transformation*, Polanyi did not aim to [...] explain the social preconditions for market efficiency; he was concerned with what happens to social order [...] when economic exchange is organized chiefly through self-regulating markets." (ibid: 17, own emphases).

trust and confidence between people, making economic institutions dispensable.

No institutionally separate economic system – no network of economic institutions – could be said to exist. ... while there was, of course, an economic system in being, it was not institutionally separate. In effect, it was simply a by-product of the working of other, noneconomic institutions (Polanyi 1977: 51–52).

According to Polanyi, this social order changed when the market pattern became the paramount economic pattern. At this point, market activities outgrew non-economic social relationships and, consequently, the establishment of separate, economic institutions became necessary to restore relationships of trust insofar as they were conducive to the efficiency of markets (Beckert 2007).

Referring to Richard Thurnwald's (1869-1954) 1932 ethnological study, Economics in Primitive Communities, Polanvi ([1944] 2001: 61) also indicates that non-market societies, i.e. societies in which economic activities are embedded in social relations of a non-economic kind, still prevail in poor regions of the world. The present scientific study empirically validates Polanyi's assessment. For example, it demonstrates how the economic institution of the MPITO[®] brand – deemed necessary to create trusted customer ties within the 'anonymous' markets of global capitalism - became redundant because trust was created through the inextricable link between market activities and non-economic social relationships within the slum community. Seen in this light, the BoP concept's prominent characterisation of institutions at the base of the pyramid as 'weak', in the sense of deficient, and the related call for institutional capacity building to selectively ensure the efficiency of market interactions (see section 3.3.3), overlooks the fact that the absence of economic institutions is an essential characteristic of the social order in non-market societies. The legitimacy of forced reconfiguration of this working social order by institutional capacity building may be questioned. If, as suggested by the BoP

concept, a bottom-up approach is applied, this case study indicates that the establishment of separate economic institutions by the people at the base of the pyramid themselves is unlikely because they generally do not perceive the need for them. Furthermore, the study demonstrates that the slum dwellers tried to *re-embed* market interactions into the social relationships of their community and, by doing so, acted in the opposite direction to inclusion in the globalised markets of modern capitalism.

In summary, an essential difference in the characteristics of the social systems inside and outside global capitalism has its roots in the degree of social embeddedness of the economic activities. However, the findings not only indicate that economic activities at the base of the pyramid are predominantly embedded in non-economic social relationships, they also allow for a description of the scale of these social relationships. The following section puts forward the argument that the social sphere at the base of the pyramid should not only be characterised as a non-market society but – with reference to Max-Neef et al.'s *Human Scale Development theory* – also as a *human scale society*, i.e. a society in which social relationships remain with a distinct human-scale dimension.

3.6.4 Manfred Max-Neef et al.'s Theory of Human Scale Development

In 1986, Manfred Max-Neef and his colleagues introduced their theory of *Human Scale Development* (original Spanish title: *Desarollo a Escala Humana*). Emphasising that "development is, among other things, a problem of scale" (Max-Neef et al. [1986] 1989: 13), an essential element of this theory is the *concept of human scale*. The concept of human scale can be traced back to the ancient philosophies of Pythagoras, Plato and Aristotle, since when it has been adopted and interpreted in various scientific disciplines including architecture, psychology, primatology and anthropology. The concept was first introduced to the field of economics by Alternative Nobel Prize winner Leopold Kohr (1909–1994) in the 1940s, and later became prominent via his protégé, E. F. Schumacher (1911– 1977).

In Human Scale Development, Max-Neef et al. ([1986] 1989: 51) define the 'human scale' sociologically as "a scale where the social does not annul the individual but, on the contrary, the individual may empower the social." In this definition, the human scale is associated with the significance (relevance, importance) of the individual for the social. More precisely, the human scale describes a scale where every single human individual is a significant, i.e. essentially determining, part of the social whole. Being such a part, an individual cannot be separated or substituted by another individual without affecting the essence of the social whole.²⁰ Intimate social ties, social cohesion and, ultimately, a genuine sense of social identity, integration and responsibility are only possible within the human scale (Max-Neef [1982] 1992: 132; see also Sale 2017). Since single individuals tend to lose their significance as the size of the social whole increases, Max-Neef pointed out that "human scale must be small; there cannot be a big human scale" (Max-Neef 2019, own transcript; see also Kohr 1957; Schumacher [1973] 2011; Sale 2017). Moreover, social relationships that remain within the human scale typically form social institutions that remain within the human scale. Such human-scale institutions include family, neighbourhood and local community (Max-Neef et al. [1986] 1989). To illustrate the human-scale character of, for example, a family as a traditionally institutionalised mother-father-child relationship, it is clear that neither the mother, nor the father, nor the child can be separated or substituted by another individual without significant consequences for the whole family.

Since one of the most remarkable manifestations at the base of the pyramid is the wide spectrum of small-scale social and economic activities – particularly evident in socially embedded micro-

²⁰ The opposite case of being *no* essentially determining part is described by Max-Neef ([1982] 1992) as a state of being *alienated*. Hence, to Max-Neef, the phenomenon of alienation is the inevitable result of exceeding the human scale.

entrepreneurship – social relationships and institutions at human scale can be said to be an essential characteristic of the social sphere outside global capitalism. For that reason, Max-Neef ([1982] 1992) described the society at the base of the pyramid as a *human scale society*. In contrast, the social relations and institutions of global capitalism tend to exceed the human scale. Consequently, Max-Neef et al. ([1986] 1989) propose the theory of Human Scale Development as a development theory that is more coherent with the social systems at the base of the pyramid than other development theories. At this point, we should remind ourselves of Schumpeter's Theory of Economic Development, which intended to describe the form of economic development *within* capitalist societies.

Human Scale Development theory contradicts Schumpeter's Theory of Economic Development in its basic postulate: "Development is about people and not about objects." (Max-Neef et al. [1986] 1989: 19). As described in section 3.3.2, at the core of Schumpeter's capitalist development theory lies his concept of combinations, by which he defines development as the carrying out of new combinations of objects. In contrast, at the core of Human Scale Development theory lies the concept of fundamental needs, which defines development as the improved satisfaction of the fundamental needs of people.²¹ Due to its emphasis on fundamental human needs, Human Scale Development theory can – in contrast to Schumpeter's Theory of Economic Development – directly integrate the needs-based concept of poverty as set out by the United Nations and the World Bank (see section 1.1). Max-Neef et al. [1986] 1989: 21) acknowledge that "any fundamental human need that is not

²¹ Please note that Human Scale Development theory's original "anthropocentric view restricted to human needs" was overturned in the 2011 paper, *Should We Care About the Needs of Non-humans?* by Jolibert, Max-Neef, Rauschmayer & Paavola (2011: 260). They showed that Human Scale Development theory can also be applied to the fundamental needs of non-human living beings. In this broader sense, the basic postulate of Human Scale Development theory can be reformulated as: 'Development is about subjects and not about objects.'

adequately satisfied, reveals a human poverty". The concept of fundamental needs also leads to two additional postulates:

First: Fundamental human needs are finite, few, and classifiable. Second: Fundamental human needs ... are the same in all cultures and in all historical periods. What changes, both over time and through cultures, is the way or the means by which the needs are satisfied (Max-Neef et al. [1986] 1989: 20).

In other words, Human Scale Development theory considers the existence of *finite*, *subjective-universal* needs, which are "essential attributes related to human evolution" (ibid.: 30). This perspective opposes most other economic theories, which avoid the issue of fundamental needs by reference to *infinite*, *subjectiveparticular* preferences or wants (ibid.). Moreover, Max-Neef (2010b: 206) points out: "In conventional economics we have two links: wants and goods. In Human Scale Development theory we have three links: Needs, satisfiers and goods." Max-Neef et al. describe the difference between satisfiers and goods as follows:

While a satisfier is in an *ultimate sense* the way in which a need is expressed, goods are in a *strict sense* the means by which individuals will empower the satisfiers to meet their needs. ... Hence, *satisfiers are what render needs historical and cultural, and economic goods are their material manifestation* (Max-Neef et al. [1986] 1989: 27–29).

To illustrate the triad of needs, satisfiers and goods, Max-Neef (2010b: 206) gives the following example: "[Imagine] there is the need of Understanding, whose satisfier is literature, whose good is a book".²² Based on this idea, Human Scale Development theory

²² Please note that Max-Neef's use of the notion of 'understanding' here is not in the strict phenomenological sense as presented in *section 2.2.4*.

does not focus on the allocation or accumulation of economic goods, but rather on the classification of: (1) fundamental needs; and (2) their satisfiers.

In terms of the classification of fundamental needs, Max-Neef et al. ([1986] 1989: 20) propose dividing human needs into two categories: existential and axiological. Within the existential category, they identify "the needs of Being, Having, Doing, and Interacting" (ibid.). Within the axiological category, they identify "the needs of Subsistence, Protection, Affection, Understanding, Participation, Idleness, Creation, Identity and Freedom" (ibid.).²³ To combine both categories, Max-Neef et al. created a needs matrix that can be completed by different satisfiers (see *Table 4*).

Human Needs	Being	Having	Doing	Interacting
Subsistence				
Protection				
Affection				
Understanding				
Participation				
Idleness				
Creation				
Identity				
Freedom				

Table 4: Matrix of Fundamental Human Needs (Max-Neef et al. [1986] 1989: 33, adapted)

In terms of the classification of satisfiers, Max-Neef et al. ([1986] 1989) and, later, Jolibert et al. (2011) proposed two complementary typologies. While the first is "intrahumanly", i.e. it refers to the satisfaction of needs "in relation with oneself" (Max-Neef [1986]

²³ In this context, Max-Neef et al. ([1986] 1989: 29) consider it "likely that in the future the need for Transcendence, which is not included in [their] proposal, as [they] do not yet consider it universal, will become as universal as the other needs."

2005b: 48), the second one is "inter-humanly", i.e. it refers to the satisfaction of needs "in relation to others" (ibid.).²⁴ The intrahuman typology identifies the following five types of satisfiers: "(a) violators or destroyers, (b) pseudo-satisfiers, (c) inhibiting satisfiers, (d) singular satisfiers, and (e) synergic satisfiers" (Max-Neef et al. [1986] 1989: 32). The inter-human typology, in turn, identifies two different types of satisfiers: (1) divergent satisfiers, and (2) convergent satisfiers (Jolibert et al. 2011: 260). A brief overview of all types is given below.

(a) Violators/destroyers are elements that pretend to satisfy a given need, but then annihilate the possibility of its satisfaction over time (Max-Neef et al. [1986] 1989: 32-34). They also impair the adequate satisfaction of other needs (ibid.: 34). Examples include censorship or bureaucracy, both of which pretend to satisfy the human need of protection. However, they do not achieve this and actually impair the adequate satisfaction of other human needs, such as participation, affection, creation and freedom (ibid.). (b) Pseudo-satisfiers are elements that generate a false sensation of having satisfied a given need (ibid.: 34). Examples include status symbols and fashion trends for the human need of identity; ageism, racism or sexism for the human need of identity; prostitution for the human need of affection; or formal democracy for the human need of participation (ibid.: 35). (c) Inhibiting satisfiers are elements that generally oversatisfy a given need and, thus, curtail the possibility of the adequate satisfaction of other needs (ibid.: 34).²⁵ Examples include paternalism, which oversatisfies the human need

²⁴A third typology is supposed to be developed referring to the satisfaction of needs in relation to the future (compare Spiering & del Valle Barrera 2021). Thereby, Human Scale Development theory claims that satisfiers which do not impair future needs satisfaction have to articulate themselves organically, i.e. they cannot be established by technocratic planning and artificial engineering, but rather have to realize themselves in an evolutionary process (Max-Neef et al. [1986] 1989).

²⁵ Needs can be *over*satisfied, since they are finite. Oversatisfaction is thereby regarded as a form of inadequate satisfaction of needs.

of protection and curtails the satisfaction of human needs such as participation and freedom; or Taylorist production that oversatisfies the human need of subsistence and curtails the satisfaction of human needs such as understanding, creation and freedom (ibid.: 35). (d) Singular satisfiers are elements that adequately satisfy one given need and do not impact on the satisfaction of other needs (ibid.: 36). Examples are curative medicine for the human need of subsistence; insurance systems for the human need of protection; or gifts for the human need of affection (ibid.). (e) Synergic satisfiers are elements that adequately satisfy a given need and simultaneously stimulate and contribute to the satisfaction of other needs (ibid.: 36). Examples include subsistence agriculture, which satisfies not only the human need of subsistence, but also contributes to the satisfaction of other human needs such as creation, understanding, protection and freedom; or preventive healthcare, which satisfies not only the human need of protection, but also contributes to the satisfaction of human needs such as subsistence and understanding (ibid.: 37). (1) Divergent satisfiers are elements that are intended to satisfy someone's own needs, but which simultaneously undermine the ability of others to satisfy their needs (Jolibert et al. 2011: 260). (2) Convergent satisfiers are elements that are intended to satisfy someone's own needs and simultaneously enhance or, at least, do not impair the ability of others to satisfy their needs (ibid.).

Having worked out these typifications, Human Scale Development theory ultimately argues in favour of *a development strategy that is based on (2e) convergent-synergic satisfiers* (Max-Neef et al. [1986] 1989; Jolibert et al. 2011, 2014). Thereby, Human Scale Development theory claims that *these satisfiers can only be generated endogenously and within the human scale* (Max-Neef et al. [1986] 1989). The endogeneity argument stresses that convergent-synergic satisfiers can only be generated by the people "whose development is at stake" and, hence, cannot be imposed or induced from the outside (Guillén-Royo 2016: 47; see also Max-Neef et al. [1986] 1989: 36). The subsequent human scale argument maintains that convergent-synergic satisfiers can only be generated by the people

whose development is at stake if they constitute a social whole of which every individual is an essentially determining part. To substantiate these arguments, Max-Neef et al. ([1986] 1989), among others, draw on the example of breastfeeding. Breastfeeding can be considered as a satisfier usually only generated endogenously by a mother and her child. Moreover, the mother-child relationship is a social relationship at the human scale, since neither the mother nor the child can be substituted by another individual without affecting the essence of that relationship. Breastfeeding can be viewed as a synergic satisfier, as it not only adequately satisfies the infant's need for subsistence but simultaneously contributes to the satisfaction of the infant's needs for affection, identity and protection. Furthermore, breastfeeding can be viewed as a convergent satisfier as it not only adequately satisfies the infant's needs, but simultaneously contributes to the satisfaction of the mother's needs for affection and identity.

The absence of such convergent-synergic satisfiers beyond the human scale, according to Max-Neef et al., makes the harmonious satisfaction of human needs hardly possible (not to say impossible). Individuals then often attempt to compensate for their inadequate satisfaction of needs by the consumption of more economic goods. This greater consumption of economic goods does not, however, lead to an increase in needs satisfaction. Referring to the example above, beyond the human scale the satisfier of breastfeeding is usually not generable; hence only bottle-feeding can be chosen as a satisfier for the infant's need of subsistence. In this case, increased consumption of the economic good of infant formula milk cannot compensate for the impaired satisfaction of needs resulting from the absence of the satisfier of breastfeeding. Overall, the de facto impossibility of harmonious needs satisfaction beyond the human scale ultimately leads to the phenomenon of greed (see Smith & Max-Neef 2011). The phenomenon of greed has been adequately described in this context by psychoanalyst and social philosopher Erich Fromm (1941: 115), as "a bottomless pit which exhausts the

person in an endless effort to satisfy the need without ever reaching satisfaction".

Shifting the poverty alleviation focus away from economic goods towards satisfiers (compare *section 1.1*), Human Scale Development theory finally decouples the concept of economic development, defined as improved needs satisfaction, from that of economic growth, defined as increased production output. As described in *section 3.3.2*, neoclassical economics considered both as conceptual equivalents. Later, Schumpeter's Theory of Economic Development substantially distinguished the two concepts, but asserted that development was impossible without growth. Max-Neef et al.'s Human Scale Development theory turn this relationship upside down and proposes that "[g]rowth is not the same as development, and development does not necessarily require growth." (Max-Neef 2010b: 204).

In summary, Human Scale Development theory argues in favour staying within, or returning to, the human scale for the sake of the harmonious satisfaction of fundamental needs. In other words, Human Scale Development theory claims that effective poverty alleviation is only possible within a distinct human-scale dimension. Since the social relationships and institutions at the base of the pyramid tend to remain within such a human-scale dimension, while those of global capitalism tend to exceed that dimension, Human Scale Development theory ultimately provides an explanation why the slum dwellers may doubt the advantages of inclusion in the globalised markets of modern capitalism as presupposed by Prahalad's BoP concept. For the same reason, the myriad microentrepreneurs at the base of the pyramid may prefer to preserve the small scales of their businesses instead of striving for scalability and large-scale operations as suggested by Schumpeter's Theory of Economic Development and the poverty alleviation concepts derived from it. The call for scalability of innovations is evident in Schumpeter's concept of creative destruction and the call for large-scale corporations is evident in the 'Schumpeter hypothesis'. Against this background, large-scale entrepreneurship and

massive market penetration is considered in Prahalad's BoP concept as essential for the profitability of inclusive businesses. Scaling by means of microbusiness replication, as suggested by Jason Fairbourne et al.'s concept of microfranchising, can also be regarded as incompatible with the human-scale dimension of the socially embedded economic activities at the base of the pyramid, since the human scale is essentially associated with the social significance of *non-replicable* human individuals. Or, to put it differently, if microenterprises are standardised by a turn-key 'business-in-a-box' solution, the significance of the individual entrepreneur is annulled.

Finally, Human Scale Development theory counters Prahalad's BoP concept with Max-Neef's 1982 "concept of revitalization" (Max-Neef [1982] 1992: 124). His revitalization concept proposes alleviating poverty not through inclusion in global capitalism and its markets but, on the contrary, by the circumvention of global capitalism and its markets (ibid.). The concept further suggests initiating "revitalization project[s]", in which "revitalization experiment[s]" are conducted at the grassroots. These experiments can be described as real-world experiments intended to stimulate the emergence of endogenous satisfiers within human-scale communities (Max-Neef [1982] 1992: 122, 124). Max-Neef's "Tiradentes Project", conducted in Brazil in the 1970s, is one example of such a revitalization project (ibid.: passim, own emphasis). The MPITO[®] project could be considered as another example. Its real-world experiment did not give rise to a scalable microfranchise system, but rather a cooperative network of microentrepreneurs socially embedded in the slum community and operating at the human scale. In this way, the MPITO[®] network was essentially generated in an endogenous manner by the slum community at the grassroots and served as a convergent-synergic satisfier, which may have simultaneously contributed to several fundamental human needs of its members as well as its community stakeholders. The needs satisfaction of the network members may have included: (1) satisfaction of the needs for participation and understanding through collective decision making; (2) satisfaction of the need for creation through the

development of joint programmes; (3) satisfaction of the needs for protection and affection through solidarity; (4) satisfaction of the need for identity through a sense of belonging and togetherness; and (5) satisfaction of the need for freedom through autonomy in decision making and the avoidance of dependence on markets outside the slum community. The specific needs satisfaction of the community stakeholders may have included: (1) satisfaction of the need for understanding through improved non-formal education; (2) satisfaction of the need for participation through extended opportunities for school enrolment; and (3) satisfaction of the need for identity through a sense of belonging and togetherness.

3.7 Conclusion

3.7.1 Summary

This scientific case study set out to use a barefoot economic research approach to answer the research question of why bottom-up franchises are rarely observed at the base of the pyramid. To achieve this: (1) a methodological pruning of language of the jargon from the prevailing academic discourse within the scientific discipline of economics was conducted; and (2) lived experiences were gained within a multi-year real-world experiment on bottom-up franchising carried out in the Mathare slums of Nairobi (Kenya).

The Mathare slums are a collection of thirteen of the most deprived and excluded informal sub-settlements characterised by the phenomenon of poverty and inhabited by a total population of approximately 200,000 people. The non-formal education sector of Mathare is dominated by native microentrepreneurs who have established and operate at least 85 small-scale primary schools with more than 16,000 pupils. A bottom-up franchise experiment was conducted from January 2015 to March 2020 with a panel of initially seven, and later ten, of these non-formal schools. Methodologically, this experiment was designed as a real-world experiment of an own independent development project, called the MPITO[®]

project. The participating schools were selected by the method of targeted sampling, in which a sample frame was created according to a pre-existing school census. Triangulated data was collected by the method of participant observation, which included the combination of active participation and direct observation with narrative interviews and document analysis. In theoretical terms, the experiment was derived from Jason Fairbourne et al.'s concept of microfranchising, suggesting the systematisation and replication of microbusinesses at the base of the pyramid with the goal of poverty alleviation. As explained, Fairbourne et al.'s concept of microfranchising is based on C. K. Prahalad's BoP concept, which stresses the ideas of: (1) large-scale entrepreneurship; (2) capacity building of market institutions; and (3) inclusion in global capitalism. Furthermore, this study demonstrated how Prahalad's BoP concept was developed from Joseph Schumpeter's Theory of Economic Development.

Following the barefoot economic method of linguistic pruning, key terminology of the theoretical background was pruned from the language used in the process of data collection, preparation and analysis. The research results of the conducted experiment were presented based on the pruned verbalisation of data. This allowed the participants of the experiment to verbalise in their own words how the bottom-up franchise concept worked in terms of their lived experience at the base of the pyramid. The primary focus was on the common actions and perceptions of the head teachers at the experiment's participating schools. The findings show that the head teachers perceived branding and standardisation differently to the conceptualisations in the theoretical background to this research. The self-reliant poverty alleviation efforts of the slum dwellers did not involve: (1) scaling business operations at the expense of close social relationships; (2) capacity building of economic institutions; or (3) inclusion in globalised capitalist markets. The subsequent discussion returned to using the pre-pruned scientific terminology, and the research findings were reflected on against the theoretical background of this study. Based on Niklas Luhmann's concept of

inclusion/exclusion, it was argued that the advantages of inclusion in global capitalism can only be evaluated if the characteristics of the affected social systems are taken into account. With reference to Karl Polanvi's concept of embeddedness, this study made the claim that the social systems at the base of the pyramid constitute a non-market society, in which economic activities are embedded in social relationships and, consequently, separate market institutions are dispensable. Moreover, with reference to Manfred Max-Neef et al.'s Human Scale Development theory, the argument was put forward that the social systems at the base of the pyramid constitute a human-scale society. In this society, social relationships and institutions remain at a scale that is sufficiently small to ensure every individual is of social significance. This discussion of the theoretically underpinned advantages of a human-scale dimension in the context of poverty alleviation has provided an explanation why the slum dwellers neither intend to scale their entrepreneurial actions nor become included in the markets of global capitalism.

3.7.2 Contributions and Implications

Through the barefoot economic method of linguistic pruning and from lived experience of the social and economic reality in the slums, this scientific case study has answered the research question of why bottom-up franchises are rarely observed at the base of the pyramid grounded in an understanding of relevant social and economic phenomena as they appear to people in the slums in lived experience. Consequently, the conducted research has indicated that bottom-up franchises are rarely observed because there is a *linguistically induced incoherence* between the given social and economic reality at the base of the pyramid and the theoretical presuppositions of the microfranchise concept. This incoherence manifests itself, particularly, in the attempt to create efficient market interactions within a non-market society, and the attempt to develop large-scale business operations within a human-scale society. As demonstrated by the case study, economic institutions and economic scalability are generally not perceived as desirable

goals within the non-market human-scale society of the slums. This incoherence may have led to the dichotomy that while a number of economists have championed bottom-up franchising as a promising model for poverty alleviation, the inhabitants of the slums have not – or, at least, not yet.

The incoherence can, moreover, be regarded as being caused by language. Any theoretical consideration made by a researcher inevitably involves a certain preconceived scientific terminology that is commonly accepted within his or her research community, rendering the theoretical consideration thinkable to him or her. With respect to this case study and the vocabulary that was pruned in its barefoot economic approach, the incoherence-inducing scientific jargon was ascribed to the current prevailing neoliberal discourse in the discipline of the economic sciences (see section 3.4.2 and section 3.4.4.4). As outlined in section 3.4.4.4, the neoliberal discourse emerged in 20th century capitalist societies in the aftermath of Joseph Schumpeter's work and aims to liberate individual entrepreneurial freedoms and skills within an institutional framework characterised by strong private property rights, free markets and free trade (Harvey 2007). The present study indicates that the discourse cannot be used to understand poverty-related phenomena as they appear to people outside modern capitalism. Among other things, this is due to the fact that the discourse compels economists to assume that: (1) strong market institutions; (2) scalable entrepreneurial actions; and (3) inclusion in the globalised markets of modern capitalism are desirable objectives. In the final analysis, the neoliberal discourse must be considered as incoherent with the given social and economic reality at the base of the pyramid.

In terms of practical implications, this means that if development agendas continue to be derived from poverty alleviation concepts from the neoliberal discourse, most notably Prahalad's BoP concept, there is a high risk of dismantling the working social orders at the base of the pyramid and, ultimately, reinforcing what postcolonial researchers have termed in this context "[t]he hegemony of global capitalist order" (Peredo et al. 2018: 13; see also Gibson-Graham 1995; Montgomery et al. 2012; Landrum 2020). In view of this, a major contribution of this case study could be the provision of empirical evidence to the theoretical arguments made in critical discourse analyses of postcolonial studies, namely that the neoliberal discourse on the poverty alleviation of subaltern groups involves a hegemonic performativity that expands modern capitalism to the detriment of functioning non-capitalist approaches to conducting economic life (e.g. Montgomery et al. 2012; Peredo et al. 2018; Landrum 2020).

Overall, the present scientific case study suggests that it may be necessary to make a profound *language shift* in poverty research in the economic sciences. In such a language shift, the neoliberal discourse should be countered by a socioeconomic human scale discourse. Such a human scale discourse may have its academic roots in Manfred Max-Neef et al.'s Theory of Human Scale Development. More precisely, a human scale discourse of this kind is meant to aim at a revitalisation of the social significance of the human individual within an institutional framework characterised by social embeddedness and small-scale operations. The discourse would allow for a language that is more coherent with the social and economic reality outside global capitalism. Subsequently, it could be possible to develop poverty alleviation concepts which overcome the inadequacies of the BoP concept and those models derived from it, such as bottom-up franchising. Among other things, such poverty alleviation concepts should allow for the conceptualisation of native microentrepreneurs at the base of the pyramid not as Schumpeterian entrepreneurs but rather as human-scale entrepreneurs. Entrepreneurship at the human scale could, thereby, be defined as the creation of socially embedded small-scale business ventures for which the individual entrepreneur as well as all other stakeholders are significant as human individuals, and which are

not intended to scale.²⁶ The call for the emergence of a more elaborated concept of the human-scale entrepreneur is one of the major outcomes of this research.

3.7.3 Limitations

The limitations of this scientific study primarily result from the research design derived from the methodology of barefoot economics.

This research was shaped by its case study design, meaning the research question was answered by a barefoot economic research approach involving lived experiences made within a certain case. The research results of scientific case studies regularly possess a higher internal than external validity (Yin 1989). In this study, the internal validity of the research findings was ensured by the validation strategies of triangulation (see section 3.4.4.3) and member-checking (see section 3.4.4.4). The triangulation included both data triangulation and investigator triangulation (see section 3.4.4.3). However, practical limitations inhibiting wider data collection resulted from the limited resources of the MPITO[®] project (see section 3.4.4.1) and the abrupt end to its real-world experiment due to the global 'COVID-19 pandemic' (see section 3.5.1). With regard to the external validity (generalisability) of this study, two types of generalisation, as proposed by Firestone (1993), can be distinguished: (1) "extrapolation from sample to population" (ibid.: 16); and (2) "analytic generalization or extrapolation using a theory" (ibid.). The sample-to-population extrapolation of research results is usually closely linked to the sampling method applied (Flick 2009). In this research study, the method of *targeted sampling* was used to create a panel of experiment participants intended to be representative of the field (see section 3.4.4.2). Accordingly, the results may be generalisable for the Mathare slums of Nairobi. However,

²⁶ Please note that the proposed definition is based on a behavioural, and not functional, idea of entrepreneurship (see *section 3.3.2*; see also Stevenson 1983).

considering the limited data and the small sample size of this case study, the validity of sample-to-population extrapolation is open to question. To claim even broader external validity for the base of the pyramid as a whole, analytic generalisation was used. For this purpose, a number of renowned theories and concepts were referred to as background. This approach could result in a lack of intersubjective comprehensibility (Steinke 1999) due to the fact that the analytic generalisation was based firmly on the phenomenological understanding presupposed to be gained by the barefoot economic research approach. As explained in *chapter 2*, this understanding cannot be shared by means of text, but only arises in lived experience - an issue that must be considered as a major constraint of barefoot economic research in general. However, the present study can be deemed to have benefited from the strengths of barefoot economics in terms of other scientific quality criteria. These strengths include maximum validation in terms of fulfilling the criteria of closeness of the researcher to the object of investigation (Mayring 2016), achieved due to the lived experience of the investigated phenomena. In terms of such validation by means of lived experience, time can be regarded as a significant parameter. In this case, a real-world experiment was conducted over more than five years. Within the duration of the experiment, a diachronic reliability (Kirk & Miller 1986) was achieved. This diachronic reliability manifests itself in the stability of observations in the temporal course of this study (see Flick 2009). The stability of these observations concerned the phenomena of social embeddedness and human scale. Diachronic reliability is usually rarely achieved within empirical social research because it involves "the precondition that the phenomenon under study in itself may not undergo any changes" (Flick 2009: 385). The barefoot economic approach, however, has the prerequisites for achieving diachronic reliability, since it investigates the essence of phenomena, which, by definition, does not undergo change (see Husserl [1900/01] 2001; see also Zhok 2012).

As explained in *chapter 2*, barefoot economics and positive economics complement each other; consequently, the research results of this scientific study should be considered as incomplete by their very nature. Future research based on the methodology of positive economics (see *chapter 2*) should attempt to produce greater scientific knowledge using the language of the socioeconomic human scale discourse suggested by the results of this research (see *section 3.7.2*).

4 Conclusion

4.1 Overall Summary

This treatise set out to contribute to the United Nations' (2015: 14) Sustainable Development Goal 1: "End poverty in all its forms everywhere". Having defined the phenomenon of poverty, in line with the United Nations and the World Bank, as the dissatisfaction of basic human needs, the relevance of long-lasting improvements to poverty alleviation efforts in the face of the global trend of increasing poverty is clear. Moreover, poverty research is particularly pertinent in the scientific discipline of economics, given the view that needs satisfaction is the purpose of all economic activities.

The research context of this treatise highlighted the fact that poverty research in the economic sciences of the 21st century is dominated by the BoP concept and the RCT method. Considering the limitations resulting from the corresponding theoretical and methodological monism and the need to counterbalance the shortcomings of pervasive approaches, the present treatise took its intellectual starting point from the frequently expressed desire for pluralism in economics. In order to meet this objective and to contribute to poverty research by stimulating a greater variety of economic approaches, 'barefoot economics' as proposed by the German-Chilean economist and Alternative Nobel laureate Manfred Max-Neef (1932-2019) in the context of poverty research was chosen as the research subject of this treatise. Two consecutive scientific studies were carried out to revive, elaborate and apply barefoot economics with the aim of identifying the scientific contribution of barefoot economics to poverty research.

The first study was dedicated to barefoot economics in theory. A hermeneutical investigation into the meaning of barefoot economics was conducted, with particular reference to Max-Neef's writings on the philosophy of science. This demonstrated that the prevailing idea of economics as a positive science is based on the philosophy of positivism; hence the methodology of economics sets out to produce positive knowledge. Barefoot economics, in contrast, is rooted in phenomenology. The phenomenological perspective argues that science should be concerned with the understanding of phenomena, i.e. grasp their essence or meaning. Consequently, phenomena can only be understood if they are 'lived through'. Furthermore, a necessary prerequisite is a language that opens the door to understanding - a language coherent with reality. As a result of the investigation, barefoot economics was defined as an approach to economics that seeks to understand the essence or meaning of poverty-related phenomena by means of lived experience and a methodological pruning of language. This definition substantially distinguishes barefoot economics (as a phenomenological economics) from positive economics and other non-positivist approaches in the realm of empirical social research and, specifically, poverty research. Despite the incommensurability of barefoot economics and positive economics, arising from their antagonistic underpinnings from a philosophy of science perspective, the findings of this study show that they can be considered as complementary opposites. Consequently, a dialectical complementation model was developed. Finally, this research suggests that from a philosophy of science perspective there are good reasons for considering barefoot economics in academia.

The second study in this treatise was dedicated to barefoot economics in practice. An empirical case study was conducted, applying the previously established barefoot economic approach in scientific research practice. In the case study, the applicability of barefoot economics as a phenomenological research approach as well as the performative impact of barefoot economics on real existent poverty were demonstrated. In more concrete terms, the case study was designed as a multi-year real-world experiment on bottom-up franchising at the base of the pyramid. The experiment was performed by an own development project, called the MPITO[®]

project, in the Mathare slums of Nairobi from January 2015 to March 2020. In terms of the barefoot economic approach of the study, extensive lived experiences were gained in the slums and a methodological pruning of language was carried out. The research results show that the self-reliant poverty alleviation efforts of the slum dwellers differed significantly to the assumptions made within the theoretical background of the study. In the final analysis, it was concluded that barefoot economics can raise awareness of the linguistically induced incoherence between, on one hand, the given social and economic reality at the base of the pyramid and, on the other hand, the theoretical presuppositions commonly taken for granted in poverty research in the economic sciences. To overcome this incoherence, a profound language shift is called for; i.e., more precisely, a shift towards a language that is coherent with the social and economic reality outside global capitalism, and, thus, towards a language that opens the door to phenomenological understanding. In terms of achieving such a language shift, the research findings indicate the need to counter the dominant neoliberal discourse in poverty research in the economic sciences by what has been termed a socioeconomic human scale discourse. Poverty alleviation concepts developed based on this human scale discourse should allow for the conceptualisation of native microentrepreneurs at the base of the pyramid as human-scale entrepreneurs.

4.2 Overall Contributions and Implications

The overall research question of this treatise asked what scientific contribution barefoot economics can make to poverty research. With regard to this overall research question, the following main findings resulted from the two consecutive scientific studies conducted.

The first study indicates that barefoot economics can make a fundamental scientific contribution to poverty research due to its capability to *complement positive knowledge with phenomenological understanding*. It has been argued that the practice of barefoot economics requires more than simply the lived experience of poverty-related phenomena. In contrast to the prevailing positivist paradigm within the scientific discipline of economics that tends to cultivate particular ways of economic thinking by taking their linguistic presuppositions for granted, barefoot economics involves challenging one's own horizon of possibility for economic thought by putting commonly accepted academic jargon in abevance. In doing so, barefoot economics grasps the essence of poverty-related phenomena. This phenomenological understanding goes hand in hand with a language that is coherent with the given social and economic reality. A language of this kind, in turn, enables poverty researchers to pose more accurate, i.e. 'fit-for-reality', questions. Those questions can be answered by means of positive economics with the aim of developing more effective and appropriate poverty alleviation concepts. In more general terms, it can be argued that barefoot economics can prevent positive economics from making inaccurate theorizations

The second study indicates that barefoot economics can make a more advanced scientific contribution to poverty research due to its capability to create awareness of the need for a post-neoliberal human scale discourse. The phenomenological understanding of povertyrelated phenomena that can be attained through practicing barefoot economics at the base of the pyramid enables poverty researchers to detect incoherence of the dominant neoliberal discourse with the given social and economic reality outside global capitalism. Simultaneously, barefoot economics leads to the conscious adoption of a discourse that centres around the idea of human scale. The latter discourse is, thereby, identified as a discourse that opens the door to a phenomenological understanding of poverty-related phenomena as they appear in lived experience at the base of the pyramid - a discourse that is coherent with reality. Finally, it can be argued that barefoot economics enables positive economists to create more effective and appropriate poverty alleviation concepts based on the language of such a human scale discourse.

By reflecting on the results of both scientific studies in the context of the theoretical and methodological monism in poverty research in economics (see *section 1.2*), this treatise has demonstrated that barefoot economics is capable of escaping the prevalent methodology of positive economics, of which the RCT method is a major manifestation, and the dominant neoliberal discourse, of which the BoP concept is a major manifestation. In doing so, barefoot economics has great potential to contribute significantly to the desired *pluralism in economics* (see *section 1.3*) and to counterbalance the limitations of the pervasive approaches to poverty research.

Apart from the scientific contributions that barefoot economics can make to poverty research, the second scientific study of this treatise revealed that barefoot economics is capable of *alleviating poverty in the course of its research practice*. In terms of its performative impact on society, barefoot economics responds to the plea of sustainability researchers for *'transformative economics'* (Schneidewind et al. 2016a, 2016b; Barth & Rommel 2020) "that does not only observe and describe societal transformation processes, but rather initiates and catalyzes them" (Schneidewind et al. 2016a: 6). An essential feature of transformative economics is also the "*participation* of non-scientific actors" (Barth & Rommel 2020: 300). As indicated by the second study of this treatise, barefoot economics is capable of ensuring such participation.

Overall, this treatise has theoretically and empirically demonstrated that barefoot economics – as a transformative, languagesensitive, phenomenological economics – is capable of making substantial contributions to poverty research within the scientific discipline of economics. Consequently, this treatise makes a clear call for barefoot economics to be more widely considered in academia. Through its effort to revive, elaborate and apply barefoot economics, this treatise has formulated barefoot economics into a scientific research approach that could be applied methodologically in future research.

4.3 Overall Limitations

Both the scientific studies presented in this treatise have limitations. The validity of the theoretical conceptualisation of barefoot economics made in the first study was limited by pre-interpretations that inevitably arise in hermeneutical investigations, given the fact that the interpreter will always be locked within a particular language. The intersubjective comprehensibility of the findings of the second study were affected by the constraint inevitably faced by barefoot economics: phenomenological understanding cannot be shared through the written word, but only arises in lived experience.

In terms of the overall contribution made by this treatise, it is apparent that there are further limitations. Having formulated barefoot economics into a methodological approach applicable for future research, it should be acknowledged that its application appears to be restricted in terms of *repeatability* by the same researcher (compare Kenett & Shmueli 2015). The reason for this lies once more in the phenomenological foundation of barefoot economics. As Martin Heidegger ([1975] 1988: 328) explains in The Basic Problems of Phenomenology: phenomenological method, if applied properly, "become[s] necessarily obsolete" in "the progress made by following it". More precisely, phenomenological method is not an arbitrarily repeatable technique (like positivist methods) but rather a 'ladder', helping us 'to get at the things themselves', that becomes itself dispensable after its usage and can be thrown away (compare Hegel [1807] 2003: 14; Wittgenstein ([1921] 2002: 89). Repeatability becomes impaired because the researcher cannot return to his or her initial state in an ordered manner (since the researcher cannot use the 'ladder' anymore). In more concrete terms, the researcher cannot simply return to their previous state of lacking understanding of the phenomenon under investigation, as that was a state of mind before the phenomenon was 'lived through' and language became coherent.

4.4 Outlook

A major result of this treatise was the identification of the need for a socioeconomic human scale discourse. Thereby, this treatise put forward the idea that such a human scale discourse would be more coherent than the prevailing neoliberal discourse for the social and economic reality at the base of the pyramid. Based on that argument, this treatise claims that the defined human scale discourse may lead to more appropriate approaches for alleviating poverty of people who live outside the global capitalist system and, by doing so, contribute to achieving the United Nations' (2015: 14) Sustainable Development Goal 1: "End poverty in all its forms everywhere". The following outlook on the avenues for future research takes a closer look at the broader potential of such a human scale discourse above and beyond the aforementioned research context.¹

Since sustainability efforts are often confronted with "dilemmas of upscaling" (Augenstein et al. 2020: 146), a socioeconomic human scale discourse could also be considered appropriate for helping to achieve a *multitude* of the United Nations' Sustainable Development Goals, even *within* capitalist societies. Considering, for example, Sustainable Development Goal 13: "Take urgent action to combat climate change and its impacts" (United Nations 2015: 14), a human scale discourse could provide important contributions to achieving the goal's more concrete targets. Among other things, these targets include a drastic reduction of anthropogenic green-

¹ This is done given the Max-Neefian postulate that "we are living [...] in a situation of dangerous incoherence: our language is incoherent with our historical reality" (Max-Neef [1989] 1991b: 109). The postulate coincides with Max-Neef's (2009: 18; italics in original removed) claim that "[w]e have arrived at a point in our human evolution where [...] we understand very little" (see also *section 2.3.2*). Moreover, the generic incoherence of our language with our historical reality can itself be explained by means of a human scale discourse in broader terms (compare *section 4.5*); arguing that the incoherence came about "[i]n the course of human evolution, [with] the transition from oral communication [...] to written communication" (Max-Neef 2005a: 10).

house gas emissions, as agreed under the 2016 Paris Agreement in the UN Framework Convention on Climate Change (UNFCCC). A socioeconomic human scale discourse not only advocates downsizing to small economic units but also supports economic relocalisation in which production processes become geographically closer to consumption processes. The ecological benefits of such a relocalisation in terms of greenhouse gas reductions are already evident in sustainability research (see e.g. Levidow & Psarikidou 2011; Bueno 2012).

This treatise has also explained that the development theory associated with a socioeconomic human scale discourse, namely Max-Neef et al.'s Human Scale Development theory, shifts the perspective on development away from economic goods towards non-material satisfiers, and ultimately decouples the concept of economic development from that of economic growth (see *section 3.6.4*). Such a *decoupling* is regularly considered as indispensable by sustainability researchers to ensure an ecologically sustainable future within the Anthropocene (e.g. von Weizsäcker et al. 1997; Schneidewind 2018).

In addition to Human Scale Development theory, further economic theories could be developed within a socioeconomic human scale discourse. The resultant set of economic theories could ultimately lead to an own school of economic thought, which Max-Neef once imagined being called "*human scale economics*" (Max-Neef 1985: 40, own italics, [1986] 2005b: 43; see also Smith & Max-Neef 2011: 134; Max-Neef 2010b). Such a human scale economics could, among other things, involve a fundamentally different theorisation of entrepreneurship (see *section 3.7.2*) and the protection of local economies, including for example local currencies and sociocratic policies (see also Smith & Max-Neef 2011; Fuders & Max-Neef 2014). Ultimately, human scale economics could meet the desire for a substantial transformation of modern economic
thought as articulated within the $21^{\mbox{\scriptsize st}}$ century sustainability discourse.^2

4.5 Final Remarks

The following final remarks address the issue of *circularity* (Flick 2009, Baur 2019) within the research process of this treatise. Circularity of the research process can be considered as a characteristic feature of many qualitative research designs (ibid.). It denotes the iterative process of asking the same research questions repeatedly to provide increasingly refined answers (ibid.).³

To enhance its clarity for the reader, the research process of this treatise was presented as linear in terms of two consecutive studies (see *section 1.3*). From this simplistic linear perspective, barefoot economics was elaborated as a research approach in the first study. Subsequently, the established research approach was put into practice in the second study. However, from a more complex circular perspective, the final outcome of the second study can be used to answer the initial research question of the first study and, therefore, to create a closed loop. Expressed in more concrete terms, the initial research question concerning how to define barefoot economics can be answered by using the language of a human scale discourse, which resulted from the practice of barefoot economics. The shift of language, thereby, changes the horizon of thinking and, consequently, the horizon of possibilities for a hermeneutical

² The 21st century's 'sustainability discourse' here refers to the political discourse that centres around the transformation of those aspects of modernity expected to harm human wellbeing in the long term.

³ As described by Flick (2009), circular research processes have advantages and disadvantages: "circularity causes problems where the general linear model of research (theory, hypotheses, operationalization, sampling, collecting data, interpreting data, validation) is used to evaluate research. In general, this is the case [...] in the evaluation of [...] research and its results by the use of traditional quality indicators [...] However, notwithstanding that problem, [...] circularity [...] forces the researcher to permanently reflect on the whole research process and on particular steps in the light of the other steps" (Flick 2009: 92).

investigation into the meaning of barefoot economics. However, in order to transfer the previously described human scale discourse from its socioeconomic context to a philosophy of science context, it is necessary to define a human scale discourse as a discourse centring around the non-nullification of the human individual in broader terms. Accordingly, a human scale discourse in a philosophy of science context has to reconsider the scientist as being an essential determining component of scientific research. Such a reconsideration allows for the following line of argument that reinterprets barefoot economics' phenomenological agenda: as explained in section 2.3.1, phenomenology aims to counter the 'naïve objectivism' of the positive sciences which underestimate the role of the individual human being as the researching subject. Consequently, phenomenology emphasises the significance of the individual researcher due to the scientific importance of his or her lived experiences and language. By implication, phenomenology can be viewed as a philosophy promoting the methodological practice of science at the human scale. Having underpinned barefoot economics by a phenomenological philosophy of science, barefoot economics, ultimately, reveals its meaning as an "economics as practised at the human scale" (Max-Neef's ([1982] 1992: 22, own italics).

References

- Adorno, T. W., Dahrendorf, R., Pilot, H., Albert, H., Habermas, J. & Popper, K. R. (1972). *Der Positivismusstreit in der deutschen Soziologie*. Neuwied: Luchterhand.
- Aghion, P. & Howitt, P. (1992). Endogenous Growth Theory Through Creative Destruction. *Econometrica*, 60(2), 323–351.
- Alcoff, L. M. (2020). Decolonizing Feminist Theory: Latina Contributions to the Debate. In: Pitts, A. J., Ortega, M. & Medina, J. (Eds.). *Theories of the Flesh: Latinx and Latin American Feminisms, Transformation, and Resistance* (11–28). New York: Oxford University Press.
- Alcorn, S. & Solarz, B. (2006). The Autistic Economist. Yale Economic Review, 1(38), 13–18.
- Aldiabat, K. M. & Navenec, L. (2011). Philosophical Roots of Classical Grounded Theory: Its Foundations in Symbolic Interactionism. *Qualitative Report*, 16(4), 1063–1080.
- Amnesty International (2009). Kenya: The Unseen Majority: Nairobi's Two Million Slum-Dwellers. London: Amnesty International Publications.
- Andvig, J. C. & Barasa, T. (2014). *A Political Economy of Slum Spaces: Mathare Valley* [NUPI Working Paper No. 838]. Oslo: Norwegian Institute of International Affairs.
- Arora, S. & Romijn, H. (2012). The Empty Rhetoric of Poverty Reduction at the Base of the Pyramid. Organization, 19(4), 481–505. DOI: 10.1177/1350508411414294.
- Augenstein, K., et al. (2020): From Niche to Mainstream: The Dilemmas of Scaling Up Sustainable Alternatives. GAIA: Ecological Perspectives for Science and Society, 29(3), 143–147.

Balkema, A. & Pols, A. (2015). Biofuels: Sustainable Innovation or Gold Rush? Identifying Responsibilities for Biofuel Innovations. In: Koops, B. J., Oosterlaken, I., Romijn, H., Swierstra, T., & Hoven, J. van den (Eds.), *Responsible innovation 2: Concepts, Approaches, and Applications* (283–303). Heidelberg: Springer.

Banerjee, A. V. (2007). Making Aid Work. Cambridge, MA: MIT Press.

- Banerjee, A. V. & Duflo, E. (2011). Poor Economics: A Radical Rethinking of the Way to Fight Global Poverty. New York: PublicAffairs.
- Banerjee, A. V., Duflo, E. & Kremer, M. (2019). The Influence of Randomized Controlled Trials on Development Economics Research and on Development Policy. In: Basu, K., Rosenblatt, D. & Sepúlveda, C. (Eds.), *The State of Economics, The State of the World* (439–487). Cambridge, MA: MIT Press.
- Barth, J. & Rommel, F. (2020). Transformative Economics: Calling for a More Conscious Relationship between Economics and Society. In: Decker, S., Elsner, W. & Flechtner, S. (Eds.), *Principles and Pluralist Approaches in Teaching Economics: Towards a Transformative Science* (298–304). London: Routledge.
- Baunsgaard, V. V. & Clegg, S. R. (2015). Innovation: A Critical Assessment of the Concept and Scope of Literature. In: Agarwal, R., Selen, W., Roos, G. & Green, R. (Eds.), *The Handbook of Service Innovation* (5–25). London: Springer.
- Baur, N. (2019). Linearity vs. Circularity? On Some Common Misconceptions on the Differences in the Research Process in Qualitative and Quantitative Research. *Frontiers in Education*, 23(4), 1–15.
- Beckert, J. (2007). The Great Transformation of Embeddedness: Karl Polanyi and the New Economic Sociology [MPIfG Discussion Paper No. 07/1]. Cologne: Max Planck Institute for the Studies of Societies.
- Bello, A. A. (2009). *The Divine in Husserl and Other Explorations*. Dordrecht: Springer.
- Bhidé, A. (2003). *The Origin and Evolution of New Businesses*. Oxford: Oxford University Press.

- Blackwell, R., Chatha, J. & Nell, E. (Eds.) (2016). *Economics as Worldly Philosophy*. New York: Springer.
- Boeri, M. W. (2007). A Third Model of Triangulation Continuing the Dialogue with Rhineberger, Hartmann, and Van Valey. *Journal of Applied Social Science*, 1(1), 42–48.
- Boland, L. A. (1991). Current Views on Economic Positivism. In: Greenaway, D., Bleaney, M. & Stewart, I. (Eds.), Companion to Contemporary Economic Thought (88–104). London: Routledge.
- Bouwel, J. van (2005). Towards a Framework for Pluralism in Economics. *Post-Autistic Economics Review*, 1(30), 24–27.
- Brady, D. & Burton, L. M. (Eds.) (2016). The Oxford Handbook of the Social Science of Poverty. New York: Oxford University Press.
- Braeckman, A. (2006). Niklas Luhmann's Systems Theoretical Redescription of the Inclusion/Exclusion Debate. *Philosophy & Social Criticism*, 32(1), 65–88.
- Brinkman, R. (1995). Economic Growth versus Economic Development: Toward a Conceptual Clarification. *Journal of Economic Issues*, 29(4), 1171–1188.
- Bueno, G. (2012). Analysis of Scenarios for the Reduction of Energy Consumption and GHG Emissions in Transport in the Basque Country. *Renewable and Sustainable Energy Reviews*, 16(4), 1988–1998.
- Burand, D. & Koch, D. W. (2010). Microfinancing: A Business Approach to Fighting Poverty. *Franchise Law Journal*, 30(1), 24–34.
- Caldwell, B. (1980). Positivist Philosophy of Science and the Methodology of Economics. *Journal of Economic issues*, 14(1), 53-76.
- Callon, M. (1998). Introduction: The Embeddedness of Economic Markets in Economics. *The Sociological Review*, 46(1), 1–57.
- Cameron, D. B., Mishra, A. & Brown, A. N. (2016). The Growth of Impact Evaluation for International Development: How Much Have We Learned? *Journal of Development Effectiveness*, 8(1), 1–21.
- Casado Cañeque, F. (2015). Lessons Learned: Moving the Inclusive Business Agenda Forward. In: Casado Cañeque, F. & Hart, S. L. (Eds.),

Base of the Pyramid 3.0: Sustainable Development Through Innovation and Entrepreneurship (5–10). Sheffield: Greenleaf.

- Casado Cañeque, F. & Hart, S. L. (Eds.) (2015). *Base of the Pyramid 3.0: Sustainable Development Through Innovation and Entrepreneurship.* Sheffield: Greenleaf.
- Cato, M. S. (2012). Green Economics: Putting the Planet and Politics Back into Economics. Cambridge Journal of Economics, 36(5), 1033–1049.
- Chatterjee, S. (2014). Engaging with an Emergent Metanarrative: A Critical Exploration of the BOP Proposition. *Organization*, 21(6), 888–906.
- Cheng, T. K. & Kariithi, S. (2008). More Than One Struggle: Ensuring Quality Education for the Next Generation in Mathare Valley [Discussion Paper]. Nairobi: Dignitas.
- Christensen, C. (1997). The Innovator's Dilemma: When New Technologies Cause Great Firms to Fail. Boston, MA: Harvard Business School.
- Coombs, P. H. & Ahmed, M. (1974). Attacking Rural Poverty: How Nonformal Education Can Help. A Research Report for the World Bank Prepared by the International Council for Educational Development. Baltimore: Johns Hopkins University Press.
- Corburn, J. & Karanja, I. (2014). Informal Settlements and a Relational View of Health in Nairobi, Kenya: Sanitation, Gender and Dignity. *Health Promotion International*, 31(2), 258–269. DOI: 10.1093/heapro/dau100.
- Corburn, J., Ngau, P., Karanja, I. & Makau, J. (2012). Mathare Zonal Plan, Nairobi, Kenya: Collaborative Plan for Informal Settlement Upgrading. Retrieved from https://static1.squarespace.com/ static/58d4504db8a79b27eb388c91/t/58e6a6991b10e38c7e857581/ 1491510980376/Mathare_Zonal_Plan_25_06_2012_low_res.pdf. Accessed on 27 January 2020.
- Cruz, I., Stahel, A. & Max-Neef, M. A. (2009). Towards a Systemic Development Approach: Building on the Human-Scale Development Paradigm. *Ecological Economics*, 68(7), 2021–2030.

- Davies, B. & Petersen, E. B. (2005). Neoliberal Discourse in the Academy: The Forestalling of Collective Resistance. *Learning and Teaching in* the Social Sciences, 2(2), 77–98.
- Davies, S. (2014). Nairobi: Silicon Savannah springs to life. *Financial Times*, 4 November 2014. Retrieved from https://www.ft.com/content/ 83b0b4f4-5fa5-11e4-986c-00144feabdc0. Accessed on 26 January 2020.
- de Souza Leão, L. & Eyal, G. (2019). The Rise of Randomized Controlled Trials (RCTs) in International Development in Historical Perspective. *Theory and Society*, 48(3), 383–418.
- Deaton, A. (2010). Instruments, Randomization, and Learning about Development. *Journal of Economic Literature*, 48(2): 424–455.
- Dembek, K., Sivasubramaniam, N. & Chmielewski, D.A. (2020). A Systematic Review of the Bottom/Base of the Pyramid Literature: Cumulative Evidence and Future Directions. *J Bus Ethics*, 165, 365–382. DOI: 10.1007/s10551-019-04105-y.
- Derrida, J. (1972). *Marges de la Philosophie*. Paris: Éditions de Minuit. [English Translation: *Margins of Philosophy*. Chicago: University of Chicago Press, 1982]
- Diekmann, A. (2014). Empirische Sozialforschung: Grundlagen, Methoden, Anwendungen (8th ed.). Reinbek: Rowohlt.
- Dignitas (2012). Parents' Guide to Mathare Valley Community Schools (2nd ed.). Nairobi: Dignitas.
- Dixon, P. (2012). Why the Denial? Low-cost Private Schools in Developing Countries and their Contributions to Education. *Econ Journal Watch*, 9(3), 186–209.
- Dobusch, L. & Kapeller, J. (2012). Heterodox United vs. Mainstream City? Sketching a Framework for Interested Pluralism in Economics. *Journal of Economic Issues*, 46(4), 1035–1058.
- Dodds, S. (1997). Towards a 'Science of Sustainability': Improving the Way Ecological Economics Understands Human Well-being. *Ecological Economics*, 23(2), 95–111.

- Drekonja-Kornat, G. (2010). Gabriel García Márquez in Wien und andere Kulturgeschichten aus Lateinamerika. Münster: LIT Verlag.
- Dutta, U. (2014). Critical Ethnography. In: Mills, J. & Birks, M. (Eds.), Qualitative Methodology: A Practical Guide (89–105). London: SAGE.
- Efron, S. E. & Ravid, R. (2019). *Writing the Literature Review*. New York: Guilford Press.
- Egri, C. P. & Ralston, D. A. (2008). Corporate Responsibility: A Review of International Management Research from 1998 to 2007. *Journal of International Management*, 14(4), 319-339.
- Fairbourne, J. S. (2006). Microfranchising: A New Tool for Creating Economic Self-Reliance. *Journal of Microfinance/ESR Review*, 8(1), 18–23.
- Fairbourne, J. S. (2007). Why Microfranchising is Needed Now: Introduction and Book Overview. In: Fairbourne, J. S., Gibson, S. W. & Dyer, W. G. (Eds.), *Microfranchising: Creating Wealth at the Bottom* of the Pyramid (1–14). Northampton: Edward Elgar Publishing.
- Fairbourne, J. S., Gibson, S. W. & Dyer, W. G. (Eds.) (2007). *Microfranchising: Creating Wealth at the Bottom of the Pyramid.* Northampton: Edward Elgar Publishing.
- Firestone, W. A. (1993). Alternative Arguments for Generalizing from Data as Applied to Qualitative Research. *Educational Researcher*, 22(4), 16–23.
- Fisher, R. A. (1925). *Statistical Methods for Research Workers*. Guildford, UK: Genesis.
- Flick, U. (2009). An Introduction to Qualitative Research (4th ed.). London: SAGE.
- Flynn, S. (2020). Protection as a Human Fundamental Need: Re-Conceiving Signs of Safety for Social Work in the Republic of Ireland. *The British Journal of Social Work*, 0(1), 1–18.
- Foucault, M. (1972). The Archaeology of Knowledge. New York: Pantheon. [Originally published as Foucault, M. (1969). L'Archéologie du Savoir. Paris: Gallimard]

- Friedman, M. (1966). The Methodology of Positive Economics. In: Friedman, M. (Ed.), *Essays in Positive Economics* (3–43). Chicago: University of Chicago Press. [First published 1953]
- Fromm, E. (1941). Escape from Freedom. New York: Rinehart & Co.
- Fuders, F. & Max-Neef, M. A. (2014). Local Money as Solution to Capitalist Global Financial Crises. In: Pirson, M., Steinvorth, U., Largacha-Martinez, C. & Dierksmeier, C. (Eds.), From Capitalistic to Humanistic Business (157–189). London: Palgrave Macmillan.
- Fullbrook, E. (2003). Real Science is Pluralist. In: Fullbrook, E. (Ed.), The Crisis in Economics: The Post-autistic Economics Movement: the first 600 days (118–124). London: Routledge.
- Gadamer, H. G. (2004). Truth and Method (2nd ed.). London: Continuum.
 [Originally published as Gadamer, H. G. (1960). Wahrheit und Methode: Grundzüge einer philosophischen Hermeneutik. Tübingen: J.C.B. Mohr (Paul Siebeck)]
- George, G., McGahan, A. M. & Prabhu, J. (2012). Innovation for Inclusive Growth: Towards a Theoretical Framework and a Research Agenda. *Journal of Management Studies*, 49(4), 661–683.
- Gibson, S. W. (2007). Microfranchising: The Next Step on the Development Ladder. In: Fairbourne, J. S., Gibson, S. W. & Dyer, W. G. (Eds.), *Microfranchising: Creating Wealth at the Bottom of the Pyramid* (17–42). Northampton: Edward Elgar Publishing.
- Gibson, S. W. & Fairbourne, J. (2005). Where There Are No Jobs, Volume 4: The MicroFranchise Handbook. Provo, UT: The Academy for Creating Enterprise.
- Gibson-Graham, J. K. (1995). Identity and Economic Plurality: Rethinking Capitalism and 'Capitalist Hegemony'. *Environment and Planning D: Society and Space*, *13*(3), 275–282.
- Giersch, H. (1984). The Age of Schumpeter. *The American Economic Review*, 74(2), 103–109.
- Glaser, B. G. & Strauss, A. L. (1967). The Discovery of Grounded Theory: Strategies for Qualitative Research. Chicago: Aldine.

- Goodman, A. (2010). US is Becoming an 'Underdeveloping Nation'. *Counterpoise*, *14*(3/4), 40–42.
- Göpel, M. (2016). The Great Mindshift: How a New Economic Paradigm and Sustainability Transformations go Hand in Hand. Berlin: Springer.
- Granovetter, M. (1985). Economic Action and Social Structure: The Problem of Embeddedness. *American Journal of Sociology*, *91*(3), 481–510.
- Greenwood, D. J. & Levin, M. (1998). Introduction to Action Research: Social Research for Social Change. London: SAGE.
- Groß, M., Hoffmann-Riem, H. & Krohn, W. (2005). *Realexperimente: Ökologische Gestaltungsprozesse in der Wissensgesellschaft.* Bielefeld: Transcript.
- Guillén-Royo, M. (2016). Sustainability and Wellbeing: Human-Scale Development in Practice. Abingdon: Routledge.
- Habermann, F. (2008). Der Homo Oeconomicus und das Andere. Baden-Baden: Nomos.
- Habermas, J. (1981). *Theorie des kommunikativen Handelns*. Frankfurt/M.: Suhrkamp.
- Hall, J., Matos, S., Sheehan, L. & Silvestre, B. (2012). Entrepreneurship and Innovation at the Base of the Pyramid: A Recipe for Inclusive Growth or Social Exclusion? *Journal of Management Studies*, 49(4), 785–812.
- Hall, S. (Ed.) (1997). Representation: Cultural Representations and Signifying Practices. London: SAGE.
- Hart, S., Sharma, S. & Halme, M. (2016). Poverty, Business Strategy, and Sustainable Development. Organization & Environment, 29(4), 401–415.
- Harvey, D. (2007). A Brief History of Neoliberalism. Oxford: Oxford University Press.
- Haucap, J. & Erlei, M. (Eds.) (2019). Mainstream vs. Heterodoxe Ökonomik: Forschungsprogramme im Vergleich, *List Forum für Wirtschafts- und Sozialpolitik*, 44(4), 403–932.

- Hausman, D. M. (Ed.) (2008). *The Philosophy of Economics: An Anthology* (3rd ed.). New York: Cambridge University Press.
- Haveman, R. H. (1997). *Poverty Policy and Poverty Research*. Madison, Wisconsin: University of Wisconsin Press.
- Hegel, G. W. F. (2003). The Phenomenology of Mind. Mineola, NY: Dover Publications. [Originally published as Hegel, G. W. F. (1807). System der Wissenschaft: Erster Theil, die Phänomenologie des Geistes. Bamberg: Goebhardt.]
- Heidegger, M. (1947). *Platons Lehre von der Wahrheit*. Bern: Francke. [First published 1942]
- Heidegger, M. (1988). The Basic Problems of Phenomenology. Bloomington: Indiana University Press. [Originally published as Heidegger, M. (1975). Die Grundprobleme der Phänomenologie. Frankfurt/M.: Vittorio Klostermann.]
- Heidegger, M. (1988). Vom Wesen der Wahrheit: Zu Platons Höhlengleichnis und Theätet. Frankfurt/M.: Vittorio Klostermann. [First published 1943]
- Heidegger, M. (1996). *Being and Time*. Albany, NY: State University of New York Press. [Originally published as Heidegger, M. (1927). *Sein und Zeit*. Halle: M. Niemeyer]
- Heilbroner, R. (1999). The Worldly Philosophers: The Lives, Times and Ideas of the Great Economic Thinkers (7th ed.). New York: Simon and Schuster. [First published 1953]
- Heinemann, F. (1934). Goethe's Phenomenological Method. *Philosophy*, 9(33), 67–81.
- Heise, A. (2017). Defining Economic Pluralism: Ethical Norm or Scientific Imperative? Int. J. Pluralism and Economics Education, 8(1), 18–41.
- Heise, A. (2018). Re-Claiming the University: Transforming Economics as a Discipline. *The Journal of Philosophical Economics*, 11(2), 37–66.
- Heise, A. (2019). *Ideology and Pluralism in Economics: A German View* [Working Paper]. Hamburg: University of Hamburg.

- Henderson, H. (1989). Mutual Development: Towards New Criteria and Indicators. *Futures*, 21(6), 571–584.
- Hendrikse, G. W. & Windsperger, J. (2012). Introducing 'Economics and Management of Franchising Networks'. In: Windsperger, J., Cliquet, G., Hendrikse, G. W. & Tuunanen, M. (Eds.), *Economics* and Management of Franchising Networks (1-13). Berlin: Springer.
- Henriques, M. & Herr, M. (2007). The Informal Economy and Microfranchising. In: Fairbourne, J. S., Gibson, S. W. & Dyer, W. G. (Eds.), *Microfranchising: Creating Wealth at the Bottom of the Pyramid* (43– 77). Northampton: Edward Elgar Publishing.
- Hidalgo-Capitán, A. L. & Cubillo-Guevara, A. P. (2017). Deconstruction and Genealogy of Latin American Good Living (Buen Vivir): The (Triune) Good Living and Its Diverse Intellectual Wellsprings. In: Carbonnier, G., Campodónico, H. & Tezanos Vázquez, S. (Eds.), *Alternative Pathways to Sustainable Development: Lessons from Latin America* (23–50). Geneva, Boston: Brill-Nijhoff.
- Holden, C. (1984). Alternative Nobels for Third World Pioneers. *Science*, 226(4677), 952–952.
- Horkheimer, M. & Adorno, T.W. (2003). *Dialektik der Aufklärung: Philosophische Fragmente*. Frankfurt/M.: Fischer. [First published 1944]
- Husserl, E. (1970). The Crisis of European Sciences and Transcendental Phenomenology: An Introduction to Phenomenological Philosophy. Evanston: Northwestern University Press. [Originally published as Husserl, E. (1936). Die Krisis der europäischen Wissenschaften und die transzendentale Phänomenologie: Eine Einleitung in die phänomenologische Philosophie. Philosophia, 1, 77–176]
- Husserl, E. (1975). *Experience and Judgment*. Evanston: Northwestern University Press. [Originally published as Husserl, E. (1939). *Erfahrung und Urteil. Untersuchungen zur Genealogie der Logik.* Prag: Academia]
- Husserl, E. (1982). Cartesian Meditations: An Introduction to Phenomenology (7th ed.). The Hague: Martinus Nijhoff. [Originally published

as Husserl, E. (1931). Meditations Cartesiennes: Introduction à la Phenomenologie. Paris: A. Colin]

- Husserl, E. (1983). Ideas Pertaining to a Pure Phenomenology and to a Phenomenological Philosophy. First Book: General Introduction to a Pure Phenomenology. The Hague: Martinus Nijhoff. [Originally published as Husserl, E. (1913). Ideen zu einer reinen Phänomenologie und phänomenologischen Philosophie, 1. Buch: Allgemeine Einführung in die reine Phänomenologie. Halle: M. Niemeyer]
- Husserl, E. (1989). The Origin of Geometry. In: Derrida, J., Edmund Husserl's origin of geometry: An introduction (157-180). Lincoln: University of Nebraska Press. [Originally published as Husserl, E. (1939). Die Frage nach dem Ursprung der Geometrie als intentionalhistorisches Problem, Revue Internationale de Philosophie, 1(2), 203-225]
- Husserl, E. (1998). The Paris Lectures. Dordrecht: Kluwer Academic Publishers [Originally published as Husserl, E. (1950): Cartesianische Meditationen und Pariser Vorträge. The Hague: Martinus Nijhoff]
- Husserl, E. (2001). Logical Investigations. London: Routledge [Originally published as Husserl, E. (1900/01): Logische Untersuchungen. Halle: M. Niemeyer]
- Husserl, E. (2002). Philosophy as Rigorous Science. New Yearbook for Phenomenology and Phenomenological Philosophy, 2, 249–295 [Originally published as Husserl, E. (1910). Philosophie als strenge Wissenschaft. Logos. Internationale Zeitschrift für Philosophie der Kultur, 1, 289–341]
- IFC (2014). Better Education for Children in Africa: IFC's Unique Combination of Investment and Advisory Services Shows Results. Retrieved from http://www.ifc.org/wps/wcm/connect/b5d3418049 70c15b9e6ede336b93d75f/Fact%2BSheet-Africa-Better%2BEducat ion%2Bfor%2BChildren.pdf?MOD=AJPERES. Accessed on 13 February 2014.
- Ilahiane, H. & Sherry, J. W. (2012). The Problematics of the 'Bottom of the Pyramid' Approach to International Development: The Case of Micro-entrepreneurs' Use of Mobile Phones in Morocco. *Information Technologies & International Development*, 8(1), 13–26.

- Imas, J. M., Wilson, N. & Weston, A. (2012). Barefoot Entrepreneurs. Organization, 19(5), 563-585.
- Jaiswal, A. K. (2007). Fortune at the Bottom of the Pyramid: An Alternate Perspective [Working paper]. Ahmedabad: Indian Institute of Management Research and Publications.
- Jolibert, C., Max-Neef, M., Rauschmayer, F. & Paavola, J. (2011). Should We Care About the Needs of Non-humans? Needs Assessment: A Tool for Environmental Conflict Resolution and Sustainable Organization of Living Beings. *Environmental Policy and Governance*, 21(4), 259–269.
- Jolibert, C., Paavola, J. & Rauschmayer, F. (2014). Addressing Needs in the Search for Sustainable Development: A Proposal for Needs-Based Scenario Building. *Environmental Values*, 23(1), 29–50.
- Karanja, I. & Makau, J. (2009). An Inventory of the Slums in Nairobi. Nairobi: Pamoja Trust. Retrieved from https://knowyourcity. info/wp-content/uploads/2015/04/Nairobi_slum_inventory_jan_ 09.pdf. Accessed on 27 January 2020.
- Karnani, A. (2007). The Mirage of Marketing to the Bottom of the Pyramid: How the Private Sector can Help Alleviate Poverty. *California Management Review*, 49(4), 90–111.
- Karnani, A. (2009). *The Bottom of the Pyramid Strategy for Reducing Poverty: A Failed Promise* [Working Paper]. New York: United Nations.
- Katouzian, H. (1980). *Ideology and Method in Economics*. London: MacMillan.
- Kelley Jr., B. & Kester, G. H. (Eds.) (2017). Collective Situations: Readings in Contemporary Latin American Art, 1995–2010. Durham: Duke University Press.
- Kenett, R. S. & Shmueli, G. (2015). Clarifying the Terminology That Describes Scientific Reproducibility. *Nature Methods*, 12(8), 699–699.
- KICD (2018). New Education Curriculum to be Fully Adopted by 2027. Retrieved from https://kicd.ac.ke/in_the_press/new-education-c urriculum-to-be-fully-adopted-by-2027/. Accessed on 27 January 2020.

- King, J. E. (2002). Three Arguments for Pluralism in Economics. Journal of Australian Political Economy, 1(50), 82–88.
- Kirk, J. L. & Miller, M. (1986). *Reliability and Validity in Qualitative Research.* Beverly Hills, CA: SAGE.
- Kistruck, G. M., Sutter, C. J., Lount Jr., R. B. & Smith, B. R. (2013). Mitigating Principal-Agent Problems in Base-of-the-Pyramid Markets: An Identity Spillover Perspective. Academy of Management Journal, 56(3), 659–682.
- Kistruck, G. M., Webb, J. W., Sutter, C. J. & Bailey, A. V. (2015). The Double-edged Sword of Legitimacy in Base-of-the-Pyramid Markets. *Journal of Business Venturing*, 30(3), 436–451.
- Kistruck, G. M., Webb, J. W., Sutter, C. J. & Ireland, R. D. (2011). Microfranchising in Base-of-the-Pyramid Markets: Institutional Challenges and Adaptations to the Franchise Model. *Entrepreneurship Theory and Practice*, 35(3), 503–531.
- Kleining, G. (1986). Das Qualitative Experiment. Kölner Zeitschrift für Soziologie und Sozialpsychologie, 38(4), 724–750.
- Kleining, G. & Witt, H. (2000). The Qualitative Heuristic Approach: A Methodology for Discovery in Psychology and the Social Sciences. Rediscovering the method of introspection as an example. *Forum Qualitative Sozialforschung*, 1(1). Retrieved from https://www.qualit ative-research.net/index.php/fqs/article/viewFile/1123/2496. Accessed on 26 January 2020.
- Kleining, G. & Witt, H. (2001). Discovery as Basic Methodology of Qualitative and Quantitative Research. *Forum Qualitative Sozialforschung*, 2(1). Retrieved from http://www.qualitative-research.net/index.php/fqs/article/viewFile/977/2131. Accessed on 26 January 2020.
- Knüppel, S. & Groß, C. (2011). Mikrofinanzierung in der Entwicklungszusammenarbeit: Bildungsunternehmertum am Beispiel der Opportunity Microschools. In: Hackenberg, H. & Empter, S. (Eds.), Social Entrepreneurship - Social Business: Für die Gesellschaft unternehmen (235–251). Wiesbaden: Springer.

Kohr, L. (1956). Meta-Economics. Social Science, 31(3), 140-146.

- Kohr, L. (1957). *The Breakdown of Nations*. London: Routledge & Kegan Paul.
- Kohr, L. (1992). Foreword. In: Max-Neef, M. A., From the Outside Looking In: Experiences in Barefoot Economics' (9–14). London: Zed Books. [First published 1982]
- Kolk, A. & Lenfant, F. (2012). Business-NGO Collaboration in a Conflict Setting: Partnership Activities in the Democratic Republic of Congo. *Business & Society*, 51(3), 478–511.
- Kolk, A., Rivera-Santos, M. & Rufín, C. (2014). Reviewing a Decade of Research on the 'Base/Bottom of the Pyramid' (BOP) Concept. *Business & Society*, 53(3), 338–377. DOI: 10.1177/0007650312474928.
- Kolk, A. & Tulder, R. van (2010). International Business, Corporate Social Responsibility and Sustainable Development. *International Business Review*, 19(2), 119–125.
- Kronauer, M. (2009). Inklusion Exklusion: Eine Historische und Begriffliche Ann\u00e4herung an die Soziale Frage der Gegenwart. In: Kronauer, M. (Ed.), Inklusion und Weiterbildung: Reflexionen zur Gesellschaftlichen Teilhabe in der Gegenwart (24–58). Bielefeld: Bertelsmann Verlag. DOI: 10.3278/14/1106w024.
- Kuhn, T. (1974). Second Thoughts on Paradigms. In: Suppe, F. (Ed.), *The Structure of Scientific Theories* (459–482). Urbana: University of Illinois Press.
- Kuhn, T. (1996). *The Structure of Scientific Revolutions* (3rd ed.). Chicago: University of Chicago Press. [First published 1962]
- Labrousse, A. (2016a). Not by Technique Alone. A Methodological Comparison of Development Analysis with Esther Duflo and Elinor Ostrom. *Journal of Institutional Economics*, 12(2), 277–303.
- Labrousse, A. (2016b). Apprendre des Expérimentations Aléatoires. Promesse de Scientificité, Complications Pratiques, Expériences Historiques. *La Vie des idées*, *4*, 1–10.
- Laclau, E. & Mouffe, C. (2001). *Hegemony and Socialist Strategy: Towards a Radical Democratic Politics* (2nd ed.). London: Verso.

- Lal, D. (2002). *The Poverty of Development Economics*'. London: Institute of Economic Affairs. [First published 1983]
- Landrum, N. E. (2007). Advancing the 'Base of the Pyramid' Debate. Strategic Management Review, 1(1), 1–12.
- Landrum, N. E. (2020). A Critical Discourse Analysis to Explain the Failure of BoP Strategies. *Critical Perspectives on International Business*, ahead-of-print. DOI: 10.1108/cpoib-08-2018-0066.
- Lategan, B. (1997). Living Alone Together: On the Boundaries of the Self and the Collective Mind. *Scriptura: Journal for Biblical, Theological and Contextual Hermeneutics*, 63(4), 467–473.
- Lawson, T. (1997). Economics and Reality. London: Routledge.
- Lawson-Lartego, L. M. (2016). Designing and Orchestrating Embedded Innovation Networks: An Inquiry into Microfranchising in Bangladesh. Atlanta, Georgia: Georgia State University.
- Lazzarotti, F., Samir Dalfovo, M. & Emil Hoffmann, V. (2011). A Bibliometric Study of Innovation Based on Schumpeter. *Journal of* technology management & innovation, 6(4), 121–135.
- Levidow, L. & Psarikidou, K. (2011). Food Relocalization for Environmental Sustainability in Cumbria. Sustainability, 3(4), 692–719.
- Lewin, K. (1946). Action Research and Minority Problems. Journal of Social Issues, 2(4), 34-46.
- Lisciandra, C. (2020). Reflections on the 2019 Nobel Memorial Prize Awarded to Banerjee, Duflo, and Kremer. *Erasmus Journal for Philos*ophy and Economics, 13(1), 79–92.
- London, T. & Hart, S. L. (2011). Creating a Fortune with the Base of the Pyramid. In: London, T. & Hart, S. L. (Eds.), Next Generation Business Strategies for the Base of the Pyramid: New Approaches for Building Mutual Value (1–18). Upper Saddle River: Pearson FT Press.
- Lucena, J. (2010). What is Engineering for? A Search for Engineering beyond Militarism and Free-markets. In: Downey, G. & Beddoes, K. (Eds.), What is Global Engineering Education For? The Making

of International Educators (361–383). San Rafael, CA: Morgan & Claypool Publishers.

- Luhmann, N. (1995a). Soziologische Aufklärung, Band 6: Die Soziologie und der Mensch. Opladen: Westdeutscher Verlag.
- Luhmann, N. (2006). System as Difference. Organization, 13(1), 37-57.
- Luhmann, N. (2008). Beyond Barbarism, Soziale Systeme, 14(1), 38–46. DOI: 10.1515/sosys-2008-0104. [Originally published as Luhmann, N. (1995b): Jenseits von Barbarei. In: Luhmann, N., Gesellschaftsstruktur und Semantik: Studien zur Wissenssoziologie der Modernen Gesellschaft, Band 4 (138–150). Frankfurt/M.: Suhrkamp.

Luhmann, N. (2017). Systemtheorie der Gesellschaft. Berlin: Suhrkamp.

- Lyotard, J. F. (1984). The Postmodern Condition: A Report on Knowledge. Manchester: Manchester University Press. [Originally published as Lyotard, J. F. (1979). La Condition Postmoderne: Rapport sur le Savoir. Paris: Minuit]
- Madrueño, R. & Martínez-Osés, P. (2019). Development Studies in Spanish: A Critical and Constructive Response from the Peripheries. In: Baud, I., Basile, E., Kontinen, T. & Itter, S. von (Eds.), *Building Development Studies for the New Millennium* (51–73). Cham: Palgrave Macmillan.
- Magnani, R., Sabin, K., Saidel, T. & Heckathorn, D. (2005). Review of Sampling Hard-to-Reach and Hidden Populations for HIV Surveillance. *Aids*, 19(2), 67–72.
- Malinowski, B. (1922). Argonauts of the Western Pacific: An Account of Native Enterprise and Adventure in the Archipelagoes of Melanesian New Guinea. London: Routledge & Kegan Paul.
- Marissa, K. E. (2020). Analyzing Neoliberal Discourse: An Integrated Dialectical-Relational Critical Discourse Analysis-Discourse Theory Framework Utilizing Conceptual Metaphor. *Text & Talk*, 40(2), 147–170.
- Max-Neef, M. A. (1985). Another Development under Repressive Rule. Development Dialogue, 1985:1(23), 30-55.

- Max-Neef, M. A. (1991a). About the Pruning of Language (and Other Unusual Exercises) For the Understanding of Social Improvement. In: Max-Neef, M. A., *Human Scale Development: Conception, Application and Further Reflections* (93–103). New York: Apex Press. [Originally published as Max-Neef, M. A. (1988): The Pruning of Language. *Development*, 2(3)]
- Max-Neef, M. A. (1991b). A Stupid Way of Life. In: Max-Neef, M. A., Human Scale Development: Conception, Application and Further Reflections (105–114). New York: Apex Press. [Originally published as Max-Neef, M. A. (1989). A Stupid Way of Life. Schumacher Memorial Lectures 1989, Bristol, UK: Schumacher Society]
- Max-Neef, M. A. (1992). From the Outside Looking In: Experiences in 'Barefoot Economics'. London: Zed Books. [First published 1982]
- Max-Neef, M. A. (2005a): Foundations of Transdisciplinarity. *Ecological Economics*, 53(1), 5–16.
- Max-Neef, M. A. (2005b): Human-Scale Economics: The Challenges Ahead. In: Ekins, P. (Ed.), *The Living Economy: A New Economics* in the Making (43–53). New York: Routledge. [First published 1986]
- Max-Neef, M. A. (2007). La Dimensión Perdida: La Deshumanización Del Gigantismo. Montevideo: Editorial Nordan.
- Max-Neef, M. A. (2009). From Knowledge to Understanding: Navigations and Returns. *Development Dialogue*, 2009(52) [Special issue: What Next: The Case of Pluralism, Vol. 2], 15–21.
- Max-Neef, M. A. (2010b). The World on a Collision Course and the Need for a New Economy. *Ambio*, *39*(3), 200–210.
- Max-Neef, M. A. (2016). Philosophy of Ecological Economics. International Journal of Economics & Management Sciences, 5(5), 1–5. DOI: 10.4172/2162-6359.1000366.
- Max-Neef, M. A. (2019, May 16). [An Interview by Chris Benner], RLC Santa Cruz, CA. Retrieved from https://www.youtube.com/watch? v=UkyXCuxCikA&t=749s. Accessed on 24 January 2020.
- Max-Neef, M. A., Elizalde, A. & Hopenhayn, M. (1989). Human Scale Development: An Option for the Future. *Development Dialogue*,

1989:1(29), 5–81. [Originally published as Max-Neef, M. A., Elizalde, A. & Hopenhayn, M. (1986). Desarrollo a Escala Humana: Una Opción para el Future. *Development Dialogue, 1986:1-2*(25), special edition]

- Mayring, P. (2016). *Einführung in die qualitative Sozialforschung* (6th ed.). Weinheim: Beltz.
- McCaffrey, G., Raffin-Bouchal, S. & Moules, N. (2012). Hermeneutics as Research Approach: A Reappraisal. *International Journal of Qualitative Methods*, 11(3), 214–229.
- McCraw, T. K. (2007). Prophet of Innovation: Joseph Schumpeter and Creative Destruction. Cambridge, MA: Harvard University Press.
- McDonald, H. P. (2012). Speculative Evaluations: Essays on a Pluralistic Universe. Amsterdam: Brill Rodopi.
- Meagher, K. (2015). Leaving No One Behind? Informal Economies, Economic Inclusion and Islamic Extremism in Nigeria. *Journal of International Development*, 27(6), 835–855. DOI: 10.1002/jid.3117.
- Meier, G. (Ed.) (1994). From Classical Economics to Development Economics. Basingstoke: MacMillan.
- Menger, K. (1979). Remarks on the Law of Diminishing Returns a Study in Meta-Economics. In: Mulder, H. L. (Ed.), Selected Papers in Logic and Foundations, Didactics, Economics (279–302). Springer, Dordrecht.
- Merleau-Ponty, M. (2005). Phenomenology of Perception. London: Routledge [Originally published as Merleau-Ponty, M. (1945). Phénoménologie de la Perception. Paris: Gallimard]
- Meyer, K. E., Estrin, S., Bhaumik, S. K. & Peng, M. W. (2009). Institutions, Resources, and Entry Strategies in Emerging Economies. *Strategic Management Journal*, *30*(1), 61–80.
- Mkoji, M. M. (2014). Trapped Between?? Struggle for the Urban Poor to Better Housing: The Case of Mathare, Kenya. The Hague: International Institute of Social Studies.
- MoEST (2015). Registration Guidelines for Alternative Provision of Basic Education and Training (APBET). Nairobi: MoEST.

- Montgomery, N., Peredo, A. & Carlson, E. (2012). The BOP Discourse as Capitalist Hegemony. *Academy of Management Proceedings*, 1(1), 1–1.
- Mookherjee, D. (2005). Is There Too Little Theory in Development Economics Today? *Economic and Political Weekly*, 40(40), 4328– 4333.
- Moore, D. B. & Schmitz, G. J. (Eds.) (1995). *Debating Development Discourse: Institutional and Popular Perspectives*. Basingstoke: MacMillan.
- Moore-Gilbert, B., Stanton, G. & Maley, W. (1997). *Postcolonial Criticism*. London: Routledge.
- MPITO[®] Interim Report (2018a). *Report No. 1*, unpublished report, presented at the MPITO[®] head teachers meeting, Nairobi, 14 May 2018.
- MPITO[®] Interim Report (2018b). *Report No.* 2, unpublished report, presented at the MPITO[®] head teachers meeting, Nairobi, 28 May 2018.
- MPITO[®] Interim Report (2018c). *Report No. 3*, unpublished report, presented at the MPITO[®] head teachers meeting, Nairobi, 28 June 2018.
- MPITO[®] Interim Report (2019a). *Report No. 4*, unpublished report, presented at the MPITO[®] head teachers meeting, Nairobi, 13 February 2019.
- MPITO[®] Interim Report (2019b). *Report No. 5*, unpublished report, presented at the MPITO[®] head teachers meeting, Nairobi, 05 March 2019.
- MPITO[®] Interim Report (2019c). *Report No. 6*, unpublished report, presented at the MPITO[®] head teachers meeting, Nairobi, 08 May 2019.
- MPITO[®] Interim Report (2019d). *Report No. 7*, unpublished report, presented at the MPITO[®] head teachers meeting, Nairobi, 02 July 2019.

- MPITO[®] Interim Report (2020). *Report No. 8*, unpublished report, presented at the MPITO[®] head teachers meeting, Nairobi, 14 February 2020.
- Munoz, J. M., Alon, I. & Mitchell, M. C. (2010). Micro-franchising Strategies: Drawing Lessons from Franchise Literature. In: Munoz, J. M. (Ed.), *Contemporary Microenterprise: Concepts and Cases* (155– 179). Cheltenham, UK: Edward Elgar.
- Nassehi, A. & Nollmann, G. (1997). Inklusionen: Organisationssoziologische Ergänzungen der Inklusions-/Exklusionstheorie. Soziale Systeme, 3(2), 393-411.
- Nekrašas, E. (2016). *The Positive Mind*. New York: Central European University Press.
- Nobel Media AB (2020): *The Price in Economic Sciences 2019*. Retrieved from https://www.nobelprize.org/prizes/economic-sciences/2019/press-release/. Accessed on 02 October 2020.
- North, D. C. (1991). Institutions. The Journal of Economic Perspectives, 5(1), 97–112.
- O'Connor, A. (2001). Poverty Knowledge: Social Science, Social Policy, and the Poor in Twentieth-Century U.S. History. Princeton, New Jersey: Princeton University Press.
- Ogden, T. N. (Ed.) (2016). Experimental Conversations: Perspectives on Randomized Trials in Development Economics. Cambridge, MA: MIT Press.
- Overgaard, S. (2004). *Husserl and Heidegger on Being in the World*. Dordrecht: Springer.
- Paterson, M. & Higgs, J. (2005). Using Hermeneutics as a Qualitative Research Approach in Professional Practice. *The Qualitative Report*, 10(2), 339–357.
- Patton, M. Q. (1990). *Qualitative Evaluation and Research Methods* (2nd ed.). Newbury Park, CA: SAGE.

- Peredo, A. M., Montgomery, N. & McLean, M. (2018). The BoP Business Paradigm: What It Promotes and What It Conceals. Oxford Development Studies, 46(3), 411–429.
- Petersen, D. J., Willers, D., Schmitt, E. M., Birnbaum, R., Meyerhoff, J. H., Gießler, S. & Roth, B. (Eds.) (2019). *Perspektiven einer Pluralen Ökonomik*. Wiesbaden: Springer.
- Phelan, S. (2014). Critiquing 'Neoliberalism': Three Interrogations and a Defense. In: Lievrouw, L. A. (Ed.), *Challenging Communication Research* (27–41). New York: Peter Lang.
- Pineda-Escobar, M. A. & Merigó, J. M. (2020). A Bibliometric Analysis of the Base/Bottom of the Pyramid Research. *Journal of Intelligent & Fuzzy Systems*, 38(5), 5537–5551.
- Plehwe, D. (2020). Schumpeter Revival? How Neoliberals Revised the Image of the Entrepreneur. In: Mirowski, P., Plehwe, D. & Slobodian, Q. (Eds.), *Nine Lives of Neoliberalism* (120–142). London: Verso Books.
- Polanyi, K. (1977). The Livelihood of Man. New York: Academic Press.
- Polanyi, K. (2001). The Great Transformation: The Political and Economic Origins of Our Time. Boston: Beacon Press. [First published 1944]
- Popper, K. (1972). Logic of Scientific Discovery. London: Hutchinson. [First published 1935]
- Potter, J., Wetherell, M., Gill, R. & Edwards, D. (1990). Discourse: Noun, Verb or Social Practice? *Philosophical Psychology*, 3(2-3), 205–217.
- Prahalad, C. K. (2005). The Fortune at the Bottom of the Pyramid: Eradicating Poverty Through Profits. Upper Saddle River, NJ: Wharton School Publishing. [First published 2004]
- Prahalad, C. K. & Hart, S. (1999). Strategies for the Bottom of the Pyramid: Creating Sustainable Development [Working paper]. Ann Arbor: University of Michigan.
- Prahalad, C. K., & Hart, S. (2002). The Fortune at the Bottom of the Pyramid. *Strategy+Business*, 26, 1–13.

- Prahalad, C. K., & Lieberthal, K. (1998). The End of Corporate Imperialism. *Harvard Business Review*, 76(4), 68–79.
- Priddat, B. P. (2017). ,*Schöpferische Zerstörung' als Agens Movens der Ökonomie?* [Working Paper No. 2017–44). Witten: Witten/Herdecke University.
- Rappaport, S. (1996). Abstraction and Unrealistic Assumptions in Economics. *Journal of Economic Methodology*, 3(2), 215–236. DOI: 10.1080/13501789600000016.
- Ravallion, M. (2009). Should the Randomistas Rule? *The Economists' Voice*, 6(2), 1–5.
- Reardon, J. (Ed.) (2009). *The Handbook of Pluralist Economics Education*. New York: Routledge.
- Rhees, R. (1998). Wittgenstein and the Possibility of Discourse. Cambridge, UK: Cambridge University Press.
- Right Livelihood Foundation (2020a). Acceptance Speech of Manfred Max-Neef [Hold at the Right Livelihood Award Ceremony on 9 December 1983 in Stockholm]. Retrieved from https://www.rightlivelih oodaward.org/speech/acceptance-speech-manfred-max-neef/. Accessed on 25 January 2020.
- Right Livelihood Foundation (2020b). *Right Livelihood Award Ceremony* on 9 December 1983 in Stockholm. Retrieved from https://www. rightlivelihoodaward.org/laureates/manfred-max-neef/. Accessed on 25 January 2020.
- Rodrik, D. (2006). Goodbye Washington Consensus, Hello Washington Confusion? A Review of the World Bank's Economic Growth in the 1990s: Learning from a Decade of Reform. *Journal of Economic literature*, 44(4), 973–987.
- Rodrik, D. (2009). The New Development Economics: We Shall Experiment, but How Shall We Learn? In: Cohen, J. & Easterly, W. (Eds.), What Works in Development? Thinking Big and Thinking Small (24–47). Washington, DC: Brookings Institution Press.
- Rose, M., Wanner, M. & Hilger, A. (2019). Das Reallabor als Forschungsprozess und -infrastruktur f
 ür nachhaltige Entwicklung: Konzepte, Heraus-

forderungen und Empfehlungen (2nd ed.) [Wuppertal Paper No. 196]. Wuppertal: Wuppertal Institute.

- Said, E. W. (1978). Orientalism. London: Routledge & Kegan Paul.
- Sale, K. (2017). *Human Scale Revisited: A New Look at the Classic Case for a Decentralist Future*. White River Junction: Chelsea Green Publishing.
- Salganik, M. J. & Heckathorn, D. D. (2004). Sampling and Estimation in Hidden Populations Using Respondent-driven Sampling. Sociological Methodology, 34(1), 193–240.
- Saßmannshausen, S. P. (2012). Entrepreneurship-Forschung: Fach oder Modetrend? Evolutorisch-wissenschaftssystemtheoretische und bibliometrischempirische Analysen. Lohmar: Eul Verlag.
- Schirmer, W. & Michailakis, D. (2015). The Luhmannian Approach to Exclusion/Inclusion and Its Relevance to Social Work. *Journal of Social Work*, 15(1), 45–64.
- Schlick, M. (1948). Positivism and Realism. Synthese, 7(6), 478–505. [Originally published as Schlick, M. (1932). Positivismus und Realismus. Erkenntnis, 3(1), 1–31]
- Schneidewind, U. (2018). Die Große Transformation: Eine Einführung in die Kunst Gesellschaftlichen Wandels. Frankfurt/M: Fischer Verlag.
- Schneidewind, U., Augenstein, K., Stelzer, F. & Wanner, M. (2018). Structure Matters: Real-World Laboratories as a New Type of Large-Scale Research Infrastructure: A Framework Inspired by Giddens' Structuration Theory. GAIA: Ecological Perspectives for Science and Society, 27(1), 12–17. DOI: 10.14512/gaia.27.S1.5.
- Schneidewind, U., Singer-Brodowski, M., Augenstein, K. & Stelzer, F. (2016a). Pledge for a Transformative Science: A Conceptual Framework [Wuppertal Paper No. 191]. Wuppertal: Wuppertal Institute.
- Schneidewind, U., et al. (2016b): Transformative Wirtschaftswissenschaft im Kontext nachhaltiger Entwicklung: Für einen neuen Vertrag zwischen Wirtschaftswissenschaft und Gesellschaft. Ökologisches Wirtschaften, 31(2), 30–34.

- Schnell, A. (2019). Was ist Phänomenologie? Frankfurt/M.: Vittorio Klostermann.
- Schumacher, E. F. (2011). Small is Beautiful: A Study of Economics as if People Mattered. London: Random House. [First published 1973]
- Schumpeter, J. A. (1923). Business Cycles: A Theoretical, Historical, and Statistical Analysis of the Capitalist Process. London: McGraw-Hill.
- Schumpeter, J. A. (1947). Theoretical Problems of Economic Growth. The Journal of Economic History, 7(1), 1–9.
- Schumpeter, J. A. (1949). The Theory of Economic Development: An Inquiry into Profits, Capital, Credit, Interest, and the Business Cycle. Cambridge, MA: Harvard University Press. [Originally published as Schumpeter, J. A. (1911). Theorie der wirtschaftlichen Entwicklung. Leipzig: Duncker & Humblot.]
- Schumpeter, J. A. (2003). *Capitalism, Socialism and Democracy*. London: Routledge. [First published 1942]
- Schutz, A. (1957). Max Scheler's Epistemology and Ethics, I. *The Review* of *Metaphysics*, 11(2), 304–314.
- Scott, M. FG. (1989). A New View of Economic Growth. Oxford: Clarendon Press.
- Seamon, D. & Zajonc, A. (Eds.) (1998). Goethe's Way of Science: A Phenomenology of Nature. Albany, NY: State University of New York Press.
- Seligman, B. B. (1969). The Impact of Positivism on Economic Thought. History of Political Economy, 1(2), 256–278.
- Sent, E. M. (2003). Pleas for Pluralism. *Post-Autistic Economics Review*, 1(18), 1–5.
- Simanis, E. & Hart, S. (2008). The Base of the Pyramid Protocol: Toward Next Generation BoP Strategy (2nd ed.). Ithaca, NY: Cornell University.
- Simms, E. M. (2005). Goethe, Husserl, and the Crisis of the European Sciences. *Janus Head*, 8(1), 160–172.

- Smith, P. & Max-Neef, M. A. (2011). Economics Unmasked: From Power and Greed to Compassion and The Common Good. Totnes, Devon: Green Books.
- Smolski, A. (2016). An Academic Object for Decolonization. *Postcolonial Studies*, *19*(3), 365–367.
- Söderbaum, P. (2012). Pluralism and Sustainable Development. Int. J. Pluralism and Economics Education, 3(1), 23–39.
- Spiering, S. & Valle Barrera, M. del (2021). Testing the Quality of Transformative Science Methods: The Example of the Human Scale Development Approach. *Sustainability Science*, 16(5), 1439-1457.
- Spivak, G. C. (1988). Can the Subaltern Speak? In: Nelson, C. & Grossberg, L. (Eds.), *Marxism and the Interpretation of Culture* (271–313). Basingstoke: Macmillan.
- Springer, S. (2012). Neoliberalism as Discourse: Between Foucauldian Political Economy and Marxian Poststructuralism. *Critical Discourse Studies*, 9(2), 133–147.
- Stahel, W. R. (2020). History of the Circular Economy. The Historic Development of Circularity and the Circular Economy. In: Eisenriegler, S. (Ed.), *The Circular Economy in the European Union* (7–19). Cham: Springer.
- Steinke, I. (1999). Kriterien Qualitativer Forschung: Ansätze zur Bewertung Qualitativ-Empirischer Sozialforschung. Weinheim: Juventa.
- Stevano, S. (2020). Small Development Questions Are Important, But They Require Big Answers. World Development, 127, 1–3.
- Stevenson, H. H. (1983). A Perspective on Entrepreneurship. Cambridge, MA: Harvard Business School.
- Stichweh, R. (1997). Inklusion/Exklusion und die Theorie der Weltgesellschaft. In: Rehberg, K.-S. (Ed.), *Differenz und Integration: Die Zukunft moderner Gesellschaften* (601–607). Opladen: Westdeutscher Verlag.
- Sunanda, S. (2016). Microfranchising an Inclusive Business Model. *Social Sciences*, *3*(3), 548–557.

- Terzic, L. (2018). The New Paradigm of Innovation Economics in the 21st Century: Solving the Enigma of Economic Growth. *Romanian Journal of Economics*, 46(1/55), 40–59.
- Tooley, J. & Dixon, P. (2005). Private Education Is Good for the Poor: A Study of Private Schools Serving the Poor in Low-income Countries. Washington, D.C.: Cato Institute.
- Torrance, H. (2012). Triangulation, Respondent Validation, and Democratic Participation in Mixed Methods Research. *Journal of Mixed Methods Research*, 6(2), 111–123.
- Toye, J. (2006). Keynes and Development Economics: A Sixty-Year Perspective. Journal of International Development, 18(7), 983–995.
- Truong, V. D., Hall, C. M., & Garry, T. (2014). Tourism and Poverty Alleviation: Perceptions and Experiences of Poor People in Sapa, Vietnam. *Journal of Sustainable Tourism*, 22(7), 1071–1089.
- UN-DESA (2009). *Rethinking Poverty: Report on the World Social Situation 2010*. New York: United Nations.
- UNDP (2008). Creating Value for All: Strategies for Doing Business with the Poor. New York: UNDP.
- UN-Habitat (2015). *Habitat III Issue Paper 22: Informal Settlements* [Participatory Slum Upgrading Programme]. New York: UN-Habitat.
- UN-Habitat (2016). *Slum Almanac 2015/2016: Tracking Improvement in the Lives of Slum Dwellers* [Participatory Slum Upgrading Programme]. Nairobi: UN-Habitat.
- UN-Habitat (2017). António Guterres visits Mathare on first field mission as UN Secretary-General. Retrieved from https://unhabitat.org/ antonio-guterres-visits-mathare-on-first-field-mission-as-un-secreta ry-general. Accessed on 27 January 2020.
- United Nations (1997). *Glossary of Environment Statistics* [Studies in Methods, Series F, No. 67]. New York: United Nations.
- United Nations (2015). Transforming Our World: the 2030 Agenda for Sustainable Development [General Assembly Resolution A/RES/70/1]. New York: United Nations.

- United Nations (2019a). World faces 'climate apartheid' risk, 120 more million in poverty. Retrieved from https://news.un.org/en/story/ 2019/06/1041261. Accessed on 02 October 2020.
- United Nations (2019b). World Population Prospects 2019 [Custom data acquired via website]. Retrieved from https://population.un.org/wpp/DataQuery/. Accessed on 30 August 2019.
- United Nations (2020a). *Goal 1: End poverty in all its forms everywhere*. Retrieved from https://www.un.org/sustainabledevelopment/pove rty/. Accessed on 02 October 2020.
- United Nations (2020b). The Sustainable Development Goals Report 2020. New York: United Nations.
- Wanjiru, M. W. & Matsubara, K. (2017). Slum Toponymy in Nairobi, Kenya. *Urban and Regional Planning Review*, *4*, 21–44.
- Wanner, M., Hilger, A., Westerkowski, J., Rose, M., Stelzer, F. & Schäpke, N. (2018). Towards a Cyclical Concept of Real-world Laboratories: A Transdisciplinary Research Practice for Sustainability Transitions. *disP: The Planning Review*, 54(2), 94–114. DOI: 10.1080/02513625.2018.1487651.
- Watters, J. K. & Biernacki, P. (1989). Targeted Sampling: Options for the Study of Hidden Populations. Social problems, 36(4), 416–430.
- Watzlawick, P., Beavin, J. & Jackson, D. (1967). Pragmatics of Human Communication: A Study of Interactional Patterns, Pathologies, and Paradoxes. New York: Norton.
- Weizsäcker, E. U. von, Lovins, A. B. & Lovins, L. H. (1997). Factor Four: Doubling Wealth, Halving Resource Use. London: Earthscan.
- Weston, A. & Imas, J. M. (2017). Towards a Barefoot Community-Based Entrepreneuring. In: Essers, C., Dey, P., Tedmanson, D. & Verduyn, K. (Eds.), *Critical Perspectives on Entrepreneurship: Challenging Dominant Discourses* (73–90). Abingdon, UK: Routledge.
- Weston, A. & Imas, J. M. (2018). Resisting Colonization in Business and Management Studies: From Postcolonialism to Decolonization. In: Cassell, C., Cunliffe, A. & Grandy, G. (Eds.), *The SAGE Handbook*

of Qualitative Business and Management Research Methods (119–137). Los Angeles: SAGE.

- Williamson, J. (1993). Democracy and the 'Washington consensus'. World Development, 21(8), 1329–1336.
- Williamson, J. (2008). A Short History of the Washington Consensus. In: Serra, N. & Stiglitz, J. E. (Eds.), *The Washington Consensus Reconsidered: Towards a New Global Governance* (14–30). Oxford: Oxford University Press.
- Wittgenstein, L. (1986). *Philosophical Investigations*. Oxford: Blackwell. [First published 1953]
- Wittgenstein, L. (2002). Tractatus Logico-Philosophicus. London: Routledge. [Originally published as Wittgenstein, L. (1921). Logischphilosophische Abhandlung. In: Ostwald, W. (Ed.), Annalen der Naturphilosophie, 14, 185–262]
- Woodworth, W. (2007). Current International Development Tools to Combat Poverty. In: Fairbourne, J. S., Gibson, S. W. & Dyer, W. G. (Eds.), *Microfranchising: Creating Wealth at the Bottom of the Pyramid* (78–98). Northampton: Edward Elgar Publishing.
- World Bank (2005). *Getting to Know the World Bank*. Washington, DC: World Bank.
- Yin, R. K. (1989). Case Study Research. Beverly Hills, CA: SAGE.
- Zhok, A. (2012). The Ontological Status of Essences in Husserl's Thought. New Yearbook for Phenomenology and Phenomenological Philosophy, 11, 96–127.

"... the practice of barefoot economics requires more than simply the lived experience of poverty-related phenomena. In contrast to the prevailing positivist paradigm within the scientific discipline of economics that tends to cultivate particular ways of economic thinking by taking their linguistic presuppositions for granted, barefoot economics involves challenging one's own horizon of possibility for economic thought by putting commonly accepted academic jargon in abeyance."